

ADVERTIMENT. L'accés als continguts d'aquesta tesi queda condicionat a l'acceptació de les condicions d'ús establertes per la següent llicència Creative Commons:

ADVERTENCIA. El acceso a los contenidos de esta tesis queda condicionado a la aceptación de las condiciones de uso establecidas por la siguiente licencia Creative Commons:

WARNING. The access to the contents of this doctoral thesis it is limited to the acceptance of the use conditions set by the following Creative Commons license: https://creativecommons.org/licenses/?lang=en

BY NC ND

Universitat Autònoma de Barcelona Departament de Matemàtiques

Singular integral operators and rectifiability

This thesis is presented to obtain the degree of PhD in Mathematics Petr Chunaev May 2018

We certify that this thesis was written by Petr Chunaev under our supervision.

Advisor Juan Eugenio Mateu Bennassar

Advisor and tutor Xavier Tolsa Domènech

Barcelona, May 2018

Acknowledgements

First of all, I would like to thank my advisors, Joan Mateu and Xavier Tolsa, not only for their wise and patient mentoring but also for their readiness to help me in overcoming numerous obstacles that a foreigner and a young mathematician can face. Moreover, I am very grateful for the financial support associated with the European Research Council Advanced Grant 320501 "Geometric Analysis in the Euclidean Space" (Principal investigator Xavier Tolsa) that made possible my research and life in Barcelona and visiting numerous scientific events all over the world.

Also, I am very thankful to the Departament de Matemàtiques for providing wonderful condition for my work and always friendly attitude towards me.

Additionally, I would like to express my sincere gratitude to Pertti Mattila for the possibility of my research stay at the University of Helsinki from January to April, 2018, and for the financial support provided by the Academy of Finland.

Furthermore, special thanks are to my mentors and colleagues Vladimir Danchenko, Lyudmila Morozova and Sergey Tikhonov who helped and influenced me greatly at earlier stages and who continue supporting me in different instances within and beyond my scientific work now.

I am also grateful to many young colleagues whom I met along the way and who became role models of a productive and passionate mathematician for me, in particular, to Jonas Azzam, Murilo Rodolfo Candido, Mihalis Mourgoglou, Martí Prats, Tuomas Orponen and Malik Younsi.

What is more, I would like to thank my friends around the world for all their encouragement, enthusiasm and faithfulness. Among others, I thank Kirill Angel, Marta Bergamo, Viktoria Bolshakova, Murilo Rodolfo Candido, Clara Darnet, Mariya Fabia, Elisa Giambra, Maria Celeste Gritti, Ainur Jumabayeva, Anton Lehnen, Roman di Liukshini, Mohammad Mahdavipour, Sergey Margolin, Nathália Oliveira, Paweł Rachel, Claudio Scalia, Mikkel Thøgersen and Aleksander Tyshchenko for the days spent together in the Vila Universitària of the Universitat Autònoma de Barcelona.

Last but not the least, I must express my profound gratitude to my mother, Svetlana, and my wife, Kseniya, for providing me with unfailing support, continuous encouragement and unconditional love throughout my years of research and living abroad (and other times of course). This accomplishment would not have been possible without them. Finally, I am grateful to my late father, Vladimir, and brother, Anton, who would be proud of who I have become and whose invisible support helps me to move forward.

Contents

Introduction 6						
1	A r bour 1.1 1.2 1.3 1.4 1.5 1.6	new parametric family of singular integral operators whose L ² - ndedness implies rectifiability by a curvature-like method Introduction Main results Auxiliary results Proof of Theorems 1.1 and 1.2 Additional remarks	15 15 16 17 30 32 38			
2	2 Singular integral operators unsuitable for the curvature method who					
-	L^2 -b	oundedness still implies rectifiability	39			
	2.1	Introduction	39			
	2.2	Main results	40			
	2.3	Proof of Theorems 2.1 and 2.2	41			
	2.4	Proof of Theorem 2.3	44			
	2.5	Additional remarks	53			
2. A family of singular integral approximation which control the Couche						
J	form					
	3.1	Introduction and Theorems	54			
	3.1	Main Lemma and proofs of Theorems	56			
	0.2	3.2.1 Proof of Theorem 3.1	56			
		3.2.2 Proof of Theorem 3.2	57			
	3.3	Several corollaries	57			
	3.4	Plan of the proof of Main Lemma	58			
3.5 The David-Mattila lattice		The David-Mattila lattice	58			
		Balanced cubes and control on beta numbers through permutations	60			
	3.7	Parameters and thresholds	62			
	3.8	Stopping cubes and trees	63			
		3.8.1 Stopping cubes	63			
		3.8.2 Properties of cubes in trees	64			
	3.9	Measure of stopping cubes from $BP(R)$ and $F(R)$	68			
	3.10	Construction of a Lipschitz function	68			
		3.10.1 Auxiliary results	69			
		3.10.2 Lipschitz function F for the good part of R	70			
		3.10.3 Extension of F to the whole L_R	76			
		3.10.4 Γ_R and R are close to each other $\ldots \ldots \ldots$	76			

3.11	Small	measure of the cubes from $LD(R)$	80
3.12	Small	measure of the cubes from $BS(R)$ for R whose best approximation	
	line is	far from the vertical	81
	3.12.1	Auxiliaries and the key estimate for the measure of cubes from $BS(R)$	81
	3.12.2	The measure of cubes from $BS(R)$ is controlled by the permutations	
		of the Hausdorff measure restricted to Γ_R	82
	3.12.3	Estimates for the permutations of the Hausdorff measure restricted	
		to Γ_R	85
	3.12.4	Estimates for the permutations of the Hausdorff measure restricted	
		to Γ_R in the case when points are far from each other $\ldots \ldots \ldots$	88
3.13	The pa	acking condition for Top cubes and the end of the proof of Main Lemma	95
	3.13.1	Properties of the trees	95
	3.13.2	New families of stopping cubes	96
	3.13.3	The corona decomposition	97
	3.13.4	The families of cubes ID_H , ID_U and ID	97
	3.13.5	The packing condition	98
	3.13.6	The end of the proof of Main Lemma	102
3.14	The ca	se of curvature \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 1	103
3.15	Furthe	r generalisations	104

Introduction

The problems that we study in this thesis lie in the area of Harmonic Analysis and Geometric Measure Theory. More precisely, we consider the connection between the analytic properties of singular integral operators defined in $L^2(\mu)$ and the geometric properties of the measure μ . In the forthcoming several pages we will make a historical review of the topic, give necessary definitions and briefly discuss the results that we obtained. Let us mention that the exposition in this thesis is based on the papers [Chu, CMT1, CMT2].

We start with necessary notation and background facts. Note that we work mostly in the plane and therefore usually skip dimension markers in definitions.

Let $E \subset \mathbb{C}$ be a Borel set and B(z, r) be an open disc with center $z \in \mathbb{C}$ and radius r > 0. We denote by $\mathcal{H}^1(E)$ the (1-dimensional) Hausdorff measure of E, i.e. length, and call E a 1-set if $0 < \mathcal{H}^1(E) < \infty$. A set E is called *rectifiable* if it is contained in a countable union of Lipschitz graphs, up to a set of \mathcal{H}^1 -measure zero. A set E is called *purely unrectifiable* if it intersects any Lipschitz graph in a set of \mathcal{H}^1 -measure zero.

By a measure often denoted by μ we mean a positive locally finite Borel measure on \mathbb{C} . We consider Calderón-Zygmund (CZ) kernels $K : \mathbb{C} \setminus \{0\} \to \mathbb{C}$ with the following

properties: there exist constants C > 0 and $\eta \in (0, 1]$ such that $|K(z)| \leq C|z|^{-1}$ for all $z \in \mathbb{C} \setminus \{0\}$, and moreover

$$|K(z) - K(z+\zeta)| \leqslant C \frac{|\zeta|^{\eta}}{|z|^{1+\eta}} \quad \text{if} \quad |\zeta| \leqslant \frac{1}{2}|z|, \qquad z, \zeta \in \mathbb{C}.$$
(0.1)

We will mostly deal with -1-homogeneous CZ kernels K.

Given a measure μ , a CZ kernel K and an $f \in L^1(\mu)$, we define a truncated singular integral operator (SIO) as

$$T_{K,\varepsilon}f(z) := \int_{E \setminus B(z,\varepsilon)} f(\zeta)K(z-\zeta)d\mu(\zeta), \quad \text{where } E = \operatorname{spt} \mu \text{ and } \varepsilon > 0.$$
(0.2)

We do not define the SIO T_K explicitly because several delicate problems such as the existence of the *principal value* (p.v.), i.e. $\lim_{\varepsilon \to 0} T_{K,\varepsilon} f(z)$, might arise. On the contrary, the integral in (0.2) always converges absolutely. Nevertheless, we say that T_K is $L^2(\mu)$ -bounded if the operators $T_{K,\varepsilon}$ are $L^2(\mu)$ -bounded uniformly on ε .

How to relate the $L^2(\mu)$ -boundedness of a certain SIO to the geometric properties of the support of μ is an old problem in Harmonic Analysis. It stems from Calderón's paper [Cal] where it is proved that the Cauchy transform, i.e. the SIO T_K with K(z) = 1/z, is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded if E is a Lipschitz graph with small slope. Later on, Coifman, McIntosh and Meyer [CMM] removed the small Lipschitz constant assumption and thus extended the result to all Lipschitz graphs. Furthermore, in [Dav1] David fully characterized rectifiable curves Γ for which the Cauchy transform is $L^2(\mathcal{H}^1 \lfloor \Gamma)$ -bounded. Namely, they have to satisfy the *linear growth* condition:

$$\mathcal{H}^1(\Gamma \cap B(z,r)) \leqslant Cr \quad \text{for all } r > 0 \text{ and } z \in \mathbb{C} \text{ and some } C > 0.$$
(0.3)

These results led to further development of tools for understanding the above-mentioned problem (see also the corresponding parts of [Chr, DS2, Mat1, Mat2, Mat3, Tol5]).

A new quantitative characterization of rectifiability in terms of the so-called β -numbers introduced by Jones [Jon1] and the concept of uniform rectifiability proposed by David and Semmes [DS1,DS2] are among these tools. Several related definitions for the plane are in order. (We refer the reader to [DS1,DS2] for definitions and results in the multidimensional case). A measure μ is called *Ahlfors-David regular* (or *AD-regular*, for short) if

$$C^{-1}r \leqslant \mu(B(z,r)) \leqslant Cr \quad \text{for all } z \in \operatorname{spt} \mu \text{ and } r \in (0, \operatorname{diam}(\operatorname{spt} \mu)), \tag{0.4}$$

where C > 0 is a fixed constant. Moreover, μ is called *uniformly rectifiable* if it is AD-regular and spt μ is contained in a curve satisfying (0.3).

The well-known David-Semmes problem is stated in the plane as follows: does the $L^2(\mu)$ -boundedness of the Cauchy transform is sufficient for the uniform rectifiability of the AD-regular measure μ ? This problem was settled by Mattila, Melnikov and Verdera:

Theorem A [MMV]. Let μ be an AD-regular measure. The measure μ is uniformly rectifiable if and only if the Cauchy transform is $L^2(\mu)$ -bounded.

Note that an analogous problem in higher dimensions is still unsolved except for the case of codimension 1 settled by Nazarov, Tolsa and Volberg in [NTV] not long ago.

The proof of Theorem A relied on the so-called *curvature* (or *symmetrisation*) *method* that was new at that time but soon became very influential in solving many long-standing problems related to the Cauchy transform and analytic capacity, for example, Painlevé's problem on metric/geometric description of removable singularities for bounded analytic functions, Vitushkin's conjecture and the semiadditivity of analytic capacity (see [Dav2, MMV, Mel, Tol1] and especially historical remarks in [Tol5]). For our purposes it is more convenient to describe a generalised version of the curvature method and so do we.

Let $(z_1, z_2, z_3) \in \mathbb{C}^3$. For a kernel K, consider the following *permutations*:

$$P_K(z_1, z_2, z_3) := \sum_{s \in \mathfrak{S}_3} K(z_{s_2} - z_{s_1}) \overline{K(z_{s_3} - z_{s_1})}, \qquad (0.5)$$

where \mathfrak{S}_3 is the group of permutations of the three elements $\{1, 2, 3\}$. Supposing that μ_1 , μ_2 and μ_3 are measures, set

$$P_K(\mu_1, \mu_2, \mu_3) := \iiint P_K(z_1, z_2, z_3) \, d\mu_1(z_1) \, d\mu_2(z_2) \, d\mu_3(z_3). \tag{0.6}$$

We write $P_K(\mu) := P_K(\mu, \mu, \mu)$ for short and call it *permutation of the measure* μ . Moreover, in what follows $P_{K,\varepsilon}(\mu_1, \mu_2, \mu_3)$ stands for the integral in the right hand side of (0.6) defined over the set

$$\{(z_1, z_2, z_3) \in \mathbb{C}^3 : |z_k - z_j| \ge \varepsilon > 0, \quad 1 \le k, j \le 3, \quad j \ne k\},\tag{0.7}$$

and $P_{K,\varepsilon}(\mu) := P_{K,\varepsilon}(\mu,\mu,\mu).$

The identities (0.5) and (0.6) were first considered by Melnikov [Mel] in the case of the Cauchy kernel. He showed that

$$P_K(z_1, z_2, z_3) = c(z_1, z_2, z_3)^2, (0.8)$$

where K(z) = 1/z,

$$c(z_1, z_2, z_3) := \frac{1}{R(z_1, z_2, z_3)}$$

is the so called *Menger curvature* and $R(z_1, z_2, z_3)$ stands for the radius of the circle passing through z_1 , z_2 and z_3 . Moreover, points z_1 , z_2 and z_3 are collinear if and only if $c(z_1, z_2, z_3) = 0$ (and $R(z_1, z_2, z_3) = \infty$). From what is said it is clear that

$$c(z_1, z_2, z_3) \ge 0$$
 for any $(z_1, z_2, z_3) \in \mathbb{C}^3$, (0.9)

which is very important in applications. Additionally, it is easily seen that Menger curvature can be calculated in geometrical terms in different ways, e.g.

$$c(z_1, z_2, z_3) = \frac{4S(z_1, z_2, z_3)}{|z_1 - z_2||z_1 - z_3||z_2 - z_3|} = \frac{2\sin\widehat{z_1 z_2 z_3}}{|z_1 - z_3|},$$
(0.10)

where $S(z_1, z_2, z_3)$ stands for the area of the triangle (z_1, z_2, z_3) and $\widehat{z_1 z_2 z_3}$ is the angle of this triangle opposite to the side $z_1 z_3$. One can find the proof of these and other formulas, e.g. in [Tol5, Section 3.2].

Melnikov also introduced in [Mel] the notion of *curvature of measure* μ :

$$c^{2}(\mu) := P_{K}(\mu), \text{ where } K(z) = 1/z.$$
 (0.11)

One can also define $c_{\varepsilon}^2(\mu)$ in an obvious way over the set (0.7).

Permutations (0.5) and (0.6) for more general kernels K were considered later by Chousionis, Mateu, Prat and Tolsa in [CMPT1] (see also [CMPT2]).

Now let K be a -1-homogeneous CZ kernel, see before (0.1). Suppose that the permutations (0.5) for K are non-negative for any $(z_1, z_2, z_3) \in \mathbb{C}^3$. If μ is a finite measure with C_* -linear growth, i.e. there exists a constant $C_* > 0$ such that

$$\mu(B(z,r)) \leqslant C_* r \qquad \text{for all } r > 0 \text{ and } z \in \operatorname{spt} \mu, \tag{0.12}$$

then the following relation between truncated versions of $P_K(\mu)$ and T_K1 holds:

$$\|T_{K,\varepsilon}1\|_{L^{2}(\mu)}^{2} = \frac{1}{6}P_{K,\varepsilon}(\mu) + \mathcal{R}_{K,\varepsilon}(\mu), \qquad |\mathcal{R}_{K,\varepsilon}(\mu)| \leq c C_{*}^{2}\mu(\mathbb{C}), \qquad (0.13)$$

with some c > 0 independent of ε . We call (0.13) generalised Melnikov-Verdera's identity. Actually, it was first proved for the Cauchy kernel in the seminal paper [MV] by Melnikov and Verdera where they gave a new geometric proof of the above-mentioned result about the L^2 -boundedness of the Cauchy transform on Lipschitz graphs. It turns out that one can follow Melnikov-Verdera's proof for the Cauchy kernel in order to obtain (0.13) as stated, see e.g. [CMPT2, Section 2].

The formulas (0.8) and (0.13), generating the *curvature method*, are remarkable in the sense that they relate an analytic notion (the SIO T_K , in particular, the Cauchy transform) with a metric-geometric one (permutations, in particular, curvature).

Later on, Theorem A was pushed even further by David and Léger [Leg]. They used the curvature method and, in particular, the property (0.9) to prove the following deep result (see a brief exposition of the proof in Section 1.5).

Theorem B [Leg]. Let E be a 1-set. If $c^2(\mathcal{H}^1 \lfloor E) < \infty$, then E is rectifiable. Moreover, if the Cauchy transform is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded, then E is rectifiable.

Note that the $L^2(\mathcal{H}^1 \lfloor E)$ -boundedness of the Cauchy transform and the identity (0.13) imply that $c^2(\mathcal{H}^1 \lfloor E) < \infty$. Consequently, to prove Theorem B it is enough to prove just the first statement and this was actually done in [Leg]. Let us mention that Theorem B is originally stated in [Leg] in a more general form.

For a long time very few results analogous to Theorem B were known for kernels different from 1/z. For instance, it was clear [MMV, CMPT1] that the same result is true

for the coordinate parts of the Cauchy kernel, in particular, for the real part that we denote by

$$K_{\mathsf{DL}}(z) := \frac{\operatorname{Re} z}{|z|^2} \equiv \operatorname{Re} \frac{1}{z}.$$
(0.14)

Indeed, considering the permutations (0.5), it is not difficult to show that

$$P_{K_{\mathsf{DL}}}(z_1, z_2, z_3) = \frac{1}{2}c(z_1, z_2, z_3)^2.$$
(0.15)

This fact was a motivation point of the paper [CMPT1] by Chousionis, Mateu, Prat and Tolsa, where analogues of Theorems A and B were proved for the kernels

$$\kappa_n(z) := \frac{(\operatorname{Re} z)^{2n-1}}{|z|^{2n}}, \qquad n \in \mathbb{N}.$$
(0.16)

(We also refer the reader to the paper [CP] where the kernels (0.16) and other related ones were studied.) Namely, it is shown in [CMPT1] that

$$P_{\kappa_n}(z_1, z_2, z_3) \ge 0$$
 for any $(z_1, z_2, z_3) \in \mathbb{C}^3$ (0.17)

and $P_{\kappa_n}(z_1, z_2, z_3) = 0$ if and only if points z_1 , z_2 and z_3 are collinear. This is an analogue of (0.9). Moreover, it is proved that the permutations $P_{\kappa_n}(z_1, z_2, z_3)$ behave similarly to $c(z_1, z_2, z_3)^2$ for triangles (z_1, z_2, z_3) with comparable sides such that one side makes a big angle with the vertical line. This fact enables the authors of [CMPT1] to adapt the curvature method from [Leg] to the kernels κ_n . This adaptation however requires several essential modifications in crucial points, where the curvature must be exchanged for the permutations P_{κ_n} . In particular, new arguments are provided in [CMPT1] when Léger's scheme does not work (see Section 1.5 for more details). As a result, the following statements were proved.

Theorem C [CMPT1]. Let $n \ge 1$ and μ be an AD-regular measure. The measure μ is uniformly rectifiable if and only if T_{κ_n} is $L^2(\mu)$ -bounded.

Theorem D [CMPT1]. Let $n \ge 1$ and E be a 1-set. If $P_{\kappa_n}(\mathcal{H}^1 \lfloor E) < \infty$, then E is rectifiable. Moreover, if the SIO T_{κ_n} is $L^2(\mathcal{H}^1 \mid E)$ -bounded, then E is rectifiable.

Obviously, for n = 1 one gets the case of (0.14) (or equivalently the case of the Cauchy kernel due to (0.15)) and thus Theorems A and B. However, for $n \ge 2$ these are the first examples of SIOs that are not directly related to the Cauchy transform and whose $L^2(\mu)$ boundedness implies certain rectifiability properties of μ . In what follows, we will typically deal with the case n = 2 in (0.16) and therefore we set

$$K_{\mathsf{CMPT}}(z) := \frac{(\operatorname{Re} z)^3}{|z|^4}.$$
(0.18)

There are two more results which are actually counterexamples to the property that L^2 -boundedness implies rectifiability. The first one is due to Huovinen.

Theorem E [Huo]. Let K belong to the class \mathcal{H} of odd kernels satisfying

$$\begin{split} |K(x-y) - K(x-z)| &\leqslant \frac{C |y-z|}{|x-y||x-z|}, \qquad |K(z)| \leqslant \frac{C}{|z|}, \qquad x, y, z \in \mathbb{C}, \\ K(r) &= 0, \qquad K(z) = -K(-\overline{z}), \qquad r \in \mathbb{R}, \qquad C = \text{const.} \end{split}$$

Then there exists a purely unrectifiable 1-set E such that the operator T_K , associated with the kernel K, is bounded on $L^2(\mathcal{H}^1|E)$ and, moreover, p.v. T_K exists \mathcal{H}^1 -a.e. and is finite. As a typical example from \mathscr{H} one can take the kernel

$$K_{\mathsf{H}}(z) := \frac{(\operatorname{Re} z)^3}{|z|^4} - \frac{\operatorname{Re} z}{|z|^2}.$$
(0.19)

Another counterexample is due to Jaye and Nazarov. We formulate it here in a slightly different form than it was done originally.

Theorem F [JN]. There exists a purely unrectifiable 1-set E such that the operator T_K , associated with the kernel $K(z) = \overline{z}/z^2$, is bounded on $L^2(\mathcal{H}^1 \lfloor E)$ and, moreover, p.v. T_k fails to exist \mathcal{H}^1 -a.e.

For further exposition it is important to notice that the same statement holds for the (renormalised) real part of $K(z) = \overline{z}/z^2$, namely, for the kernel

$$K_{\mathsf{JN}}(z) := \frac{1}{4} \frac{\overline{z}}{z^2} = \frac{(\operatorname{Re} z)^3}{|z|^4} - \frac{3}{4} \frac{\operatorname{Re} z}{|z|^2}.$$
 (0.20)

Take into account that the examples of purely unrectifiable sets from Theorems E and F are quite intricate, essentially use analytic properties of the kernels and avoid the curvature method (as it is actually unavailable as we will see below).

Note that up to now we meant only -1-homogeneous CZ kernels in the plane which we are mostly interested in. However, there are other interesting examples of kernels in different settings with a certain connection between the $L^2(\mu)$ -boundedness of the corresponding operator and the geometry of μ . For example, David [Dav3] constructed odd and even CZ kernels in the plane that define $L^2(\mathcal{H}^1|E)$ -bounded SIOs, where E is a purely unrectifiable set (namely, the corner quarters Cantor set). Unfortunately, David's kernels are not -1homogeneous. Chousionis [Cho] had similar results for a larger class of kernels for some s-dimensional fractals with s < 1. Recently Chousionis and Li [CL] gave examples of -1-homogeneous kernels in the Heisenberg group with a direct connection between the $L^2(\mu)$ -boundedness of the corresponding singular integral operators and the rectifiability of μ . Unlike the Euclidean case, where most of kernels related to rectifiability are odd, Chousionis-Li's kernels defined in the Heisenberg group are even and non-negative.

Now we are going to summarise the above-mentioned results for -1-homogeneous kernels in the plane (defined in (0.14) and (0.18)-(0.20)) and formulate more precisely the questions that we consider in this thesis. From what is said we conclude that an affirmative answer to the question does the $L^2(\mathcal{H}^1|E)$ -boundedness of T_K implies that the 1-set E is rectifiable is given by curvature-like methods essentially only for the SIOs associated with the kernels K_{DL} and K_{CMPT} . On the other hand, only two kernels, $K = K_{\text{H}}$ and $K = K_{\rm JN}$, are known to be such that T_K is still $L^2(\mathcal{H}^1|E)$ -bounded for some purely unrectifiable 1-sets E (the curvature method is not available in both cases). Thus there is a natural problem consisting in finding other -1-homogeneous CZ kernels in the plane and corresponding SIOs whose $L^2(\mu)$ -boundedness does or does not imply that μ is rectifiable (uniformly rectifiable in the AD-regular case). Of course, we also want to partly move forward in characterizing the class of such kernels but this seems to be a much more ambitious and difficult objective so we will make just a few remarks on it. Another related problem that we will briefly discuss is whether $L^2(\mu)$ -boundedness of a certain SIO (besides the Cauchy transform) implies that all other "reasonable" SIOs in the plane are $L^2(\mu)$ -bounded.

To move in achieving the above-mentioned objectives, in Chapter 1 we introduce the following t-parametric family of kernels:

$$k_t(z) := \frac{(\operatorname{Re} z)^3}{|z|^4} + t \cdot \frac{\operatorname{Re} z}{|z|^2}, \quad \text{where } t \in \mathbb{R}, \quad \text{and} \quad k_\infty(z) := \frac{\operatorname{Re} z}{|z|^2}.$$
(0.21)

Sign of permutations



Figure 0.1: The overall picture of previously known results for the kernels k_t .

One can immediately notice that this is a reasonable generalisation of the kernels (0.14) and (0.18)–(0.20), see Figure 0.1. Indeed,

$$K_{\mathsf{DL}} \equiv k_{\infty}, \qquad K_{\mathsf{CMPT}} \equiv k_0, \qquad K_{\mathsf{H}} \equiv k_{-1} \qquad \text{and} \qquad K_{\mathsf{JN}} \equiv k_{-3/4}.$$

Thus studying SIOs associated with the kernels (0.21) for different t may provide useful settings to further understanding the connection between the L^2 -boundedness and rectifiability. Now we give an overall picture of the results that we obtained in this direction. To avoid repetitions, we do not number the statements here but indicate the corresponding results in further chapters.

In Chapter 1 we study the permutations associated with the kernels (0.21) and prove the following result that complements the ones for t = 0 and $t = \infty$ (see Theorem 1.1 and Remark 1.1).

Theorem. Let k_t be a kernel of the form (0.21), where $t \in (-\infty, -2] \cup (0, +\infty)$. Then

 $P_{k_t}(z_1, z_2, z_3) \ge 0$ for all $(z_1, z_2, z_3) \in \mathbb{C}^3$.

Furthermore, the range of the parameter t is sharp in the sense that there are triples (z_1, z_2, z_3) such that $P_{k_t}(z_1, z_2, z_3)$ change sign if $t \in (-2, 0)$.

This theorem implies that the curvature method is not available directly if $t \in (-2, 0)$, and surprisingly exactly this interval contains the known counterexamples from Theorems E and F. However, by adapting a curvature-type method from [Leg, CMPT1] we will prove the following result for the rest values of t that provides a big family of SIOs whose L^2 boundedness implies rectifiability (see Theorem 1.2).

Theorem. Let k_t be a kernel of the form (0.21), where $t \in (-\infty, -2] \cup (0, +\infty)$, and E a 1-set. If $P_{k_t}(\mathcal{H}^1 \lfloor E) < \infty$, then E is rectifiable. Moreover, if the SIO T_{k_t} is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded, then E is rectifiable.

What is more, the situation in the plane where $t \in (-2, 0)$ is somehow similar to the one in higher dimensions where curvature-type methods are *likely unavailable* (see [NTV, Far]) and one has to come up with other tools instead. As for our case, in Chapter 2 we propose a *perturbation method* that enables us to prove the following result, being the first example in the plane when the curvature method fails but it is still possible to prove that L^2 boundedness implies rectifiability (see Theorem 2.2).

Theorem. Let k_t be a kernel of the form (0.21), where $t \in (-2, -\sqrt{2})$, and E a 1-set. If the SIO T_{k_t} is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded, then E is rectifiable.

The proof of this result is based on the following (see Lemma 2.2 and Remark 2.2).

Lemma. It holds that

$$P_{k_0}(z_1, z_2, z_3) \leqslant 2P_{k_\infty}(z_1, z_2, z_3)$$
 for all $(z_1, z_2, z_3) \in \mathbb{C}^3$.

Additionally, the constant 2 is sharp in the sense that for any small $\varepsilon > 0$ there are triples (z_1, z_2, z_3) such that $P_{k_0}(z_1, z_2, z_3)/P_{k_{\infty}}(z_1, z_2, z_3) \ge 2 - \varepsilon$.

Note that this lemma with an implicit constant is contained in [CMPT2, Lemma 7]. Nevertheless, the explicitness of our constant is essential here and actually enables us to obtain the required result by the perturbation method in Chapter 2.

The latter lemma yields that $P_{k_0,\varepsilon}(\mu) \leq 2P_{k_{\infty},\varepsilon}(\mu)$ for any measure μ . This and (0.13) immediately imply that the L^2 -norm of $T_{k_0}1$ is controlled by the L^2 -norm of $T_{k_{\infty}}1$ (with an explicit constant). The triangle inequality and a proper version of T1-theorem which form the perturbation method then give the required theorem.

Unfortunately, an inequality of the form $P_{k_{\infty}}(z_1, z_2, z_3) \leq C \cdot P_{k_0}(z_1, z_2, z_3)$ with an absolute constant C > 0 hardly exists and this makes the study of kernels (0.21) for small negative t more difficult. Nevertheless, the following lemma holds as shown in Chapter 3 (see Inequality (3.4)).

Lemma. There exist absolute constants $t_0 > 0$ and c > 0 such that for any finite measure μ with C_* -linear growth it holds that

$$P_{k_{\infty}}(\mu) \leq t_0^{-2} P_{k_0}(\mu) + c C_*^2 \mu(\mathbb{C}).$$

In a similar manner as in Chapter 2, from the latter lemma we get that the L^2 -norm of $T_{k_{\infty}}1$ (and thus of the Cauchy transform of measure) is controlled by the L^2 -norm of $T_{k_0}1$ (see Theorem 3.1). Applying the perturbation method yields the following theorem using the same absolute constant t_0 as above (see Theorem 3.2).

Theorem. Let k_t be a kernel of the form (0.21), where $t \in (-t_0, 0)$, and E a 1-set. If the SIO T_{k_t} is $L^2(\mathcal{H}^1|E)$ -bounded, then E is rectifiable.

To prove the latter Lemma and Theorem, we use the David-Mattila dyadic lattice from [DM] and a corona decomposition that is similar to the one in [AT]. More precisely, we split the lattice into some collections of cubes, which are called *trees*, where the density of μ does not oscillate too much and most of the measure is concentrated close to the graph of a Lipschitz function. To construct this function, we use a variant of the Whitney extension theorem adapted to the David-Mattila dyadic lattice. Further, we show that the family of trees of the corona decomposition satisfies a packing condition by arguments inspired by some of the techniques used in [AT] and earlier in [Tol3] to prove the bilipschitz "invariance" of analytic capacity.

It is worth mentioning that the structure of our trees is more complicated than in [AT]. This is because we deal with permutations which are not pointwise comparable to curvature in general and this leads to additional technical difficulties.

Sign of permutations



Figure 0.2: The overall picture of known results for the kernels k_t .

Finally, we summarise all the results known for the kernels k_t in Figure 0.2, cf. Figure 0.1.

Let us also mention that our results in Chapters 1–3 (more precisely, Lemmas 1.5, 1.6 and 2.3 and Theorem 3.1) together with the obvious case $t = \infty$ imply the following necessary and sufficient condition.

Theorem. Let μ be a measure with linear growth and

$$t \in (-\infty, -\sqrt{2}) \cup (-t_0, \infty]. \tag{0.22}$$

The Cauchy transform is $L^2(\mu)$ -bounded if and only if so is the SIO T_{k_t} .

This has further consequences. For example, one can characterise uniformly rectifiable measures via L^2 -bounded SIOs T_{k_t} (see Theorems 2.1, 2.3 and Corollary 3.1).

Corollary. Let μ be an AD-regular measure and k_t a kernel of the form (0.21) with t as in (0.22). The measure μ is uniformly rectifiable if and only if the SIO T_{k_t} is $L^2(\mu)$ -bounded.

The part of this Corollary for $t = \infty$, i.e. in fact for the Cauchy transform, was proved in [MMV] (see Theorem A) and for t = 0 in [CMPT1] (see Theorem C).

The following result is from Section 2.5 and Corollary 3.2.

Corollary. Let μ be a measure with linear growth and k_t a kernel of the form (0.21) with t as in (0.22). If the SIO T_{k_t} is $L^2(\mu)$ -bounded, then so are all 1-dimensional SIOs associated with a wide class of kernels.

By "a wide class of kernels" we mean the class of kernels K = K(z), where $z \in \mathbb{C} \setminus \{0\}$, which are odd \mathcal{C}^2 -functions satisfying

$$\left|\nabla^{j} K(z)\right| \leqslant \frac{C(j)}{|z|^{1+j}} \quad \text{for all } z \in \mathbb{C} \setminus \{0\} \text{ and } j \in \{0, 1, 2\}.$$

$$(0.23)$$

It is easily seen that (0.23) implies that K is a CZ kernel with the properties indicated before (0.1), where $\eta = 1$.

The latter Corollary for $t = \infty$, i.e. actually for the Cauchy transform, was earlier proved in [Tol2, Tol4], see also [Gir].

It is worth mentioning that some of our results stated above are actually proved in Chapters 1–3 not just for k_t but for more general *t*-parametric kernels:

$$K_t(z) := \kappa_N(z) + t \cdot \kappa_n(z), \ t \in \mathbb{R}, \quad K_\infty(z) := \kappa_n(z), \quad n \leqslant N, \ n, N \in \mathbb{N},$$
(0.24)

i.e. linear combinations of the kernels (0.16) of different order. Clearly, $K_t(z) \equiv k_t(z)$ for (n, N) = (1, 2). For the sake of simplicity, we do not state the corresponding results in Introduction but thoroughly discuss them in the subsequent chapters (see Theorems 1.1, 1.2, 2.3, 3.3, 3.4 and 3.5).

A few words about notation. We use the letters c and C to denote constants which may change their values at different occurrences. On the other hand, constants with subscripts such as A_0 or c_1 do not change their values throughout *each chapter*. In a majority of cases constants depend on some parameters which are usually indicated explicitly, for instance, we write $C(\varepsilon)$ or C_{ε} if C depends on ε . If there is a constant C such that $A \leq CB$, we write $A \leq B$. Furthermore, $A \approx B$ is equivalent to saying that $A \leq B \leq A$, possible with different implicit constants. If the implicit constant in expressions with " \leq " or " \approx " depends on some positive parameter, say α , we write $A \leq_{\alpha} B$ or $A \approx_{\alpha} B$.

Equations, theorems, lemmas and other statements are numbered within a chapter. Introduction is considered as Chapter 0 for numbering.

Chapter 1

A new parametric family of singular integral operators whose L^2 -boundedness implies rectifiability by a curvature-like method

1.1 Introduction

The exposition in this chapter is based on [Chu]. Below, we consider the SIOs associated with the kernels (0.24), i.e. with

$$K_t(z) = \frac{(\operatorname{Re} z)^{2N-1}}{|z|^{2N}} + t \cdot \frac{(\operatorname{Re} z)^{2n-1}}{|z|^{2n}} \quad \text{and} \quad K_\infty(z) = \frac{(\operatorname{Re} z)^{2n-1}}{|z|^{2n}},$$
(1.1)

where n and N are positive integer numbers such that $N \ge n$, and $t \in \mathbb{R}$. We prove that the $L^2(\mathcal{H}^1 \lfloor E)$ -boundedness of these operators implies that E is rectifiable for $t \in \mathbb{R} \setminus (t_1, t_2)$ with certain t_1, t_2 depending only on n and N. We use the curvature method for this so all the definitions related to curvature and permutations given in Introduction will be used in this chapter. We nevertheless remind some of them and introduce new ones now.

For a CZ kernel K, we introduced the following *permutations* in (0.5):

$$P_K(z_1, z_2, z_3) = \sum_{s \in \mathfrak{S}_3} K(z_{s_2} - z_{s_1}) \overline{K(z_{s_3} - z_{s_1})},$$

where \mathfrak{S}_3 is the group of permutations of three elements $\{1, 2, 3\}$. Note that the kernels (1.1) that we deal with in this chapter are real-valued so the bar in the latter sum may be skipped. For this reason it is more convenient for us to consider the permutations

$$p_{K}(z_{1}, z_{2}, z_{3})$$

$$:= K(z_{1} - z_{2})K(z_{1} - z_{3}) + K(z_{2} - z_{1})K(z_{2} - z_{3}) + K(z_{3} - z_{1})K(z_{3} - z_{2})$$
(1.2)
$$= \frac{1}{2}P_{K}(z_{1}, z_{2}, z_{3}),$$

where K is an odd and real-valued kernel.

Taking into account the definitions (0.6) and (0.7) and supposing that μ , μ_1 , μ_2 and μ_3 are measures, we also set

$$p_K(\mu_1, \mu_2, \mu_3) := \frac{1}{2} P_K(\mu_1, \mu_2, \mu_3), \qquad p_{K,\varepsilon}(\mu_1, \mu_2, \mu_3) := \frac{1}{2} P_{K,\varepsilon}(\mu_1, \mu_2, \mu_3)$$
(1.3)

and

$$p_K(\mu) := \frac{1}{2} p_K(\mu, \mu, \mu), \qquad p_{K,\varepsilon}(\mu) := \frac{1}{2} P_{K,\varepsilon}(\mu, \mu, \mu).$$
 (1.4)

We will use (1.3) and (1.4) many times below without additional mentioning.

Recall that by a measure we mean a positive locally finite Borel measure on \mathbb{C} . Bear in mind that (1.1) can be written in the form (0.24), i.e.

 $K_t(z) = \kappa_N(z) + t \cdot \kappa_n(z), \quad t \in \mathbb{R}, \qquad K_\infty(z) = \kappa_n(z), \quad n \leq N, \ n, N \in \mathbb{N},$

where, according to (0.16),

$$\kappa_m(z) = \frac{(\operatorname{Re} z)^{2m-1}}{|z|^{2m}}, \qquad m \in \mathbb{N}.$$

Note also that in our new terms it follows from (0.17) and (1.3) that

 $p_{\kappa_m}(z_1, z_2, z_3) \ge 0$ for any $(z_1, z_2, z_3) \in \mathbb{C}^3$ and $m \in \mathbb{N}$,

and $p_{\kappa_m}(z_1, z_2, z_3) = 0$ if and only if z_1, z_2 and z_3 are collinear.

In the forthcoming sections, in order to find values of t such that a result analogous to Theorems B and D is valid for the kernels (1.1), we first study the sign of the permutations (1.2) and then, for the case when these permutations are non-negative, adapt the scheme from [CMPT1] to our situation.

1.2 Main results

First of all let us mention that the case t = 0 in the theorems below agrees with the inequality (0.17) and Theorem D, proved in [CMPT1]. We now indicate the values of t such that the permutations $p_{K_t}(z_1, z_2, z_3)$ are non-negative for all triples (z_1, z_2, z_3) .

Theorem 1.1. Let K_t be a kernel of the form (1.1) with t = 0 or

$$t \in \mathbb{R} \setminus \left(-\frac{1}{2} \left(3 + \sqrt{9 - 4\frac{N}{n}} \right), 2 - \frac{N}{n} \right), \qquad n < N \leq 2n, \quad (1.5)$$

$$t \in \mathbb{R} \setminus \left(-\frac{1}{2} \left(3 + \rho_{n,N} + \sqrt{(3 + \rho_{n,N})^2 - 4\frac{N}{n}} \right), \rho_{n,N} \right), \qquad N \ge 2n, \tag{1.6}$$

where $\rho_{n,N} := (\frac{N}{n} - 2)\sqrt{N - 2n}$. In particular, $t \in \mathbb{R} \setminus (-2; 0)$ for N = 2n. Then

$$p_{K_t}(z_1, z_2, z_3) \ge 0$$
 for all $(z_1, z_2, z_3) \in \mathbb{C}^3$

Furthermore, the range of the parameter t in the case N = 2n is sharp.

Remark 1.1. The conditions (1.5) and (1.6), guaranteeing that $p_{K_t}(z_1, z_2, z_3) \ge 0$, cannot be weakened much in the following sense. As we will show in Section 1.4, there are triples (z_1, z_2, z_3) such that $p_{K_t}(z_1, z_2, z_3)$ can change sign if

$$t \in \left(-\frac{N}{n}, 0\right) \quad \text{for all } n \text{ and } N,$$

$$t \in \left(0, \frac{2}{e^{3/2}} \frac{N}{n}\right) \quad \text{for } N \gg n \quad (\text{i.e. } N \text{ is large enough with respect to } n).$$
(1.7)

Surprisingly, in this context the case t = 0 is an isolated point with non-negative permutations. Thus the curvature method, requiring the permutations to be non-negative, cannot be applied directly for t indicated in (1.7).

From the aforesaid it follows that (1.5) and (1.6) are indeed sharp for N = 2n.

Figure 1.1 illustrates Theorem 1.1 and this remark for n = 3 and different N > 3(consider the horizontal line passing through a fixed *positive integer* N in order to determine the corresponding t). The green area represents the values of t, indicated in (1.5) and (1.6), i.e. those where p_{K_t} are guaranteed to be non-negative (the boundaries are included). In the blue area (the boundaries are not included), obtained by computer-based exhaustive search, the permutations can change sign. Note that the part of the blue area for t < 0is exactly the former in (1.7). Moreover, the part for t > 0 quite agrees with the latter in (1.7). The white area is not covered by our results and, generally speaking, we can say nothing about the sign of p_{K_t} therein. However, computer experiments suggest that the permutations p_{K_t} are non-negative everywhere except the blue area and thus (1.7) seems to give likely boundaries for t, whose corresponding permutations can change sign.

Relying deeply on Theorem 1.1, we will prove the following result.

Theorem 1.2. Let K_t be a kernel of the form (1.1) with t as in Theorem 1.1 and E a 1-set. If $p_{K_t}(\mathcal{H}^1 \lfloor E) < \infty$, then E is rectifiable. Moreover, if the operator T_{K_t} is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded, then E is rectifiable.

Remark 1.2. It is known that for t = -1 (see the red line in Figure 1.1), which belongs to the area, where the permutations can change sign, the statement of Theorem 1.2 is not valid anymore, i.e. L^2 -boundedness does not imply rectifiability. Indeed, it is a corollary of Theorem E since all the kernels K_t of the form (1.1) with t = -1 belong to Huovinen's class \mathscr{H} as can be easily checked.

For the subfamily (0.21) of the kernels (1.1) with (n, N) = (1, 2), i.e. for

$$k_t(z) = \frac{(\operatorname{Re} z)^3}{|z|^4} + t \cdot \frac{\operatorname{Re} z}{|z|^2}$$

even more is known as we have already mentioned in Introduction. Namely, for t = -3/4, whose corresponding p_{k_t} change sign, there also exists a purely unrectifiable set E such that T_{k_t} is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded as follows from Theorem F.

Recall Figure 0.2 that illustrates the known results for the kernels k_t . By Theorems 1.1 and 1.2, if $t \in \mathbb{R} \setminus (-2,0)$, then the permutations p_{k_t} are always non-negative and the $L^2(\mathcal{H}^1 \lfloor E)$ -boundedness of T_{k_t} implies the rectifiability of E. By the arguments from Remarks 1.1 and 1.2, the permutations p_{k_t} for $t \in (-2,0)$ change sign and there are two values of t such that the operator T_{k_t} is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded but E is not rectifiable.

1.3 Auxiliary results

This section is devoted to several auxiliary lemmas, which will be used to prove Theorems 1.1 and 1.2 in Section 1.5.

If a kernel K is real and odd, then one can show that the permutations (1.2) are invariant under translations. This can be done, e.g. by the substitutions $u = z_1 - z_2$ and $v = z_1 - z_3$. Consequently, one point can be always fixed and it is enough to consider only permutations of the form

$$p_K(0, u, v) = K(u)K(v) + K(u)K(u - v) + K(v)K(v - u),$$
(1.8)



Figure 1.1: Theorem 1.1 and Remark 1.1 for n = 3 and different N > 3.

where $u, v \in \mathbb{C} \setminus \{0\}$ are distinct points. The kernels (0.16) and (1.1) that we study are real and odd and hence we can use (1.8) instead of (1.2). Furthermore, the case of collinear points u and v is trivial as then $p_K(0, u, v) \equiv 0$ and thus we can skip it.

We will use the following lemma many times below. Note that it can be easily generalised for another couple of kernels instead of κ_n and κ_N .

Lemma 1.1. Given K_t of the form (1.1),

$$p_{K_t}(0, u, v) = p_{\kappa_N}(0, u, v) + \varphi_{n,N}(0, u, v) t + p_{\kappa_n}(0, u, v) t^2,$$
(1.9)

where

$$\varphi_{n,N}(0,u,v) := \kappa_N(u)(\kappa_n(v) + \kappa_n(u-v)) + \kappa_N(v)(\kappa_n(u) + \kappa_n(v-u)) + \kappa_N(u-v)(\kappa_n(u) - \kappa_n(v)).$$
(1.10)

Proof. We substitute (1.1) into (1.8) and get

$$p_{K_{t}}(0, u, v) = (\kappa_{N}(u) + t \kappa_{n}(u))(\kappa_{N}(v) + t \kappa_{n}(v)) + (\kappa_{N}(u) + t \kappa_{n}(u))(\kappa_{N}(u - v) + t \kappa_{n}(u - v)) + (\kappa_{N}(v) + t \kappa_{n}(v))(\kappa_{N}(v - u) + t \kappa_{n}(v - u)) = \kappa_{N}(u)\kappa_{N}(v) + (\kappa_{N}(u)\kappa_{n}(v) + \kappa_{n}(u)\kappa_{N}(v))t + \kappa_{n}(u)\kappa_{n}(v)t^{2} + \kappa_{N}(u)\kappa_{N}(u - v) + (\kappa_{N}(u)\kappa_{n}(u - v) + \kappa_{n}(u)\kappa_{N}(u - v))t + \kappa_{n}(u)\kappa_{n}(v - u)t^{2} + \kappa_{N}(v)\kappa_{N}(v - u) + (\kappa_{N}(v)\kappa_{n}(v - u) + \kappa_{n}(v)\kappa_{N}(v - u))t + \kappa_{n}(v)\kappa_{n}(v - u)t^{2}.$$

To finish the proof it is enough to group the terms and take into account (1.8).

It is important that the leading coefficient of the quadratic polynomial (1.9) (with respect to t) is always non-negative by (0.17) and (1.2).

From now on, in order to simplify formulas we skip (0, u, v) in permutations and other expressions if there is no confusion. For example, we write p_K instead of $p_K(0, u, v)$. In addition, we use the following notations:

$$\lambda_1 := \frac{\operatorname{Re} u}{|u|}, \qquad \lambda_2 := \frac{\operatorname{Re} v}{|v|}, \qquad \lambda_3 := \frac{\operatorname{Re} (u-v)}{|u-v|}, \qquad \Lambda := \lambda_1 \lambda_2 \lambda_3, \tag{1.11}$$

where the denominators do not vanish as the points u and v are assumed to be distinct and non-collinear. Note that in these terms,

$$p_{\kappa_n} = \frac{(\lambda_1 \lambda_2)^{2n-1}}{|u||v|} + \frac{(\lambda_1 \lambda_3)^{2n-1}}{|u||u-v|} - \frac{(\lambda_2 \lambda_3)^{2n-1}}{|v||v-u|}$$
(1.12)

and

$$\varphi_{n,N} = \frac{\lambda_1^{2N-1}}{|u|} \left(\frac{\lambda_2^{2n-1}}{|v|} + \frac{\lambda_3^{2n-1}}{|u-v|} \right) + \frac{\lambda_2^{2N-1}}{|v|} \left(\frac{\lambda_1^{2n-1}}{|u|} - \frac{\lambda_3^{2n-1}}{|v-u|} \right) + \frac{\lambda_3^{2N-1}}{|u-v|} \left(\frac{\lambda_1^{2n-1}}{|u|} - \frac{\lambda_2^{2n-1}}{|v|} \right).$$
(1.13)

What is more, another representation of $\varphi_{n,N}$ is valid.

Lemma 1.2. In terms of (1.11) it holds that

$$\varphi_{n,N} = \tau_1 p_{\kappa_n} - \tau_2, \tag{1.14}$$

where

$$\tau_1 := \lambda_1^{2(N-n)} + \lambda_2^{2(N-n)} + \lambda_3^{2(N-n)}, \qquad 0 \le \tau_1 \le 3, \tag{1.15}$$

and

$$\tau_2 := \Lambda^{2(N-n)} \left(\frac{(\lambda_1 \lambda_2)^{2(2n-N)-1}}{|u||v|} + \frac{(\lambda_1 \lambda_3)^{2(2n-N)-1}}{|u||u-v|} - \frac{(\lambda_2 \lambda_3)^{2(2n-N)-1}}{|v||v-u|} \right).$$
(1.16)

In particular, $\tau_2 \equiv 0$ if N = 2n.

Proof. Direct multiplication of τ_1 by p_{κ_n} gives

$$\begin{pmatrix} \lambda_1^{2(N-n)} + \lambda_2^{2(N-n)} + \lambda_3^{2(N-n)} \end{pmatrix} \cdot \left(\frac{(\lambda_1 \lambda_2)^{2n-1}}{|u||v|} + \frac{(\lambda_1 \lambda_3)^{2n-1}}{|u||u-v|} - \frac{(\lambda_2 \lambda_3)^{2n-1}}{|v||v-u|} \right)$$

$$= \left(\frac{\lambda_3^{2(N-n)} (\lambda_1 \lambda_2)^{2n-1}}{|u||v|} + \frac{\lambda_2^{2(N-n)} (\lambda_1 \lambda_3)^{2n-1}}{|u||u-v|} - \frac{\lambda_1^{2(N-n)} (\lambda_2 \lambda_3)^{2n-1}}{|v||v-u|} \right)$$

$$+ \frac{\lambda_1^{2N-1}}{|u|} \left(\frac{\lambda_2^{2n-1}}{|v|} + \frac{\lambda_3^{2n-1}}{|u-v|} \right) + \frac{\lambda_2^{2N-1}}{|v|} \left(\frac{\lambda_1^{2n-1}}{|u|} - \frac{\lambda_3^{2n-1}}{|v-u|} \right) + \frac{\lambda_3^{2N-1}}{|u-v|} \left(\frac{\lambda_1^{2n-1}}{|u|} - \frac{\lambda_2^{2n-1}}{|v-u|} \right)$$
which is exactly $\tau_2 + \varphi_{n,N}$ by (1.13) and (1.16).

Lemma 1.3. Given κ_n and κ_N of the form (0.16),

$$\frac{N}{n} \cdot \Lambda^{2(N-n)} \cdot p_{\kappa_n} \leqslant p_{\kappa_N}, \qquad 1 \leqslant n \leqslant N.$$
(1.17)

Note that this inequality for n = 1 was obtained in [CMPT1, Proof of Lemma 2.3]. We will use the following lemma from there in order to prove the general form.

Lemma 1.4 (Proof of Proposition 2.1 in [CMPT1]). One has the representation

$$p_{\kappa_m} = \sum_{k=1}^m \binom{m}{k} \Lambda^{2(m-k)} h_k(u, v),$$

where $h_k(u, v) \ge 0$ and are defined as follows:

$$h_k(u,v) = (|u||v||u-v|)^{-2k} \left((\operatorname{Re} u \operatorname{Re} v)^{2k-1} (\operatorname{Im} (u-v))^{2k} + (\operatorname{Re} u \operatorname{Re} (u-v))^{2k-1} (\operatorname{Im} v)^{2k} + (\operatorname{Re} v \operatorname{Re} (v-u))^{2k-1} (\operatorname{Im} u)^{2k} \right).$$

Proof. Within the settings of Lemma 1.4,

$$\Lambda^{2(N-n)} \frac{p_{\kappa_n}}{p_{\kappa_N}} = \frac{\sum_{k=1}^n \binom{n}{k} \binom{N}{k}^{-1} H_k(u,v)}{\sum_{k=1}^N H_k(u,v)},$$

where $H_k(u, v) := {\binom{N}{k}} \Lambda^{2(N-k)} h_k(u, v) \ge 0$. Furthermore,

$$\binom{n}{k}\binom{N}{k}^{-1} = \frac{n!}{(n-k)!}\frac{(N-k)!}{N!} = \frac{(n-k+1)\cdots n}{(N-k+1)\cdots N} \leqslant \frac{n}{N}, \qquad 1 \leqslant k \leqslant n,$$

and finally

$$\Lambda^{2(N-n)} \frac{p_{\kappa_n}}{p_{\kappa_N}} \leqslant \frac{n}{N} \cdot \frac{\sum_{k=1}^n H_k(u, v)}{\sum_{k=1}^N H_k(u, v)} \leqslant \frac{n}{N}, \qquad n \leqslant N,$$

which is the desired result.

Lemmas 1.1, 1.2 and 1.3 enable us to obtain lower pointwise estimates for the permutations p_{K_t} via the permutations p_{κ_n} for some t. To do so, we will use (1.14) and (1.17) to estimate the coefficients of the quadratic polynomial (1.9). Let us start with the case $n < N \leq 2n.$

Lemma 1.5. Given K_t of the form (1.1) with $n < N \leq 2n$, if

$$t \in \mathbb{R} \setminus \left[-\frac{1}{2} \left(3 + \sqrt{9 - 4\frac{N}{n}} \right), 2 - \frac{N}{n} \right],$$

then $p_{K_t} \ge C(t) \cdot p_{\kappa_n}$ with some C(t) > 0.

Proof. To get the required estimate, we first look at the expression for τ_2 in (1.16) for our case. Since $n < N \leq 2n$, from (1.12) and (1.16) we immediately get

$$\tau_2 = \Lambda^{2(N-n)} \cdot p_{\kappa_{2n-N}}, \qquad 0 \leqslant 2n - N \leqslant n - 1.$$

with $\tau_2 \equiv 0$ if N = 2n. Consequently, by (1.9) and (1.14),

$$p_{K_t} = p_{\kappa_N} + (\tau_1 p_{\kappa_n} - \Lambda^{2(N-n)} p_{\kappa_{2n-N}}) t + p_{\kappa_n} t^2.$$
(1.18)

Now we show that the right hand side of (1.18) for t mentioned in the lemma is bounded from below by p_{κ_n} , multiplied by a positive constant, depending only on t.

Applying the inequality (1.17) to p_{κ_N} and $p_{\kappa_{2n-N}}$ in (1.18) for $t \ge 0$ gives

$$p_{K_t} \ge \left(\frac{N}{n} \Lambda^{2(N-n)} + (\tau_1 - 2 + \frac{N}{n}) t + t^2\right) \cdot p_{\kappa_n} = f(\xi_1, \xi_2, \xi_3) \cdot p_{\kappa_n}, \tag{1.19}$$

where $\xi_j := \lambda_j^{2(N-n)} \in [0, 1], \ j = 1, 2, 3$, and

$$f(\xi_1, \xi_2, \xi_3) := \frac{N}{n} \,\xi_1 \xi_2 \xi_3 + (\xi_1 + \xi_2 + \xi_3 - 2 + \frac{N}{n}) \,t + t^2. \tag{1.20}$$

Analysis of $\partial f / \partial \xi_i$ shows that f is non-decreasing for $t \ge 0$ with respect to each $\xi_i \in [0, 1]$. Consequently,

$$f(\xi_1, \xi_2, \xi_3) \ge f(0, 0, 0) = t(t - 2 + \frac{N}{n}),$$

which is strictly positive for $t > 2 - \frac{N}{n} \ge 0$. For $t \le 0$ we apply (1.17) to p_{κ_N} and use that $p_{\kappa_{2n-N}} \ge 0$ (see (0.17)). This yields

$$p_{K_t} \ge \left(\frac{N}{n} \Lambda^{2(N-n)} + \tau_1 t + t^2\right) \cdot p_{\kappa_n} = F(\xi_1, \xi_2, \xi_3) \cdot p_{\kappa_n}, \tag{1.21}$$

where the function

$$F(\xi_1, \xi_2, \xi_3) := \frac{N}{n} \,\xi_1 \xi_2 \xi_3 + (\xi_1 + \xi_2 + \xi_3) \,t + t^2 \tag{1.22}$$

is non-increasing for $t \leq -\frac{N}{n}$ with respect to each $\xi_j \in [0, 1]$. Consequently,

$$F(\xi_1, \xi_2, \xi_3) \ge F(1, 1, 1) = \frac{N}{n} + 3t + t^2,$$

where the latter expression is positive for $t < -\frac{1}{2}\left(3 + \sqrt{9 - 4\frac{N}{n}}\right) \leq -\frac{N}{n}$.

Now let N > 2n. Note that the following lemma coincides with the previous one if we put N = 2n.

Lemma 1.6. Given K_t of the form (1.1) with N > 2n, if

$$t \in \mathbb{R} \setminus \left[-\frac{1}{2} \left(3 + \rho_{n,N} + \sqrt{(3 + \rho_{n,N})^2 - 4\frac{N}{n}} \right), \rho_{n,N} \right], \quad \rho_{n,N} = \left(\frac{N}{n} - 2 \right) \sqrt{N - 2n},$$

then $p_{K_t} \ge C(t) \cdot p_{\kappa_n}$ with some C(t) > 0.

Proof. We will again estimate the coefficients of the polynomial (1.9) in terms of p_{κ_n} . At first, we will estimate $|\phi_{n,N}|$. By (1.14), this will only need to estimate $|\tau_2|$.

As we have already mentioned before Lemma 1.1, the permutations p_{K_t} and p_{κ_n} are invariant under translations. Therefore we can assume without loss of generality that all triangles (0, u, v) that we consider belong to the half plane $\operatorname{Re} z \ge 0$. This will be necessary in the further analysis of angles of these triangles.

From now on, we use the following notation additionally to (1.11):

$$\sin \alpha_j := \lambda_j, \qquad \lambda_j \in [-1; 1], \qquad j = 1, 2, 3.$$
 (1.23)

We also suppose that λ_j^2 are pairwise distinct. One can get the other case by passage to a limit below. For the geometrical interpretation of α_j see Figures 1.2 and 1.3.

Now we aim to represent τ_2 from (1.16) in terms of the curvature written in the form (0.10). For this purpose we will segregate the area squared $S(0, u, v)^2$ in the numerator and $|u|^2|v|^2|u-v|^2$ in the denominator of τ_2 . First, from (1.16), taking into account (1.11), we obtain

$$\begin{aligned} \tau_2 &= \frac{\Lambda^{2n-1}}{|u||v||u-v|} \frac{(\lambda_1 \lambda_2)^{2(2n-N)-1} |u-v| + (\lambda_1 \lambda_3)^{2(2n-N)-1} |v| - (\lambda_2 \lambda_3)^{2(2n-N)-1} |u|}{\Lambda^{2(2n-N)-1}} \\ &= \frac{\Lambda^{2n-1}}{|u||v||u-v|} \left(\lambda_3^{2(N-2n)} \operatorname{Re} (u-v) + \lambda_2^{2(N-2n)} \operatorname{Re} v - \lambda_1^{2(N-2n)} \operatorname{Re} u \right) \\ &= \frac{\Lambda^{2n-1}}{|u||v||u-v|} \left(\operatorname{Re} u \left(\lambda_3^{2(N-2n)} - \lambda_1^{2(N-2n)} \right) - \operatorname{Re} v \left(\lambda_3^{2(N-2n)} - \lambda_2^{2(N-2n)} \right) \right) \\ &= \frac{\Lambda^{2n-1}}{|u||v||u-v|} \left(\lambda_1 |u| \left(\lambda_3^2 - \lambda_1^2 \right) A_1(u,v) - \lambda_2 |v| \left(\lambda_3^2 - \lambda_2^2 \right) A_2(u,v) \right), \end{aligned}$$

where

$$A_1(u,v) := \frac{\lambda_3^{2(N-2n)} - \lambda_1^{2(N-2n)}}{\lambda_3^2 - \lambda_1^2} \quad \text{and} \quad A_2(u,v) := \frac{\lambda_3^{2(N-2n)} - \lambda_2^{2(N-2n)}}{\lambda_3^2 - \lambda_2^2}.$$
 (1.24)

Finally, we can rewrite τ_2 as

$$\tau_2 = \frac{\Lambda^{2n-1}}{|u|^2 |v|^2 |u-v|^2} \cdot A(u,v), \tag{1.25}$$

where $A(u,v) := |u||v||u-v| \left(\lambda_1 \left(\lambda_3^2 - \lambda_1^2\right) |u|A_1(u,v) - \lambda_2 \left(\lambda_3^2 - \lambda_2^2\right) |v|A_2(u,v)\right)$. By (1.23) and the formulas for the sum of sines and the sine of a double angle,

$$\lambda_3^2 - \lambda_1^2 = (\sin \alpha_3 + \sin \alpha_1)(\sin \alpha_3 - \sin \alpha_1)$$

= $2 \sin \frac{\alpha_3 + \alpha_1}{2} \cos \frac{\alpha_3 - \alpha_1}{2} \cdot 2 \sin \frac{\alpha_3 - \alpha_1}{2} \cos \frac{\alpha_3 + \alpha_1}{2}$
= $\sin(\alpha_3 + \alpha_1) \sin(\alpha_3 - \alpha_1).$

Analogously, $\lambda_3^2 - \lambda_2^2 = \sin(\alpha_3 + \alpha_2)\sin(\alpha_3 - \alpha_2)$. Thus

$$A(u,v) = |u|^{2}|v||u-v|\sin(\alpha_{3}+\alpha_{1})\sin(\alpha_{3}-\alpha_{1})\lambda_{1}A_{1}(u,v) -|u||v|^{2}|u-v|\sin(\alpha_{3}+\alpha_{2})\sin(\alpha_{3}-\alpha_{2})\lambda_{2}A_{2}(u,v).$$
(1.26)



Figure 1.2: Triangles in the case 1 (the proof of Lemma 1.6).

Now let us see how one can calculate the angles $\angle(u, 0, v)$, $\angle(0, u, v)$ and $\angle(0, v, u)$ of the triangle (0, u, v), using the angles α_j , j = 1, 2, 3. Recall that the triangles (0, u, v) belong to the half plane Re $z \ge 0$. Thus only two cases are possible:

1. The vertexes u and v both lie in the same (first or forth) quarter of the plane.

2. The vertexes u and v lie in different quarters of the plane.

One can check that four options are realizable in the case 1 (see the examples in Figure 1.2; several other situations are possible but they produce the same cases):

1a. $\angle (u, 0, v) = \alpha_1 - \alpha_2$, $\angle (0, u, v) = -(\alpha_1 - \alpha_3)$, $\angle (0, v, u) = \pi + (\alpha_2 - \alpha_3)$; 1b. $\angle (u, 0, v) = -(\alpha_1 - \alpha_2)$, $\angle (0, u, v) = \alpha_1 - \alpha_3$, $\angle (0, v, u) = \pi - (\alpha_2 - \alpha_3)$; 1c. $\angle (u, 0, v) = \alpha_1 - \alpha_2$, $\angle (0, u, v) = \pi - (\alpha_1 + \alpha_3)$, $\angle (0, v, u) = \alpha_2 + \alpha_3$; 1d. $\angle (u, 0, v) = -(\alpha_1 - \alpha_2)$, $\angle (0, u, v) = \pi + (\alpha_1 + \alpha_3)$, $\angle (0, v, u) = -(\alpha_2 + \alpha_3)$. In the case 2 (see Figure 1.3) one always has

$$\angle (u, 0, v) = \pi - (\alpha_1 + \alpha_2), \quad \angle (0, u, v) = \alpha_1 - \alpha_3, \quad \angle (0, v, u) = \alpha_2 + \alpha_3.$$

Consequently, taking into account the formulas

$$S(0, u, v) = \frac{1}{2}|u||v|\sin \angle (u, 0, v) = \frac{1}{2}|u||u - v|\sin \angle (0, u, v) = \frac{1}{2}|v||u - v|\sin \angle (0, v, u),$$

we conclude from (1.26) that,



Figure 1.3: Triangles in the case 2 (the proof of Lemma 1.6).

• in the cases 1.a and 1.b:

$$\begin{aligned} A(u,v) &= |u||v|\sin(\pm(\alpha_2 - \alpha_1))|u||u - v|\sin(\pm(\alpha_1 - \alpha_3))\frac{\sin(\alpha_3 + \alpha_1)}{\sin(\alpha_1 - \alpha_2)}\lambda_1A_1(u,v) \\ &- |u||v|\sin(\pm(\alpha_2 - \alpha_1))|v||u - v|\sin(\pi \pm (\alpha_3 - \alpha_2))\frac{\sin(\alpha_3 + \alpha_2)}{\sin(\alpha_1 - \alpha_2)}\lambda_2A_2(u,v) \\ &= 4S(0,u,v)^2 \frac{\sin(\alpha_3 + \alpha_1)\lambda_1A_1(u,v) - \sin(\alpha_3 + \alpha_2)\lambda_2A_2(u,v)}{\sin(\alpha_1 - \alpha_2)}; \end{aligned}$$

• in the cases 1.c and 1.d:

$$\begin{aligned} A(u,v) &= |u||v|\sin(\pm(\alpha_1 - \alpha_2))|u||u - v|\sin(\pi \mp (\alpha_1 + \alpha_3))\frac{\sin(\alpha_3 - \alpha_1)}{\sin(\alpha_1 - \alpha_2)}\lambda_1 A_1(u,v) \\ &- |u||v|\sin(\pm(\alpha_1 - \alpha_2))|v||u - v|\sin(\pm(\alpha_2 + \alpha_3))\frac{\sin(\alpha_3 - \alpha_2)}{\sin(\alpha_1 - \alpha_2)}\lambda_2 A_2(u,v) \\ &= 4S(0,u,v)^2 \frac{\sin(\alpha_3 - \alpha_1)\lambda_1 A_1(u,v) - \sin(\alpha_3 - \alpha_2)\lambda_2 A_2(u,v)}{\sin(\alpha_1 - \alpha_2)}; \end{aligned}$$

• in the case 2:

$$A(u,v) = -|u||v|\sin(\alpha_1 + \alpha_2)|u||u - v|\sin(\alpha_1 - \alpha_3)\frac{\sin(\alpha_3 + \alpha_1)}{\sin(\alpha_1 + \alpha_2)}\lambda_1A_1(u,v) - |u||v|\sin(\alpha_1 + \alpha_2)|v||u - v|\sin(\alpha_2 + \alpha_3)\frac{\sin(\alpha_3 - \alpha_2)}{\sin(\alpha_1 + \alpha_2)}\lambda_2A_2(u,v) = -4S(0, u, v)^2 \frac{\sin(\alpha_3 + \alpha_1)\lambda_1A_1(u, v) + \sin(\alpha_3 - \alpha_2)\lambda_2A_2(u, v)}{\sin(\alpha_1 + \alpha_2)}.$$

Note that the substitutions $\alpha_1 \mapsto -\alpha_1$, $\alpha_2 \mapsto -\alpha_2$ ($\lambda_1 \mapsto -\lambda_1$, $\lambda_2 \mapsto -\lambda_2$) in the expression for A(u, v) for the case 1.a and 1.b give A(u, v) in the cases 1.c and 1.d. Moreover, the substitution $\alpha_2 \mapsto -\alpha_2$ ($\lambda_2 \mapsto -\lambda_2$) in A(u, v) for the case 1.a and 1.b gives -A(u, v) in the case 2. In what follows, this allows us to consider only one expression for A(u, v), say, the one corresponding to 1.a and 1.b, instead of the three. This reduction will not affect the final result. By this reason, let

$$A(u, v) = 4S(0, u, v)^{2} \cdot \frac{V(u, v)}{\sin(\alpha_{1} - \alpha_{2})},$$

where

$$V(u,v) := \sin(\alpha_3 + \alpha_1)\lambda_1 A_1(u,v) - \sin(\alpha_3 + \alpha_2)\lambda_2 A_2(u,v).$$
(1.27)

From this and (1.25) by the formula (0.10), connecting the curvature c(0, u, v) and the area S(0, u, v), we get

$$\tau_2 = \frac{4S(0, u, v)^2}{|u|^2 |v|^2 |u - v|^2} \cdot \Lambda^{2n-1} \cdot \frac{V(u, v)}{\sin(\alpha_1 - \alpha_2)} = \frac{1}{4}c(0, u, v)^2 \cdot \Lambda^{2n-1} \cdot \frac{V(u, v)}{\sin(\alpha_1 - \alpha_2)}.$$

Note that $\frac{1}{4}c(0, u, v)^2 = p_{\kappa_1}(0, u, v)$ by (0.15). Consequently, the inequality (1.17) and the fact that $|\Lambda| \leq 1$ yield

$$|\tau_2| = n \Lambda^{2(n-1)} p_{\kappa_1} \cdot \frac{|\Lambda|}{n} \cdot \frac{|V(u,v)|}{|\sin(\alpha_1 - \alpha_2)|} \leqslant \frac{p_{\kappa_n}}{n} \cdot \frac{|V(u,v)|}{|\sin(\alpha_1 - \alpha_2)|}.$$
 (1.28)

Now we want to show that $|V(u,v)| \leq \text{const} \cdot |\sin(\alpha_1 - \alpha_2)|$. If we rewrite $A_1(u,v)$ and $A_2(u,v)$, defined in (1.24), using the formula

$$\frac{a^m - b^m}{a - b} = \sum_{v=0}^{m-1} a^{m-1-v} b^v, \qquad m \in \mathbb{N}^+,$$

for $m := N - 2n \ge 1$, then (1.27) takes the form

$$V(u,v) = \sum_{v=0}^{m-1} \lambda_3^{2(m-1-v)} \left(\sin(\alpha_3 + \alpha_1) \cdot \lambda_1^{2v+1} - \sin(\alpha_3 + \alpha_2) \cdot \lambda_2^{2v+1} \right).$$

Now we substitute $\lambda_j = \sin \alpha_j$, j = 1, 2, by (1.23) and apply the well-known formula

$$(\sin\theta)^{2\nu+1} = \frac{1}{2^{2\nu}} \sum_{k=0}^{\nu} (-1)^{\nu-k} \binom{2\nu+1}{k} \sin(2\nu+1-2k)\theta.$$

This leads to the following representation:

$$V(u,v) = \sum_{\nu=0}^{m-1} \lambda_3^{2(m-1-\nu)} \frac{1}{2^{2\nu}} \sum_{k=0}^{\nu} (-1)^{\nu-k} \binom{2\nu+1}{k} B_{\nu,k}(\alpha_1,\alpha_2,\alpha_3), \qquad (1.29)$$

where

$$B_{v,k}(\alpha_1, \alpha_2, \alpha_3) := \sin(\alpha_3 + \alpha_1)\sin(2v + 1 - 2k)\alpha_1 - \sin(\alpha_3 + \alpha_2)\sin(2v + 1 - 2k)\alpha_2.$$

By the formulas for the product of sines and the difference of cosines we obtain

$$B_{v,k}(\alpha_1, \alpha_2, \alpha_3) = \frac{1}{2} \left(\cos(\alpha_3 - 2(v - k)\alpha_1) - \cos(\alpha_3 + 2(v - k + 1)\alpha_1) \right) \\ - \cos(\alpha_3 - 2(v - k)\alpha_2) + \cos(\alpha_3 + 2(v - k + 1)\alpha_2) \right) = \frac{1}{2} \sin(\alpha_3 - (v - k)(\alpha_1 + \alpha_2)) \sin((v - k)(\alpha_1 - \alpha_2)) \\ + \sin(\alpha_3 + (v - k + 1)(\alpha_1 + \alpha_2)) \sin((v - k + 1)(\alpha_1 - \alpha_2)).$$

Since $|\sin rx| \leq r |\sin x|, r \geq 0$, it follows that

$$|B_{v,k}(\alpha_1, \alpha_2, \alpha_3)| \leq (2v - 2k + 1)|\sin(\alpha_1 - \alpha_2)|.$$

This and the obvious estimate of |V(u, v)| from (1.29) yield

$$\frac{|V(u,v)|}{|\sin(\alpha_2 - \alpha_1)|} \leqslant \sigma(m) := \sum_{v=0}^{m-1} \frac{1}{2^{2v}} \sum_{k=0}^{v} \binom{2v+1}{k} (2v-2k+1).$$

One can check by successive use of the formulas (4.2.1.6), (4.2.2.13) and (4.2.3.19) from [PBM, §4.2] that

$$\sigma(m) = \frac{4m^2 - 1}{3 \cdot 4^{m-1}} \binom{2m - 2}{m - 1}.$$

Moreover, it can by easily proved by induction that

$$\sigma(m) \leqslant m^{3/2}, \qquad m \in \mathbb{N}.$$

Since $m = N - 2n \ge 1$, (1.28) yields

$$|\tau_2| \leq \rho_{n,N} \cdot p_{\kappa_n}, \qquad \rho_{n,N} = \frac{(N-2n)^{3/2}}{n} = \left(\frac{N}{n} - 2\right)\sqrt{N-2n}, \qquad N > 2n.$$
(1.30)

Now we come back to the representation (1.9) from Lemma 1.1 and estimation of its terms. By (1.14), (1.17) and (1.30), we deduce for $t \ge 0$ that

$$p_{K_t} \ge \left(\frac{N}{n} \Lambda^{2(N-n)} + (\tau_1 - \rho_{n,N}) t + t^2\right) \cdot p_{\kappa_n} = g(\xi_1, \xi_2, \xi_3) \cdot p_{\kappa_n}, \tag{1.31}$$

where $\xi_j = \lambda_j^{2(N-n)} \in [0, 1]$ as in the proof of the previous lemma, and

$$g(\xi_1,\xi_2,\xi_3) := \frac{N}{n} \,\xi_1 \xi_2 \xi_3 + (\xi_1 + \xi_2 + \xi_3 - \rho_{n,N}) \,t + t^2. \tag{1.32}$$

The function g is non-decreasing for $t \ge 0$ with respect to each $\xi_j \in [0,1]$, hence for $t > \rho_{n,N} > 0$ we obtain the inequality

$$g(\xi_1,\xi_2,\xi_3) \ge g(0,0,0) = t(t-\rho_{n,N}) > 0.$$

For $t \leq 0$ we have

$$p_{K_t} \ge \left(\frac{N}{n} \Lambda^{2(N-n)} + (\tau_1 + \rho_{n,N}) t + t^2\right) \cdot p_{\kappa_n} = G(\xi_1, \xi_2, \xi_3) \cdot p_{\kappa_n}, \qquad (1.33)$$

where the function

$$G(\xi_1, \xi_2, \xi_3) := \frac{N}{n} \,\xi_1 \xi_2 \xi_3 + (\xi_1 + \xi_2 + \xi_3 + \rho_{n,N}) \,t + t^2 \tag{1.34}$$

is non-increasing for $t \leq -\frac{N}{n}$ with respect to each $\xi_j \in [0,1]$ and therefore

$$G(\xi_1, \xi_2, \xi_3) \ge G(1, 1, 1) = \frac{N}{n} + (3 + \rho_{n,N}) t + t^2.$$

The roots of the latter quadratic polynomial are

$$-\frac{1}{2}\left(3+\rho_{n,N}\pm\sqrt{(3+\rho_{n,N})^2-4\frac{N}{n}}\right),$$

so it has only positive values if $t<-\frac{1}{2}\left(3+\rho_{n,N}+\sqrt{(3+\rho_{n,N})^2-4\frac{N}{n}}\right)\leqslant-\frac{N}{n}.$

Note that Lemmas 1.5 and 1.6 give Theorem 1.1 by continuity. For the proof of Theorem 1.2, additionally to Theorem 1.1, we will also need lower estimates of p_{K_t} for t, which are the end points of the intervals excluded in (1.5) and (1.6) from the real line. In order to obtain these estimates, we first introduce additional notation.

Given two distinct points $z, w \in \mathbb{C}$, we denote by $L_{z,w}$ the line passing through z and w. Given three pairwise distinct points $z_1, z_2, z_3 \in \mathbb{C}$, we denote by $\angle(z_1, z_2, z_3)$ the smallest angle formed by the lines L_{z_1,z_2} and L_{z_1,z_3} . This angle belongs to $[0; \pi/2]$. If L and L'are lines, then $\angle(L, L')$ is the smallest angle between them. This angle belongs to $[0; \pi/2]$, too. Also, $\theta_V(L) := \angle(L, V)$ and $\theta_H(L) := \angle(L, H)$, where V and H are the vertical and horizontal lines, correspondingly. Furthermore, for a fixed constant $\tau \ge 1$, we set

$$\mathcal{O}_{\tau} = \left\{ (z_1, z_2, z_3) : \frac{|z_i - z_j|}{|z_i - z_k|} \leqslant \tau \text{ for pairwise distinct } i, j, k \in \{1, 2, 3\} \right\},$$
(1.35)

so that all the triangles with vertexes z_1 , z_2 and z_3 in \mathcal{O}_{τ} have comparable sides.

Given $\alpha_0 \in (0, \pi/2)$ and (z_1, z_2, z_3) , in what follows we will use the conditions

$$\theta_V(L_{z_1, z_2}) + \theta_V(L_{z_2, z_3}) + \theta_V(L_{z_1, z_3}) \ge \alpha_0 \tag{1.36}$$

and

$$\theta_H(L_{z_1, z_2}) + \theta_H(L_{z_2, z_3}) + \theta_H(L_{z_1, z_3}) \ge \alpha_0.$$
(1.37)

Note that (1.36) and (1.37) can be correspondingly replaced by the conditions

$$\theta_H(L_{z_1, z_2}) + \theta_H(L_{z_2, z_3}) + \theta_H(L_{z_1, z_3}) \leqslant \frac{3}{2}\pi - \alpha_0$$

and

$$\theta_V(L_{z_1, z_2}) + \theta_V(L_{z_2, z_3}) + \theta_V(L_{z_1, z_3}) \leqslant \frac{3}{2}\pi - \alpha_0.$$

To obtain the desired result, we first prove several geometrical lemmas.

Lemma 1.7. Fix $\alpha_0 \in (0, \pi/2)$. Given $(0, u, v) \in \mathcal{O}_{\tau}$, if the condition (1.36) is satisfied, then

$$\tau_1(0, u, v) = \lambda_1^{2(N-n)} + \lambda_2^{2(N-n)} + \lambda_3^{2(N-n)} \ge C_1(\alpha_0) > 0.$$

Proof. Clearly,

$$\lambda_1^2 = \sin^2 \theta_V(L_{0,u}), \quad \lambda_2^2 = \sin^2 \theta_V(L_{0,v}), \quad \lambda_3^2 = \sin^2 \theta_V(L_{u,v}),$$

Moreover, from (1.36) it follows that at least one of the angles $\theta_V(L_{0,u})$, $\theta_V(L_{0,v})$, $\theta_V(L_{u,v})$ is not less than $\alpha_0/3$. Thus $\tau_1 \ge (\sin \frac{\alpha_0}{3})^{2(N-n)}$.



Figure 1.4: Triangles in the case 2 (the proof of Lemma 1.8).

Lemma 1.8. Fix $\alpha_0 \in (0, \pi/2)$. Given $(0, u, v) \in \mathcal{O}_{\tau}$, if the condition (1.37) is satisfied, then

$$\Upsilon(0, u, v) := 2 + (\lambda_1 \lambda_2 \lambda_3)^{2(N-n)} - (\lambda_1^{2(N-n)} + \lambda_2^{2(N-n)} + \lambda_3^{2(N-n)}) \ge C_2(\alpha_0, \tau) > 0.$$

Proof. First we note that

$$\Upsilon(0,u,v) \geqslant 2 + \lambda_1^2 \lambda_2^2 \lambda_3^2 - (\lambda_1^2 + \lambda_2^2 + \lambda_3^2)$$

as the function $2 + \xi_1 \xi_2 \xi_3 - (\xi_1 + \xi_2 + \xi_3)$ is non-increasing with respect to each $\xi_j \in [0, 1]$, j = 1, 2, 3, and $\lambda_j^2 \ge \lambda_j^{2(N-n)}$ for $N > n \ge 1$.

In addition, we have

$$\lambda_1^2 = 1 - \sin^2 \theta_H(L_{0,u}), \quad \lambda_2^2 = 1 - \sin^2 \theta_H(L_{0,v}), \quad \lambda_3^2 = 1 - \sin^2 \theta_H(L_{u,v}),$$

and hence

$$\begin{split} \Upsilon(0, u, v) &\geq \sin^2 \theta_H(L_{0,u}) \sin^2 \theta_H(L_{0,v}) + \sin^2 \theta_H(L_{0,u}) \sin^2 \theta_H(L_{u,v}) \\ &+ \sin^2 \theta_H(L_{0,v}) \sin^2 \theta_H(L_{u,v}) - \sin^2 \theta_H(L_{0,u}) \sin^2 \theta_H(L_{0,v}) \sin^2 \theta_H(L_{u,v}) \\ &\geq \frac{2}{3} \left(\sin^2 \theta_H(L_{0,v}) \sin^2 \theta_H(L_{0,u}) + \sin^2 \theta_H(L_{0,u}) \sin^2 \theta_H(L_{u,v}) \\ &+ \sin^2 \theta_H(L_{0,v}) \sin^2 \theta_H(L_{u,v}) \right). \end{split}$$

Consider a triangle $(0, u, v) \in \mathcal{O}_{\tau}$ such that (1.37) is satisfied. Fix some $\varepsilon \in (0; \alpha_0/3)$. Two cases are possible:

- 1. amongst $\theta_H(L_{0,u})$, $\theta_H(L_{0,v})$, $\theta_H(L_{u,v})$, there exists a pair of angles, each being greater than ε and then it is easily seen that $\Upsilon(0, u, v) \ge \frac{2}{3} \sin^4 \varepsilon$;
- 2. amongst those, there exists no pair of angles, each being greater than ε .

Let us consider the second case in detail (see Figure 1.4). It is clear that at least two angles amongst $\theta_H(L_{0,u})$, $\theta_H(L_{0,v})$, $\theta_H(L_{u,v})$ are less than ε then. In other words, two sides of the triangle cut the horizontal line at angles less than ε . We call these sides A and B.

Furthermore, let the angle γ between A and B be acute; then obviously it is smaller than 2ε . Then the acute angle between the third side C and the horizontal line is greater than $\alpha_0 - 2\varepsilon$ and the acute angle between A and C is greater than $\alpha_0 - 3\varepsilon$. Consequently, the obtuse angle between A and C is smaller than $\pi - (\alpha_0 - 3\varepsilon)$. Thus we have for the angle β of the triangle:

$$\alpha_0 - 3\varepsilon < \beta < \pi - (\alpha_0 - 3\varepsilon).$$

Therefore by the law of sines, the inequalities $(2/\pi) x \leq \sin x \leq x$ for $x \in [0, \pi/2]$, and (1.35), we get

$$\frac{1}{\tau^2} \leqslant \frac{\operatorname{length}(C)}{\operatorname{length}(B)} = \frac{\sin \gamma}{\sin \beta} < \frac{\sin 2\varepsilon}{\sin(\alpha_0 - 3\varepsilon)} \leqslant \frac{\pi\varepsilon}{\alpha_0 - 3\varepsilon} \Rightarrow \varepsilon > \varepsilon_0(\alpha_0, \tau) := \frac{\alpha_0}{3 + \pi\tau^2}.$$

Now let the angle γ between A and B be not acute (it is greater than $\pi - 2\varepsilon$). Then for one of acute angles of the triangle, say β , we have

$$\beta < \varepsilon - (\alpha_0 - 2\varepsilon) = 3\varepsilon - \alpha_0 < 0, \qquad \varepsilon \in (0; \alpha_0/3),$$

which is impossible.

It follows from the aforesaid that there is a contradiction for $\varepsilon = \varepsilon_0(\alpha_0, \tau)$ in the second case and thus $\Upsilon(0, u, v) \ge \frac{2}{3} \sin^4 \varepsilon_0(\alpha_0, \tau)$.

We will also need the following result.

Lemma 1.9 (Lemma 2.3 in [CMPT1]). Fix $\alpha_0 \in (0, \pi/2)$. Given κ_n of the form (0.16) and $(z_1, z_2, z_3) \in \mathcal{O}_{\tau}$, if the condition (1.36) is satisfied, then

$$p_{\kappa_n}(z_1, z_2, z_3) \ge C_3(\alpha_0, \tau) \cdot c(z_1, z_2, z_3)^2$$

for some $C_3(\alpha_0, \tau) > 0$.

Now we are able to obtain necessary lower pointwise estimates for p_{K_t} if t are the end points of the intervals excluded in (1.5) and (1.6) from the real line. Recall that $\rho_{n,N} = \left(\frac{N}{n} - 2\right)\sqrt{N - 2n}.$

Lemma 1.10. Fix $\alpha_0 \in (0, \pi/2)$. Given K_t of the form (1.1) and $(z_1, z_2, z_3) \in \mathcal{O}_{\tau}$,

(i) if (1.36) is satisfied and $t = 2 - \frac{N}{n}$ for $n < N \leq 2n$ or $t = \rho_{n,N}$ for $N \ge 2n$, or

(ii) if (1.36) and (1.37) are satisfied and
$$t = -\frac{1}{2} \left(3 + \sqrt{9 - 4\frac{N}{n}} \right)$$
 for $n < N \leq 2n$ or
 $t = -\frac{1}{2} \left(3 + \rho_{n,N} + \sqrt{(3 + \rho_{n,N})^2 - 4\frac{N}{n}} \right)$ for $N \ge 2n$,

then

$$p_{K_t}(z_1, z_2, z_3) \ge C(\alpha_0, \tau) \cdot c(z_1, z_2, z_3)^2$$

for some $C(\alpha_0, \tau) > 0$.

Proof. It is enough to prove it for triples (0, u, v). What is more, the statement for t = 0in (i), i.e. when N = 2n, is just Lemma 1.9 and therefore we may exclude it. We also recall the notation $\xi_j := \lambda_j^{2(N-n)} \in [0,1], j = 1,2,3.$ Now let $t = 2 - \frac{N}{n}$ and n < N < 2n. Then for the function given in (1.20) we have

$$f(\xi_1,\xi_2,\xi_3) = \frac{N}{n} \,\xi_1\xi_2\xi_3 + (2-\frac{N}{n})(\xi_1+\xi_2+\xi_3) \ge (2-\frac{N}{n})\tau_1$$

where τ_1 is as in (1.15). If $t = \rho_{n,N}$ and N > 2n, then from (1.32) it follows that

$$g(\xi_1,\xi_2,\xi_3) = \frac{N}{n} \,\xi_1\xi_2\xi_3 + \rho_{n,N}(\xi_1 + \xi_2 + \xi_3) \ge \frac{N}{n}\tau_1.$$

Thus from the inequalities (1.19) for n < N < 2n and (1.31) for N > 2n, both being valid for $t \ge 0$, and Lemmas 1.7 and 1.9 (with the assumption (1.36)) we get

$$p_{K_t}(0, u, v) \ge \frac{N}{n} \tau_1(0, u, v) \cdot p_{\kappa_n}(0, u, v) \ge \frac{N}{n} C_1(\alpha_0) C_3(\alpha_0, \tau) \cdot c(0, u, v)^2,$$

which is the required result in the case (i).

Now consider (*ii*). Set $t = -\frac{1}{2}\left(3 + \sqrt{9 - 4\frac{N}{n}}\right)$ and $n < N \leq 2n$. Then, by (1.22),

$$F(\xi_1,\xi_2,\xi_3) = \frac{N}{n}(\xi_1\xi_2\xi_3 - 1) + \frac{1}{2}(3 - (\xi_1 + \xi_2 + \xi_3))(3 + \sqrt{9 - 4\frac{N}{n}}).$$

Since $-\frac{1}{2}\left(3+\sqrt{9-4\frac{N}{n}}\right) \leqslant -\frac{N}{n}$,

$$F(\xi_1, \xi_2, \xi_3) \ge \frac{N}{n} (2 + \xi_1 \xi_2 \xi_3 - (\xi_1 + \xi_2 + \xi_3)) \ge \frac{N}{n} \Upsilon.$$

The function (1.34) for

$$t = t_0 := -\frac{1}{2} \left(3 + \rho_{n,N} + \sqrt{(3 + \rho_{n,N})^2 - 4\frac{N}{n}} \right)$$

and $N \ge 2n$ after some simplifications takes the form

$$G(\xi_1,\xi_2,\xi_3) = \frac{N}{n}(\xi_1\xi_2\xi_3 - 1) + (\xi_1 + \xi_2 + \xi_3 - 3)t_0,$$

and hence $G(\xi_1, \xi_2, \xi_3) \ge \frac{N}{n} \Upsilon$ since $t_0 \le -\frac{N}{n}$ for $N \ge 2n$.

Thus for the last two values of t, by the inequalities (1.21) for $n < N \leq 2n$ and (1.33) for $N \geq 2n$, both being valid for $t \leq -\frac{N}{n}$, and Lemma 1.9 (with the assumption (1.36)), we get

$$p_{K_t}(0, u, v) \ge \frac{N}{n} \Upsilon(0, u, v) \cdot p_{\kappa_n}(0, u, v) \ge \frac{N}{n} \Upsilon(0, u, v) C_3(\alpha_0, \tau) \cdot c(0, u, v)^2.$$

If (1.37) is also satisfied, then by Lemma 1.8 we obtain the desired inequality

$$p_{K_t}(0, u, v) \ge \frac{N}{n} C_2(\alpha_0, \tau) C_3(\alpha_0, \tau) \cdot c(0, u, v)^2,$$

and we are done.

1.4 Examples

In this section we present triples (0, u, v) such that the permutations $p_{K_t}(0, u, v)$ change sign for t mentioned in (1.7), namely, $t \in (-N/n, 0)$ for all n and N and $t \in \left(0, \frac{2}{e^{3/2}} \frac{N}{n}\right)$ for $N \gg n$. We use the notations of Lemma 1.1 below. Note that by this lemma, p_{K_t} can be calculated via p_{κ_m} for m, equal to n and N, and $\varphi_{n,N}$. To obtain p_{κ_m} we substitute (0.16) into (1.8); $\varphi_{n,N}$ is calculated by (1.10).

We first show that $p_{K_t}(0, u, v)$ is positive for any t if u = a + i and v = a - i, where $a \in \mathbb{R} \setminus \{0\}$ is suitably chosen. By (1.8) and taking into account that $K_t(u) = K_t(v)$ and $K_t(u - v) \equiv 0$,

$$p_{K_t} = \left(\frac{a^{2N-1}}{(1+a^2)^N} + t \cdot \frac{a^{2n-1}}{(1+a^2)^n}\right)^2 = \frac{a^{4n-2}}{(1+a^2)^{2n}} \left(\frac{a^{2(N-n)}}{(1+a^2)^{N-n}} + t\right)^2,$$

which is positive for any real t if a is chosen so that the expression in the latter brackets does not vanish.

Now the aim is to show that for any fixed t from (1.7) there exist triples (0, u, v) such that $p_{K_t}(0, u, v)$ is negative. To do so, we find families of (0, u, v) such that the quadratic polynomial

$$\frac{p_{K_t}}{p_{\kappa_n}} = \frac{p_{\kappa_N}}{p_{\kappa_n}} + \frac{\varphi_{n,N}}{p_{\kappa_n}}t + t^2, \qquad p_{\kappa_n} > 0,$$

(with respect to t) has two different roots t_1 and t_2 , depending on u and v, and thus p_{K_t} (of the form (1.9)) is negative for $t \in (t_1; t_2)$. In addition, we prove that the union of the intervals $(t_1; t_2)$ when (0, u, v) runs the whole above-mentioned family is either the interval (-N/n; 0) or $\left(0; \frac{2}{e^{3/2}} \frac{N}{n}\right)$, indicated in (1.7).

Let us consider the case t < 0.

Example 1.1. Set u = -a + i, v = a + i, where $a \in \mathbb{R} \setminus \{0\}$. Then

$$p_{\kappa_m} = -\frac{a^{2(2m-1)}}{(a^2+1)^{2m}} + \frac{a^{2m-2}}{2(a^2+1)^m} + \frac{a^{2m-2}}{2(a^2+1)^m} = \frac{a^{2m-2}\left((a^2+1)^m - a^{2m}\right)}{(a^2+1)^{2m}},$$

where m equals n or N, and

$$\begin{split} \varphi_{n,N} &= -\frac{a^{2N-1}}{(a^2+1)^N} \left(\frac{a^{2n-1}}{(a^2+1)^n} - \frac{1}{2a} \right) \\ &+ \frac{a^{2N-1}}{(a^2+1)^N} \left(-\frac{a^{2n-1}}{(a^2+1)^n} + \frac{1}{2a} \right) - \frac{1}{2a} \left(-\frac{a^{2n-1}}{(a^2+1)^n} - \frac{a^{2n-1}}{(a^2+1)^n} \right) \\ &= \frac{a^{2n-2} \left((a^2+1)^N + (a^2+1)^n a^{2(N-n)} - 2a^{2N} \right)}{(a^2+1)^{N+n}}. \end{split}$$

From this by (1.9) we deduce that

$$\frac{p_{K_t}}{p_{\kappa_n}} = \frac{d_1(a)d_2(a)}{d_3(a)^2} + \frac{d_1(a) + d_2(a)}{d_3(a)}t + t^2, \qquad a \neq 0,$$
(1.38)

where

$$d_1(a) := (a^2 + 1)^n \left(1 - \frac{a^{2N}}{(a^2 + 1)^N} \right), \quad d_2(a) := (a^2 + 1)^n \left(\frac{a^{2(N-n)}}{(a^2 + 1)^{N-n}} - \frac{a^{2N}}{(a^2 + 1)^N} \right)$$

and $d_3(a) := (a^2 + 1)^n - a^{2n}$. The polynomial (1.38) has two different negative roots $t_1(a) = -d_1(a)/d_3(a)$ and $t_2(a) = -d_2(a)/d_3(a)$, where $d_1(a) > d_2(a) > 0$ and $d_3(a) > 0$. It is easy to check that the roots $t_1(a)$ and $t_2(a)$ run the intervals (-N/n; -1) and (-1; 0), correspondingly, when a runs $(0; \infty)$. Furthermore, we see by continuity that

$$\bigcup_{a \in (0;\infty)} (t_1(a); t_2(a)) = (-N/n; 0).$$

This means that p_{K_t} is negative for any t in (-N/n; 0) for a suitably chosen.

As we have already mentioned above, this example shows that (1.6) is sharp for N = 2n in the sense of Remark 1.1. In addition, the left hand side of (1.6) is also sharp for N = 2n + 1.

Now we give the example for t > 0.

Example 1.2. Let n be fixed. Consider the triples (0, u, v) such that

$$u = -r(1 + \delta_N(q)i), \qquad v = r(1 - \frac{1}{r} + \delta_N(q)i), \qquad \delta_N(q) := \sqrt{\frac{\ln q}{N - n}}, \qquad (1.39)$$

where r > 0 and $q \ge e$. We can also calculate p_{κ_m} and $\varphi_{n,N}$ for these (0, u, v) using (0.16), (1.8) and (1.10). However, the expression of p_{K_t}/p_{κ_n} obtained is too big and therefore we do not place it here. Instead, we give the following identity (the permutations are calculated for (0, u, v) as in (1.39)):

$$P(t) := \lim_{r \to \infty} \frac{p_{K_t}}{p_{\kappa_n}} = c_N(q) + b_N(q)t + t^2,$$

where

$$c_N(q) = \frac{N}{n \left(1 + \delta_N^2(q)\right)^{2(N-n)}} \cdot \frac{(2N-1)\delta_N^2(q) + 1}{(2n-1)\delta_N^2(q) + 1}$$

and

$$b_N(q) = -\frac{\left(2(N-n)^2 + N - 4nN + n\right)\delta_N^2(q) - (n+N)}{n\left(1 + \delta_N^2(q)\right)^{N-n}\left((2n-1)\delta_N^2(q) + 1\right)}.$$

Note that

$$c_N(q) \sim \frac{(2\ln q + 1)N}{q^2 n}, \qquad b_N(q) \sim -\frac{(2\ln q - 1)N}{qn}, \qquad N \to \infty.$$

The quadratic polynomial P (with respect to t) has two different positive roots $t_1(N,q)$ and $t_2(N,q)$ if N is large enough (as the discriminant is positive). Additionally, one can check that

$$t_1(N,q) \sim \tilde{t}_1(q) := \frac{2\ln q + 1}{q \left(2\ln q - 1\right)}, \quad t_2(N,q) \sim \tilde{t}_2(N,q) := \frac{(2\ln q - 1)N}{qn}, \quad N \to \infty.$$

Taking into account the properties

$$\tilde{t}_1(q) \to 0 \text{ as } q \to \infty \qquad \text{and} \qquad \max_{q \in [e;\infty)} \frac{2\ln q - 1}{q} = \frac{2}{e^{3/2}},$$

we deduce by continuity that

$$\bigcup_{q\in[e;\,\infty)}(\tilde{t}_1(q);\tilde{t}_2(q))=\left(0;\frac{2}{e^{3/2}}\frac{N}{n}\right).$$

Thus, $p_{K_t}(0, u, v)$ with u and v as in (1.39) are negative for any t in $\left(0; \frac{2}{e^{3/2}} \frac{N}{n}\right)$, if N (with respect to n) and r are large enough and q is suitably chosen.

1.5 Proof of Theorems 1.1 and 1.2

Recall that Lemmas 1.5 and 1.6 state that if K_t is of the form (1.1) and

$$t \in \mathbb{R} \setminus \left[-\frac{1}{2} \left(3 + \sqrt{9 - 4\frac{N}{n}} \right), 2 - \frac{N}{n} \right], \qquad n < N \leq 2n$$

$$t \in \mathbb{R} \setminus \left[-\frac{1}{2} \left(3 + \rho_{n,N} + \sqrt{(3 + \rho_{n,N})^2 - 4\frac{N}{n}} \right), \rho_{n,N} \right], \qquad N \ge 2n,$$

where $\rho_{n,N} = \left(\frac{N}{n} - 2\right)\sqrt{N - 2n}$, then

$$p_{K_t} \ge C(t) \cdot p_{\kappa_n}, \qquad C(t) > 0.$$

These lemmas immediately give Theorem 1.1 by continuity if we take into account the fact that $p_{\kappa_n}(z_1, z_2, z_2) \ge 0$ for all $(z_1, z_2, z_3) \in \mathbb{C}^3$ (see (0.17)).

What is said from now on is related to Theorem 1.2.

First of all we note that the latter statement of Theorem 1.2, i.e. the one asserting that if the operator T_{K_t} is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded, then E is rectifiable, is a corollary of the fact that the $L^2(\mathcal{H}^1 \lfloor E)$ -boundedness of T_{K_t} implies that $p_{K_t}(\mathcal{H}^1 \lfloor E) < \infty$. This follows from the generalised Melnikov-Verdera identity (0.13) for the kernels K_t and permutations p_{K_t} .

Now we come to the proof of the former statement in Theorem 1.2.

The proof for t, mentioned in Lemmas 1.5 and 1.6 (see also the beginning of the current section), is direct via Theorem D, which states that if $p_{\kappa_n}(\mathcal{H}^1[E) < \infty$, then E is rectifiable. Indeed, if $p_{K_t}(\mathcal{H}^1 \lfloor E) < \infty$ for such t, then $p_{\kappa_n}(\mathcal{H}^1 \lfloor E) < \infty$ by the inequality $p_{K_t} \ge C(t) \cdot p_{\kappa_n}, C(t) > 0$, and thus the set E is rectifiable.

What is left is to prove the former statement in Theorem 1.2 for

$$t = -\frac{1}{2} \left(3 + \sqrt{9 - 4\frac{N}{n}} \right), \qquad t = 2 - \frac{N}{n}, \qquad n < N \le 2n,$$
$$t = -\frac{1}{2} \left(3 + \rho_{n,N} + \sqrt{(3 + \rho_{n,N})^2 - 4\frac{N}{n}} \right), \qquad t = \rho_{n,N}, \qquad N \ge 2n.$$

It requires some additional work and therefore for the reader's convenience we first make several observations, which could help to clarify the forthcoming proof.

We start with a very brief exposition of the proof of Theorem B given in [Leg] (note that one can find a modified version of the proof from [Leg] in [Tol5, Chapter 7] and follow that instead). Recall that Theorem B states that for a 1-set E, if $c^2(\mathcal{H}^1|E) < \infty$, then E is rectifiable. We emphasize again that it is essential in the proof that the curvature is non-negative.

The first step is to show that there exists a compact subset F of the given set E such that, among other things, $c^2(\mathcal{H}^1|F)$ is well-controlled and can be made very small (this is done in [Leg] by a quite standard uniformization procedure). Then the second and most important step follows: to prove that if μ is a measure satisfying a few special conditions (see Proposition 1.2), then there exists a Lipschitz graph Γ such that $\mu(\Gamma) \ge C \cdot \mu(\mathbb{C})$, where C is an absolute constant.

The problem is to choose an adequate coordinate system of $\mathbb C$ and construct a Lipschitz function A whose graph will be the one needed. For this purpose, the author of [Leg] first defines some functions used to measure how well the spt μ is approximated by straight lines at a given location and a given scale. It is shown that these functions are related to the $c^2(\mu)$ in the case when the measure μ does not degenerate too much. These preliminary results are then used to construct the function A by stopping time arguments, which demand fine adjustments to many parameters and thresholds. Starting with choosing a point $x_0 \in \operatorname{spt} \mu$ and fixing an approximating line D_0 (which will be the domain of the function A) such that the mean distance from spt μ to the line D_0 is suitably small, the author of [Leg] comes to cutting spt μ in four disjoint pieces Z, F_1 , F_2 and F_3 such that

$$\operatorname{spt} \mu = Z \cup F_1 \cup F_2 \cup F_3.$$

It is shown that Z is very nice for constructing the graph but the three others admit "bad events". Then the goal is to prove that these bad pieces carry only a small part of the measure μ , namely, $\mu(F_j) \leq 10^{-6}\mu(\mathbb{C})$ for each j and thus $\mu(Z) \geq \frac{99}{100}\mu(\mathbb{C})$. This allows to construct the required Lipschitz function $A: D_0 \to D_0^{\perp}$ such that the set Z is contained in the graph of A.

Coming back to the initial settings, if $\mu = \mathcal{H}^1[F]$, where F is the above-mentioned subset of E, then there exists a Lipschitz graph Γ such that $\mathcal{H}^1(\Gamma \cap F) \ge C \cdot \mathcal{H}^1(F)$. This fact is used in the last step of the proof from [Leg], which is as follows. Since $\mathcal{H}^1(E) < \infty$ by the assumptions, the set E can be decomposed into a rectifiable and purely unrectifiable part, i.e. $E = E_{\text{rect}} + E_{\text{unrect}}$. Suppose that

$$\mathcal{H}^1(E_{\text{unrect}}) > 0. \tag{1.40}$$

Then there exists a compact set $F \subset E_{\text{unrect}}$ and Lipschitz graph Γ such that $\mathcal{H}^1(\Gamma \cap F) \geq C \cdot \mathcal{H}^1(F)$ that contradicts the fact that F is purely unrectifiable.

Let us now say a few words about the proof of Theorem D given in [CMPT1]. Recall that this theorem is an analogue of Theorem B, where the kernel 1/z and curvature squared $c^2(\mathcal{H}^1 \lfloor E)$ are replaced by the kernels $\kappa_n(z) = (\operatorname{Re} z)^{2n-1}/|z|^{2n}$, $n \in \mathbb{N}$, and corresponding permutations p_{κ_n} . We will use the definitions given near the formula (1.35) and in the discussion of Theorem B above. First we mention that it is proved in [CMPT1] that the permutations $p_{\kappa_n}(z_1, z_2, z_3)$ behave similarly to $c^2(z_1, z_2, z_3)$ for all triangles with comparable sides, whose one side makes a big angle with the vertical line. More precisely (see Lemma 1.9), it is shown there that for a fixed $\alpha_0 \in (0, \pi/2)$ and given $(z_1, z_2, z_3) \in \mathcal{O}_{\tau}$, if the condition (1.36), i.e.

$$\theta_V(L_{z_1,z_2}) + \theta_V(L_{z_2,z_3}) + \theta_V(L_{z_1,z_3}) \ge \alpha_{0,z_1}$$

or

$$\theta_H(L_{z_1,z_2}) + \theta_H(L_{z_2,z_3}) + \theta_H(L_{z_1,z_3}) \leqslant \frac{3}{2}\pi - \alpha_0,$$

is satisfied, then

$$p_{\kappa_n}(z_1, z_2, z_3) \ge C(\alpha_0, \tau) \cdot c(z_1, z_2, z_3)^2, \qquad C(\alpha_0, \tau) > 0.$$

This enables the authors of [CMPT1] to use the above-described scheme from [Leg] in order to construct the required Lipschitz graph Γ in the case when the first approximating line D_0 for spt μ is far from the vertical line. Note that exchanging the curvature for the permutations p_{κ_n} still requires new arguments in several key points of the proof. Otherwise, when D_0 is close to the vertical line and the scheme from [Leg] does not work (as $\mu(F_3)$ may be too big), they tune thresholds and apply some coverings so that they can use the result for D_0 , being far from the vertical line, to construct countably many Lipschitz graphs, which give Γ after appropriate joining.

We are now at the position to finish the proof of our Theorem 1.2. This will be an adaptation of the arguments from [CMPT1].

On the one hand, by the clause (i) of Lemma 1.10, for a fixed $\alpha_0 \in (0, \pi/2)$ and given $(z_1, z_2, z_3) \in \mathcal{O}_{\tau}$, if the condition (1.36), i.e. the same as in the result for t = 0from [CMPT1] mentioned above, is satisfied and $t = 2 - \frac{N}{n}$ for $n < N \leq 2n$ or $t = \rho_{n,N}$ for $N \geq 2n$, then we also have

$$p_{K_t}(z_1, z_2, z_3) \ge C(\alpha_0, \tau) \cdot c(z_1, z_2, z_3)^2, \qquad C(\alpha_0, \tau) > 0.$$
 (1.41)

It means that we can undeviatingly follow the scheme from [CMPT1] (exchanging p_{κ_n} for p_{K_t}) in order to get our result for $t = 2 - \frac{N}{n}$, $n < N \leq 2n$, and $t = \rho_{n,N}$, $N \geq 2n$.

On the other hand, by the clause (ii) of Lemma 1.10, the inequality (1.41) is true for

$$t = -\frac{1}{2} \left(3 + \sqrt{9 - 4\frac{N}{n}} \right), \qquad n < N \leq 2n,$$

$$t = -\frac{1}{2} \left(3 + \rho_{n,N} + \sqrt{(3 + \rho_{n,N})^2 - 4\frac{N}{n}} \right), \quad N \ge 2n,$$
 (1.42)

only if both the conditions (1.36) and (1.37) are satisfied, i.e.

$$\alpha_0 \leqslant \theta_V(L_{z_1, z_2}) + \theta_V(L_{z_2, z_3}) + \theta_V(L_{z_1, z_3}) \leqslant \frac{3}{2}\pi - \alpha_0,$$

or

$$\alpha_0 \leqslant \theta_H(L_{z_1, z_2}) + \theta_H(L_{z_2, z_3}) + \theta_H(L_{z_1, z_3}) \leqslant \frac{3}{2}\pi - \alpha_0,$$

and thus the triangles (z_1, z_2, z_3) are far from both the vertical and horizontal line.

Consequently, the scheme from [CMPT1] cannot be applied directly for t from (1.42). However, as we will see, it works after a few modifications (besides the exchange of p_{κ_n} for p_{K_t}) connected basically with adapting geometrical arguments to both the conditions (1.36) and (1.37). Since the cases where we are close to either the vertical or horizontal line are well-separated and similar geometrically, the arguments for the first approximating line D_0 , being close (far) to (from) the vertical line, can be easily transferred into the ones for D_0 , being close (far) to (from) the horizontal line.

We now reproduce the main steps of the proof, stemming from [CMPT1], with necessary changes when our permutations and the conditions (1.36) and (1.37) are involved. Several statements are formulated without proofs because they are the same as in [CMPT1] (or [Leg]) modulus the permutations involved.

Below we consider only t from (1.42). The following two propositions will then imply Theorem 1.2 by the same contradiction arguments as in the proof from [Leg] (see the arguments around (1.40) above). Note that one has to take $\mu = 40\mathcal{H}^1 \lfloor F$ in Proposition 1.2, where the set F is from Proposition 1.1 (it may be suitably rescaled if necessary).

Proposition 1.1 (An analogue of Lemma 3.4 in [CMPT1] and Proposition 1.1 in [Leg]). Let E be a 1-set and $p_{K_t}(\mathcal{H}^1 \lfloor E) < \infty$. Then for all $\eta > 0$ there is a set $F \subset E$ such that

- F is compact,
- $p_{K_t}(\mathcal{H}^1 \lfloor F) \leq \eta \operatorname{diam} F$,
- $\mathcal{H}^1(F) > \frac{1}{40} \operatorname{diam} F$,
- for all $z \in F$, for all r > 0, $\mathcal{H}^1(F \cap B(z, r)) \leq 3r$.

Proposition 1.2. For any $C_0 \ge 10$, there is $\eta > 0$ such that if a measure μ satisfies

- $\mu(B(0,1)) \ge 1$, $\mu(\mathbb{C} \setminus B(0,2)) = 0$,
- for any ball B, $\mu(B) \leq C_0 \operatorname{diam} B$,
- $p_{K_t}(\mu) \leq \eta$,

then there exists a Lipschitz graph Γ such that $\mu(\Gamma) \ge 10^{-5}\mu(\mathbb{C})$.
The rest of the section is devoted to the proof of Proposition 1.2 which is an analogue of [CMPT1, Proposition 3.1]. First, we give several definitions that will be needed below, see [CMPT1, Leg] for further ones. Given a measure μ , set

$$p_{K_t,\tau}(\mu) := \iiint_{\mathcal{O}_{\tau}} p_{K_t}(z_1, z_2, z_3) d\mu(z_1) d\mu(z_2) d\mu(z_3),$$

see (0.6), (1.2) and (1.35). For a ball B = B(x, r) set

$$\delta_{\mu}(x,r) := \frac{\mu(B(x,r))}{r}.$$

We will use a small density threshold $\delta > 0$ for this quantity.

Given a fixed k > 1, for any ball $B = B(x, r) \subset \mathbb{C}$ and D a line in \mathbb{C} , set

$$\begin{split} \beta_{1,\mu}^D(x,r) &:= \frac{1}{r} \int_{B(x,kr)} \frac{\operatorname{dist}\,(y,D)}{r} \, d\mu(y), \\ \beta_{2,\mu}^D(x,r) &:= \left(\frac{1}{r} \int_{B(x,kr)} \left(\frac{\operatorname{dist}\,(y,D)}{r}\right)^2 \, d\mu(y) \right)^{1/2} \end{split}$$

Geometrical notation connected with lines and angles is given near the formula (1.35) at the end of Section 1.3.

Lemma 1.11. Let μ be a measure with C_0 -linear growth and $B(x,r) \subset \mathbb{C}$ a ball with $\delta_{\mu}(x,r) \geq \delta$. Suppose that τ is big enough, then for any $\varepsilon > 0$, there exists some $\delta_1 = \delta_1(\delta, \varepsilon) > 0$ such that

$$\frac{p_{K_t,\tau}(\mu\lfloor kB)}{\mu(B)} \leqslant \delta_1 \quad \Longrightarrow \quad \inf_D \beta_{2,\mu}^D(x,r) \leqslant \varepsilon.$$

Proof. The proof is the same as for Lemma 4.4 in [CMPT1]. We just have to use our Lemma 1.10 instead of Lemma 2.3 there for the case when both the conditions (1.37) and (1.36) are satisfied, and say that in the case

$$\theta_H(L_{z_1,z_2}) + \theta_H(L_{z_1,w}) + \theta_H(L_{z_2,w}) \leqslant \alpha_0$$

we obtain the same estimate for $dist(w, L_{z_1, z_2})$ as in the case

$$\theta_V(L_{z_1,z_2}) + \theta_V(L_{z_1,w}) + \theta_V(L_{z_2,w}) \leqslant \alpha_0.$$

By Lemma 1.11, chosen a point $x_0 \in \operatorname{spt} \mu$, there exists an approximating line D_0 such that $\beta_{1,\mu}^{D_0}(x_0, 1) \leq \varepsilon$. The next step is to construct a first Lipschitz graph in the case when D_0 is far from both the horizontal and vertical lines.

To do so, one first has to introduce a family of stopping time regions and obtain the partition spt $\mu = Z \cup F_1 \cup F_2 \cup F_3$ (see the exposition of the proof from [Leg] above). As this entirely repeats the corresponding part of [CMPT1, Section 5] (cf. [Leg, Subsection 3.1]), we omit it. We just have to mention that the thresholds θ_0 and α , arising there, have to be adapted to that D_0 is far from both the horizontal and vertical line. Namely, θ_0 is now a threshold for both $\theta_V(D_0)$ and $\theta_H(D_0)$. It means that one has to distinguish not only the cases $\theta_V(D_0) \ge \theta_0$ and $\theta_V(D_0) < \theta_0$ but also $\theta_H(D_0) \ge \theta_0$ and $\theta_H(D_0) < \theta_0$. Moreover, α is tuned as follows: if $\theta_V(D_0)$ or $\theta_H(D_0)$ are greater than θ_0 , then $\alpha \le \theta_0/10$; if $\theta_V(D_0)$ or $\theta_H(D_0)$ are not greater than θ_0 , then $\alpha = 10\theta_0$. Furthermore, see [Leg, CMPT1] for the way how one can define the Lipschitz function A on the line D_0 , using Z, F_1 , F_2 , F_3 , and appropriate thresholds.

Now we come to the main step of the proof of Proposition 1.2. The following lemma is an analogue of Lemma 6.1 from [CMPT1].

Lemma 1.12. Under the assumptions of Proposition 1.2, if furthermore

$$\theta_0 < \theta_V(D_0) < \frac{\pi}{2} - \theta_0,$$

then there exists a Lipschitz graph Γ such that $\mu(\Gamma) \geq \frac{99}{100}\mu(\mathbb{C})$.

For the proof one uses the above-mentioned function A to obtain the graph Γ , $Z \subset \Gamma$, and show that

$$\mu(F_1) + \mu(F_2) + \mu(F_3) \leq \frac{1}{100}\mu(\mathbb{C}).$$

Indeed, the following lemmas are valid (recall that $\mu(\mathbb{C}) \ge 1$ by the assumptions).

Lemma 1.13. Under the assumptions of Proposition 1.2,

$$\mu(F_1) \leqslant 10^{-6}.$$

Proof. This is an analogue of [CMPT1, Proposition 6.3], whose proof includes consideration of the two cases: 1) $\theta_V(D_0) > \theta_0$ and 2) $\theta_V(D_0) \leq \theta_0$ (see the proof of [CMPT1, Lemma 6.4]).

Under our settings, we have to consider three cases. Namely, the case 1) has to be exchanged for $\theta_0 < \theta_V(D_0) < \frac{\pi}{2} - \theta_0$, although the proof remains the same. The case 2) splits up into the following two: $\theta_V(D_0) \leq \theta_0$ and $\theta_V(D_0) \geq \frac{\pi}{2} - \theta_0$ (i.e. $\theta_H(D_0) \leq \theta_0$). Arguments in the latter case are the same as in the former one.

Lemma 1.14 (An analogue of Proposition 6.2 in [CMPT1]). Under the assumptions of Proposition 1.2,

$$\mu(F_2) \leqslant 10^{-6}.$$

Lemma 1.15. Under the assumptions of Lemma 1.12,

$$\mu(F_3) \leqslant 10^{-6}.$$

Proof. The proof stems from the one of [CMPT1, Proposition 6.5], but with exchange of $\theta_V(D_0) > \theta_0$ for $\theta_0 < \theta_V(D_0) < \frac{\pi}{2} - \theta_0$ as in Lemma 1.12.

Thus Proposition 1.2 is proved under the assumptions of Lemma 1.12. What is left is to consider the other case.

Lemma 1.16. Under the assumptions of Proposition 1.2, if furthermore

$$\theta_V(D_0) \leq \theta_0$$
 or $\theta_V(D_0) \geq \frac{\pi}{2} - \theta_0$ (*i.e.* $\theta_H(D_0) \leq \theta_0$),

then there exists a Lipschitz graph Γ such that $\mu(\Gamma) \ge 10^{-5}\mu(\mathbb{C})$.

Proof. To prove this, we repeat arguments from the proof of [CMPT1, Lemma 7.1], given for $\theta_V(D_0) \leq \theta_0$, for the case $\theta_H(D_0) \leq \theta_0$.

1.6 Additional remarks

In this section we generalise Theorem 1.2 to higher dimensions. Let us introduce necessary notation first. For $d \in \mathbb{N}^+$ and $E \subset \mathbb{R}^d$ with finite length we consider a SIO $\mathbf{T}_{K_t} = (T_{K_t}^j)_{j=1}^d$ such that formally

$$\Gamma^j_{K_t}f(x) := \int_E f(y)K^j_t(x-y)d\mathcal{H}^1(y), \qquad K^j_t(x) := \kappa^j_N(x) + t \cdot \kappa^j_n(x),$$

where $\kappa_n^j(x) := x_j^{2n-1}/|x|^{2n}$ and $x = (x_1, \ldots, x_d) \in \mathbb{R}^d \setminus \{0\}$. As before, we suppose that N > n, where $n, N \in \mathbb{N}^+$, and $t \in \mathbb{R}$. We also need the permutations

$$\mathbf{P}_{K_t}(x, y, z) := \sum_{j=1}^d P_{K_t^j}(x, y, z) \qquad \text{for distinct points } x, y, z \in \mathbb{R}^d,$$

where $P_{K_t^j}(x, y, z)$ are the same as in (0.5) with K_t^j instead of K. We also define $\mathbf{P}_{K_t}(\mu)$ analogously to (0.6).

Theorem 1.3. Let t be as mentioned in Theorem 1.1. Given a Borel set $E \subset \mathbb{R}^d$ such that $0 < \mathcal{H}^1(E) < \infty$, if $\mathbf{P}_{K_t}(\mathcal{H}^1 \lfloor E) < \infty$, then E is rectifiable. Moreover, if the operator \mathbf{T}_{K_t} is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded, then E is rectifiable.

This result for t = 0 was recently proved in [CP, Theorem 1.2(1) and Theorem 6.2]. To prove Theorem 1.3 for all required t we only need to use our Lemmas 1.5, 1.6 and 1.10 in order to show that for all $x, y, z \in \mathbb{R}^d$ such that $(x, y, z) \in \mathcal{O}_{\tau}$ and the assumptions of Lemma 1.10 are satisfied,

$$P_{K_t^j}(x, y, z) \ge C(t, \alpha_0, \tau) P_{K_0^j}(x, y, z), \qquad C(t, \alpha_0, \tau) > 0.$$
(1.43)

See the definitions of α_0 , τ and \mathcal{O}_{τ} before Lemma 1.7. Then by [CP, Proposition 3.3], adapted to the conditions (1.36) and (1.37), and the arguments similar to those in [CP, Section 6] and our Section 1.5 we immediately get the result. Note that [CP, Proposition 3.3] slightly simplifies the approach from [CMPT1] (and improves Lemma 1.9) in the case t = 0 as the parameter τ is not needed anymore. In our case this parameter is still necessary because of the inequality (1.43).

To finish, it is also worth mentioning here that under Ahlfors-David regularity assumption one can expect that for t as in Theorem 1.1 the L^2 -boundedness of the operator associated with K_t implies uniform rectifiability. This is indeed true and will be proved in Chapter 2.

Chapter 2

Singular integral operators unsuitable for the curvature method whose L^2 -boundedness still implies rectifiability

2.1 Introduction

The exposition in this chapter is based on [CMT1]. Below we continue working with the family of kernels defined in (0.24),

$$K_t(z) = \frac{(\operatorname{Re} z)^{2N-1}}{|z|^{2N}} + t \cdot \frac{(\operatorname{Re} z)^{2n-1}}{|z|^{2n}}, \quad t \in \mathbb{R}, \qquad K_\infty(z) = \frac{(\operatorname{Re} z)^{2n-1}}{|z|^{2n}},$$

and its subfamily with (n, N) = (1, 2) defined in (0.21),

$$k_t(z) = \frac{(\operatorname{Re} z)^3}{|z|^4} + t \cdot \frac{\operatorname{Re} z}{|z|^2}, \quad t \in \mathbb{R}, \qquad k_{\infty}(z) = \frac{\operatorname{Re} z}{|z|^2}.$$

We are going to use the same definitions related to curvature and permutations as given in Introduction and Chapter 1 so it is recommended to recall them. We however emphasize that as in Chapter 1 it is more convenient for us to deal with the following simplified version of the permutations (0.5) introduced in (1.2) in Chapter 1:

$$p_K(z_1, z_2, z_3) = K(z_1 - z_2)K(z_1 - z_3) + K(z_2 - z_1)K(z_2 - z_3) + K(z_3 - z_1)K(z_3 - z_2) = \frac{1}{2}P_K(z_1, z_2, z_3),$$

where K is supposed to be an odd and real-valued kernel. Note that the kernels that we consider are odd and real-valued.

Take into account that, under the same assumption on K, by (1.3) and (1.4),

$$p_K(\mu_1, \mu_2, \mu_3) = \frac{1}{2} P_K(\mu_1, \mu_2, \mu_3), \qquad p_{K,\varepsilon}(\mu_1, \mu_2, \mu_3) = \frac{1}{2} P_{K,\varepsilon}(\mu_1, \mu_2, \mu_3)$$

and

$$p_K(\mu) = \frac{1}{2} p_K(\mu, \mu, \mu), \qquad p_{K,\varepsilon}(\mu) = \frac{1}{2} P_{K,\varepsilon}(\mu, \mu, \mu),$$

where μ , μ_1 , μ_2 and μ_3 are measures.

Recall that by a measure we mean a positive locally finite Borel measure on \mathbb{C} .

We will need several results proved in Chapter 1. Namely, Theorem 1.1 states that if t belongs to the set

$$\Omega(n,N) := \begin{cases} \{0\} \cup \mathbb{R} \setminus \left(-\frac{1}{2}\left(3 + \sqrt{9 - 4\frac{N}{n}}\right), 2 - \frac{N}{n}\right) & \text{if } n < N \leq 2n, \\ \{0\} \cup \mathbb{R} \setminus \left(-\frac{1}{2}\left(\sigma_{n,M} + \sqrt{\sigma_{n,M}^2 - 4\frac{N}{n}}\right), \sigma_{n,M} - 3\right) & \text{if } N \geq 2n, \end{cases}$$

$$(2.1)$$

where $\sigma_{n,M} := 3 + (\frac{N}{n} - 2)\sqrt{N - 2n}$, then

$$p_{K_t}(z_1, z_2, z_3) \ge 0$$
 for all $(z_1, z_2, z_3) \in \mathbb{C}^3$. (2.2)

Moreover, by Theorem 1.2, if the SIO T_{K_t} , where $t \in \Omega(n, N)$, is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded, then E is rectifiable. From the other side, it follows from Remark 1.1 that there exist triples (z_1, z_2, z_3) such that $p_{K_t}(z_1, z_2, z_3)$ change sign if t belongs to the interval

$$\omega(n,N) := (-N/n,0) \tag{2.3}$$

Obviously, $\omega(n, N) \subseteq \mathbb{R} \setminus \Omega(n, N)$. Note also that $\omega(n, 2n) = (-2, 0) = \mathbb{R} \setminus \Omega(n, 2n)$. For this reason, a curvature-like method cannot be applied directly for $t \in \omega(n, N)$ (see also Remark 1.2 and Figure 0.2).

Thus we come to the question of what happens when $t \in \omega(n, N)$, i.e. the permutations $p_{K_t}(z_1, z_2, z_3)$ change sign and curvature-like methods as in [MMV, Leg, CMPT1, Chu] do not work. In this chapter a partial answer is given in the case of kernels (0.21). Namely, we show that for any fixed $t \in (-2, -\sqrt{2}) \subset \omega(1, 2)$ the analogues of Theorems A and B are still valid (a plausible conjecture for the kernels (0.24) with $t \in \omega(n, N)$ is also stated). To the best of our knowledge, this is the first example of kernels with this property in the plane. We also establish an analogue of Theorem A for the kernels (0.24) with $t \in \Omega(n, N)$. The corresponding results are given in the next section.

2.2 Main results

The following two theorems are analogues of Theorems A and B for the kernels (0.21) with $t \in (-2, -\sqrt{2})$, whose corresponding permutations change sign and a curvature-like method cannot be applied directly. We will prove them in Section 2.3 by exploiting sharp estimates for permutations related to the kernels (0.16) but not to the ones in (0.21). This will form the perturbation method that we mentioned above. Recall that $\omega(1, 2) = (-2, 0)$, see (2.3).

Theorem 2.1. Let μ be an AD-regular measure and k_t a kernel of the form (0.21), where $t \in (-2, -\sqrt{2}) \subset \omega(1, 2)$. The measure μ is uniformly rectifiable if and only if the SIO T_{k_t} is $L^2(\mu)$ -bounded.

Note that this theorem fails if $t = -1 \in \omega(1, 2)$ as follows from Theorem E (take into account that the purely unrectifiable set there is AD-regular).

Theorem 2.2. Let E be a 1-set and k_t a kernel of the form (0.21), where $t \in (-2, -\sqrt{2}) \subset \omega(1, 2)$. If the SIO T_{k_t} is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded, then E is rectifiable.

This theorem supplements the results about SIO T_{k_t} , see Figure 0.2.

Remark 2.1. As we will see at the end of Section 2.3, it is plausible that analogues of Theorems 2.1 and 2.2 are valid for the kernels (0.24) with $|t| > \sqrt{N/n}$. Note that in particular $(-N/n, -\sqrt{N/n}) \subset \omega(n, N)$, i.e. for t from this interval the corresponding permutations change sign.

We now formulate a Theorem A type result for the kernels (0.24), where $t \in \Omega(n, N)$ and thus the corresponding permutations are non-negative (see (2.1) and (2.2)).

Theorem 2.3. Let μ be an AD-regular measure and K_t a kernel of the form (0.24), where $t \in \Omega(n, N)$. The measure μ is uniformly rectifiable if and only if the SIO T_{K_t} is $L^2(\mu)$ -bounded.

Since the permutations are non-negative here, we can use a curvature-like method. The proof that will be given in Section 2.4 is more or less analogous to the one used for the kernels (0.16) in [CMPT1, Section 8].

2.3 Proof of Theorems 2.1 and 2.2

Recall that

$$k_{\infty}(z) = \kappa_1(z) = \frac{\operatorname{Re} z}{|z|^2}, \quad k_0(z) = \kappa_2(z) = \frac{(\operatorname{Re} z)^3}{|z|^4} \quad \text{and} \quad k_t(z) = \kappa_2(z) + t \cdot \kappa_1(z).$$

The following result from [CMPT1] will be necessary below.

Lemma 2.1 (Proof of Proposition 2.1 in [CMPT1]). For any u = (x, y) and v = (a, b) in the complex plane,

$$p_{\kappa_m}(0, u, v) = \sum_{k=1}^m \binom{m}{k} \frac{(ax(x-a))^{2(m-k)}}{|u|^{2m}|v|^{2m}|u-v|^{2m}} h_k(u, v),$$
(2.4)

where $h_k(u,v) := (ax)^{2k-1}(y-b)^{2k} + (x(x-a))^{2k-1}b^{2k} + (a(a-x))^{2k-1}y^{2k} \ge 0.$

To prove Theorems 2.1 and 2.2 we first obtain sharp pointwise estimates for the permutations related to the kernels (0.16).

Lemma 2.2. It holds that

$$p_{\kappa_2}(z_1, z_2, z_3) \leqslant 2p_{\kappa_1}(z_1, z_2, z_3) \quad \text{for all} \quad (z_1, z_2, z_3) \in \mathbb{C}^3.$$
 (2.5)

Proof. It is enough to prove (2.5) for $(z_1, z_2, z_3) = (0, u, v)$ as the permutations of the form (1.2) are invariant under translations (see Section 1.3). Given u = (x, y) and v = (a, b), by (2.4) we get

$$\begin{split} & 2p_{\kappa_1}(0, u, v) - p_{\kappa_2}(0, u, v) \\ & = \frac{2h_1(u, v)}{|u|^2 |v|^2 |u - v|^2} - \frac{2x^2 a^2 (x - a)^2 h_1(u, v) + h_2(u, v)}{|u|^4 |v|^4 |u - v|^4} \\ & = \frac{2\left[|u|^2 |v|^2 |u - v|^2 - x^2 a^2 (x - a)^2\right] h_1(u, v) - h_2(u, v)}{|u|^4 |v|^4 |u - v|^4} \end{split}$$

Now we obtain a lower estimate of the expression in the square brackets before $h_1(u, v)$. Expanding $|u|^2 |v|^2 |u - v|^2$ gives

$$\begin{aligned} & (x^2+y^2)(a^2+b^2)\left((x-a)^2+(y-b)^2\right)-x^2a^2(x-a)^2\\ &=x^2a^2(y-b)^2+(x^2b^2+a^2y^2+b^2y^2)\left((x-a)^2+(y-b)^2\right)\\ &\geqslant x^2a^2(y-b)^2+(x^2b^2+a^2y^2)(x-a)^2. \end{aligned}$$

Thus,

$$2p_{\kappa_1}(0,u,v) - p_{\kappa_2}(0,u,v) \ge \frac{G(x,y,a,b)}{|u|^4|v|^4|u-v|^4},$$

where

$$G(x, y, a, b) := 2(x^2a^2(y-b)^2 + (x^2b^2 + a^2y^2)(x-a)^2)h_1(u, v) - h_2(u, v).$$

Notice that by Lemma 2.1,

$$h_1(u,v) = ax(y-b)^2 + x(x-a)b^2 + a(a-x)y^2$$

$$h_2(u,v) = (ax)^3(y-b)^4 + (x(x-a))^3b^4 + (a(a-x))^3y^4.$$

Consequently, to prove the required inequality it is enough to show that $G(x, y, a, b) \ge 0$. We separate the discussion into three cases.

1) Let a = 0. Then

$$G(x, y, 0, b) = 2x^4b^2 \cdot x^2b^2 - x^6b^4 = x^6b^4 \ge 0.$$

2) Let b = 0. Then

$$\begin{split} G(x, y, a, 0) &= 2 \left(a^2 x^2 y^2 + a^2 y^2 (x - a)^2 \right) \left(a x y^2 + a (a - x) y^2 \right) - \left(a^3 x^3 y^4 + a^3 (a - x)^3 y^4 \right) \\ &= 2 a^3 y^4 \left(x^2 + (x - a)^2 \right) \left(x + (a - x) \right) - a^3 y^4 \left(x^3 - (x - a)^3 \right) \\ &= a^4 y^4 \left(2 (x^2 + (x - a)^2) - (x^2 + x (x - a) + (x - a)^2) \right) \\ &= a^4 y^4 \left(x^2 - x (x - a) + (x - a)^2 \right) \\ &= a^4 y^4 \left(x^2 - a x + a^2 \right) \\ &= a^4 y^4 \left((x - \frac{1}{2}a)^2 + \frac{3}{4}a^2 \right) \ge 0. \end{split}$$

3) Let $a \neq 0$ and $b \neq 0$. We divide G(x, y, a, b) by a^6b^4 , put $\alpha = x/a$ and $\beta = y/b$ and take into account that by Lemma 2.1 in these settings one has

$$\frac{h_k(u,v)}{a^{4k-2}b^{2k}} = \alpha^{2k-1}(\beta-1)^{2k} + \alpha^{2k-1}(\alpha-1)^{2k-1} - (\alpha-1)^{2k-1}\beta^{2k}, \qquad k = 1, 2.$$

Therefore

$$\frac{G(x, y, a, b)}{a^{6}b^{4}} = 2\left(\alpha^{2}(\beta - 1)^{2} + (\alpha^{2} + \beta^{2})(\alpha - 1)^{2}\right)\left(\alpha(\beta - 1)^{2} + \alpha(\alpha - 1) - (\alpha - 1)\beta^{2}\right) \\ - \left(\alpha^{3}(\beta - 1)^{4} + \alpha^{3}(\alpha - 1)^{3} - (\alpha - 1)^{3}\beta^{4}\right).$$

Removing brackets and further collecting terms give

$$\frac{G(x, y, a, b)}{a^6 b^4} = \left(\alpha^2 - \alpha + 1\right) \left(\beta^4 - 4\alpha\beta^3 + 6\alpha^2\beta^2 - 4\alpha^3\beta + \alpha^4\right)$$
$$= \left(\left(\alpha - \frac{1}{2}\right)^2 + \frac{3}{4}\right) (\alpha - \beta)^4 \ge 0.$$

Thus G(x, y, a, b) is non-negative in all the cases and so we are done.

Remark 2.2. The inequality (2.5) is sharp as it is known from Lemma 1.3 proved in Chapter 1 that

$$2\left[\frac{\operatorname{Re}(z_1-z_2)}{|z_1-z_2|}\frac{\operatorname{Re}(z_1-z_3)}{|z_1-z_3|}\frac{\operatorname{Re}(z_2-z_3)}{|z_2-z_3|}\right]^2p_{\kappa_1}(z_1,z_2,z_3)\leqslant p_{\kappa_2}(z_1,z_2,z_3).$$

Indeed, when all sides of the triangle (z_1, z_2, z_3) make a small angle with the horizontal, the multiplier in the square brackets is close to 1 in modulus.

The estimate (2.5) allows us to obtain an inequality for L^2 -norms.

Lemma 2.3. Let μ be a finite measure with linear growth. Then for any $\varepsilon > 0$ we have

$$||T_{\kappa_2,\varepsilon}1||_{L^2(\mu)} \leq \sqrt{2} ||T_{\kappa_1,\varepsilon}1||_{L^2(\mu)} + C\sqrt{\mu(\mathbb{C})}, \qquad C > 0.$$
 (2.6)

Proof. From Lemma 2.2 we immediately get that

$$p_{\kappa_2,\varepsilon}(\mu) \leqslant 2p_{\kappa_1,\varepsilon}(\mu).$$
 (2.7)

Now we use the identity (0.13) for the permutations (1.2), i.e.

$$||T_{K,\varepsilon}1||_{L^{2}(\mu)}^{2} = \frac{1}{3}p_{K,\varepsilon}(\mu) + \mathcal{R}_{K,\varepsilon}(\mu), \qquad |\mathcal{R}_{K,\varepsilon}(\mu)| \leq C_{K}\mu(\mathbb{C}), \qquad C_{K} > 0,$$

where K is a real and odd CZ kernel with non-negative permutations. In these terms the inequality (2.7) gives

$$\frac{1}{3}p_{\kappa_2,\varepsilon}(\mu) + \mathcal{R}_{\kappa_2,\varepsilon}(\mu) \leq 2\left(\frac{1}{3}p_{\kappa_1,\varepsilon}(\mu) + \mathcal{R}_{\kappa_1,\varepsilon}(\mu)\right) + \mathcal{R}_{\kappa_2,\varepsilon}(\mu) - 2\mathcal{R}_{\kappa_1,\varepsilon}(\mu),$$

and, consequently,

$$||T_{\kappa_2,\varepsilon}1||^2_{L^2(\mu)} \leq 2||T_{\kappa_1,\varepsilon}1||^2_{L^2(\mu)} + C\mu(\mathbb{C}), \qquad C > 0.$$

Applying the inequality $\sqrt{ax^2 + b} \leq \sqrt{ax} + \sqrt{b}$ valid for $a, b, x \geq 0$, we get (3.3).

Remark 2.3. Note that Lemma 2.2 is a particular case of [CMPT2, Lemma 7] but with an explicit constant. Nevertheless, the explicitness of the constant is essential here and actually enables us to obtain the result.

We are now ready to prove Theorems 2.1 and 2.2. By (2.6) and the triangle inequality,

$$\begin{aligned} \|T_{k_{t},\varepsilon}1\|_{L^{2}(\mu)} &= \|(T_{\kappa_{2},\varepsilon} + t \cdot T_{\kappa_{1},\varepsilon})1\|_{L^{2}(\mu)} \\ &\geqslant |t| \|T_{\kappa_{1},\varepsilon}1\|_{L^{2}(\mu)} - \|T_{\kappa_{2},\varepsilon}1\|_{L^{2}(\mu)} \\ &\geqslant (|t| - \sqrt{2}) \|T_{\kappa_{1},\varepsilon}1\|_{L^{2}(\mu)} - C\sqrt{\mu(\mathbb{C})}. \end{aligned}$$

Consequently,

$$||T_{\kappa_1,\varepsilon}1||_{L^2(\mu)} \leqslant \frac{||T_{k_t,\varepsilon}1||_{L^2(\mu)} + C\sqrt{\mu(\mathbb{C})}}{|t| - \sqrt{2}}, \qquad |t| > \sqrt{2},$$
(2.8)

and therefore for any cube $Q \subset \mathbb{C}$,

$$\|T_{\kappa_1,\varepsilon}\chi_Q\|_{L^2(\mu\lfloor Q)} \leqslant \frac{\|T_{k_t,\varepsilon}\chi_Q\|_{L^2(\mu\lfloor Q)} + C\sqrt{\mu(Q)}}{|t| - \sqrt{2}}, \qquad |t| > \sqrt{2}.$$

Applying a variant of the T1 Theorem of Nazarov, Treil and Volberg from [Tol5, Theorem 9.40], we infer that the $L^2(\mu)$ -boundedness of the SIO T_{k_t} , where t is fixed and such that $|t| > \sqrt{2}$, implies that T_{κ_1} (or equivalently the Cauchy transform by (0.13) and (0.15)) is $L^2(\mu)$ -bounded. Therefore, Theorems A and B give the desired result. Note that the "only if" part of Theorem 2.1 follows from [Dav1].

Remark 2.4. Computer experiments suggest that the following inequality holds:

$$p_{\kappa_N}(z_1, z_2, z_3) \leqslant \frac{N}{n} p_{\kappa_n}(z_1, z_2, z_3).$$
 (2.9)

(Lemma 2.2 corresponds to the case (n, N) = (1, 2).) Moreover, if $u = -\gamma + i$, $v = \gamma + i$ and $\gamma > 0$, then (see Example 1.1 in Chapter 1)

$$p_{\kappa_m}(0, u, v) = \frac{\gamma^{2m-2} \left((\gamma^2 + 1)^m - \gamma^{2m} \right)}{(\gamma^2 + 1)^{2m}}, \qquad m \in \mathbb{N},$$

and therefore

$$\lim_{\gamma \to \infty} \frac{p_{\kappa_N}(0, u, v)}{p_{\kappa_n}(0, u, v)} = \lim_{\gamma \to \infty} \frac{1 - (\gamma^2 / (\gamma^2 + 1))^N}{1 - (\gamma^2 / (\gamma^2 + 1))^n} = \frac{N}{n}$$

It means that the constant N/n is sharp if (2.9) is true.

It would follow from (2.9) in the same manner as above that the $L^2(\mu)$ -boundedness of T_{K_t} , where t is fixed and such that $|t| > \sqrt{N/n}$, implies that T_{κ_n} is $L^2(\mu)$ -bounded. This would give the analogues of Theorems 2.1 and 2.2 for the more general class of kernels (0.24) via Theorems C and D. However, we are not able to prove (2.9) yet.

2.4 Proof of Theorem 2.3

We now consider the kernels (0.24) with $t \in \Omega(n, N)$ (see (2.1)). As mentioned above, the corresponding permutations are non-negative and hence a curvature-like method can be used directly. Namely, we will adapt the arguments from [CMPT1, Section 8], which in turn stem from [DS1], to our settings. Note that the "only if" part of Theorem 2.3 follows from [Dav1]. Thus we only need to prove the "if" part.

Suppose that μ is an AD-regular measure and T_{K_t} the SIO associated with the kernels (0.24), $t \in \Omega(n, N)$. It is proved in Lemmas 1.5 and 1.6 in Chapter 1 that if

$$t \in \mathbb{R} \setminus \left[-\frac{1}{2} \left(3 + \sqrt{9 - 4\frac{N}{n}} \right), 2 - \frac{N}{n} \right], \qquad n < N \leq 2n, \qquad (2.10)$$

$$t \in \mathbb{R} \setminus \left[-\frac{1}{2} \left(\sigma_{n,M} + \sqrt{\sigma_{n,M}^2 - 4\frac{N}{n}} \right), \sigma_{n,M} - 3 \right], \qquad N \ge 2n, \tag{2.11}$$

where $\sigma_{n,M} = 3 + \left(\frac{N}{n} - 2\right)\sqrt{N - 2n}$ as above, then

$$p_{K_t}(z_1, z_2, z_3) \ge C(t) \cdot p_{\kappa_n}(z_1, z_2, z_3), \qquad C(t) > 0, \qquad (z_1, z_2, z_3) \in \mathbb{C}^3.$$

Consequently, $p_{K_t,\varepsilon}(\mu) \ge C(t) \cdot p_{\kappa_n,\varepsilon}(\mu)$ and hence from (0.13) we conclude that for t as in (2.10) and (2.11) and any cube $Q \subset \mathbb{C}$,

$$\|T_{\kappa_n,\varepsilon}\chi_Q\|_{L^2(\mu\lfloor Q)} \leqslant C(t) \left(\|T_{K_t,\varepsilon}\chi_Q\|_{L^2(\mu\lfloor Q)} + C\sqrt{\mu(Q)}\right).$$
(2.12)

By a variant of the T1 Theorem from [Tol5, Theorem 9.40] and Theorem C (see also Lemma 3.50 that will be proved in Chapter 3), the measure μ is uniformly rectifiable.

What is left, according to (2.1), is to prove Theorem 2.3 for

t

$$t = 2 - \frac{N}{n}, \qquad \qquad n < N \leqslant 2n, \qquad (2.13)$$

$$=\sigma_{n,M}-3, \qquad N \ge 2n, \qquad (2.14)$$

$$t = -\frac{1}{2} \left(3 + \sqrt{9 - 4\frac{N}{n}} \right), \qquad n < N \le 2n, \qquad (2.15)$$

$$t = -\frac{1}{2} \left(\sigma_{n,M} + \sqrt{\sigma_{n,M}^2 - 4\frac{N}{n}} \right), \qquad N \ge 2n.$$
(2.16)

To manage these cases, we introduce additional notation. Given two distinct points $z, w \in \mathbb{C}$, we denote by $L_{z,w}$ the line passing through z and w. Given three pairwise distinct points $z_1, z_2, z_3 \in \mathbb{C}$, we denote by $\measuredangle(z_1, z_2, z_3)$ the smallest angle (belonging to $[0; \pi/2]$) formed by the lines L_{z_1,z_2} and L_{z_1,z_3} . If L and L' are lines, then $\measuredangle(L, L')$ is the smallest angle (belonging to $[0; \pi/2]$) between them. Also, $\theta_V(L) := \measuredangle(L, V)$, where V is the vertical. Furthermore, for a fixed constant $\tau \ge 1$ and complex numbers z_1, z_2 and z_2 , set

$$\mathcal{O}_{\tau} := \left\{ (z_1, z_2, z_3) : \frac{|z_i - z_j|}{|z_i - z_k|} \leqslant \tau \text{ for pairwise distinct } i, j, k \in \{1, 2, 3\} \right\},$$
(2.17)

so that all triangles with vertexes z_1 , z_2 and z_3 in \mathcal{O}_{τ} have comparable sides.

Given $\alpha_0 \in (0, \pi/2)$ and $(z_1, z_2, z_3) \in \mathbb{C}^3$, in what follows we will sometimes need the conditions

$$\theta_V(L_{z_1, z_2}) + \theta_V(L_{z_2, z_3}) + \theta_V(L_{z_1, z_3}) \ge \alpha_0 \tag{2.18}$$

and

$$\theta_V(L_{z_1, z_2}) + \theta_V(L_{z_2, z_3}) + \theta_V(L_{z_1, z_3}) \leqslant \frac{3}{2}\pi - \alpha_0.$$
(2.19)

We will also use the following result which is Lemma 1.10 in Chapter 1.

Lemma 2.4. Fix $\alpha_0 \in (0, \pi/2)$. Given K_t and $(z_1, z_2, z_3) \in \mathcal{O}_{\tau}$,

- (i) if (2.18) is satisfied and t is as in (2.13) or (2.14), or
- (ii) if (2.18) and (2.19) are satisfied and t is as in (2.15) or (2.16),

then the following inequality holds:

$$p_{K_t}(z_1, z_2, z_3) \ge C(\alpha_0, \tau) \cdot c(z_1, z_2, z_3)^2, \qquad C(\alpha_0, \tau) > 0.$$
 (2.20)

On the one hand, if we are in the clause (i) of Lemma 2.4, i.e. in the same settings as in [CMPT1], then we can undeviatingly follow the scheme from [CMPT1, Section 8] (exchanging p_{κ_n} for p_{K_t}) in order to get our result for t as in (2.13) or (2.14).

On the other hand, by the clause (*ii*) of Lemma 2.4, we can ensure that the inequality (2.20) is true for t as in (2.15) or (2.16) if the sides of the triangles (z_1, z_2, z_3) are far from both the vertical and horizontal. Consequently, the scheme from [CMPT1, Section 8] cannot be applied directly for such t. Nevertheless, as we show below, it works after several modifications (besides the exchange of p_{κ_n} for p_{K_t}) connected basically with adapting geometrical arguments to both the conditions (2.18) and (2.19). Note that some of the arguments in [CMPT1, Section 8] are very sketchy and so, for the sake of completeness, we give a proof that is more detailed than the corresponding one in [CMPT1, Section 8].

The fact that the $L^2(\mu)$ -boundedness of T_{K_t} implies that μ is uniformly rectifiable will be proved by means of a corona type decomposition. We now recall how such a decomposition is defined in [DS1, Chapter 2] for a given AD-regular measure μ . The elements Qplaying the role of dyadic cubes are usually called μ -cubes.

Given an AD regular measure μ on \mathbb{C} , for each $j \in \mathbb{Z}$ (or $j \ge j_0$ if $\mu(\mathbb{C}) < \infty$) there exists a family \mathcal{D}_j of Borel subsets of spt μ , i.e. μ -cubes Q of the *j*th generation, such that

• each \mathcal{D}_j is a disjoint partition of spt μ , i.e. if $Q, Q' \in \mathcal{D}_j$ and $Q \neq Q'$, then

spt
$$\mu = \bigcup_{Q \in \mathcal{D}_j} Q$$
 and $Q \cap Q' = \emptyset;$

- if $Q \in \mathcal{D}_j$ and $Q' \in \mathcal{D}_k$ with $k \leq j$, then either $Q \subseteq Q'$ or $Q \cap Q' = \emptyset$;
- for all $j \in \mathbb{Z}$ and $Q \in D_j$, we have

$$2^{-j} \lesssim \operatorname{diam}(Q) \lesssim 2^{-j}$$
 and $\mu(Q) \approx 2^{-j}$

In what follows, $\mathcal{D} := \bigcup_{j \in \mathbb{Z}} \mathcal{D}_j$. Moreover, given $Q \in \mathcal{D}_j$, we define the *side length* of Q as $\ell(Q) = 2^{-j}$, which actually indicates the generation of Q. Obviously, $\ell(Q) \approx \text{diam}(Q)$. The value of $\ell(Q)$ is not well defined if the μ -cube Q belongs to $\mathcal{D}_j \cap \mathcal{D}_k$ with $j \neq k$. To avoid this, one may consider a $Q \in \mathcal{D}_j$ as a couple (Q, j).

Given $\lambda > 1$ and $Q \in \mathcal{D}$, set

$$\lambda Q := \{ x \in \operatorname{spt} \mu : \operatorname{dist} (x, Q) \leqslant (\lambda - 1)\ell(Q) \}.$$

We will also need the following version of P. Jones' β -numbers for μ -cubes (see [DS2]):

$$\beta_q(Q) := \inf_L \left(\frac{1}{\ell(Q)} \int_{\eta_1 Q} \left(\frac{\operatorname{dist}(x,L)}{\ell(Q)} \right)^q d\mu(x) \right)^{1/q}, \qquad 1 \leqslant q \leqslant \infty.$$

where $\eta_1 > 4$ is some constant to be fixed later and the infimum is taken over all affine lines L. We will mostly use $\beta_1(Q)$ and denote by L_Q the best approximating line for $\beta_1(Q)$.

Given $Q \in \mathcal{D}_j$, the sons of Q, forming the collection $\mathsf{Sons}(Q)$, are the μ -cubes $Q' \in \mathcal{D}_{j+1}$ such that $Q' \subseteq Q$.

By [DS1, Chapter 2], one says that μ admits a *corona decomposition* if there are parameters $\eta, \delta > 0$ and a triple ($\mathcal{B}, \mathcal{G}, \mathsf{Tree}$), where \mathcal{B} and \mathcal{G} are subsets of \mathcal{D} and Tree is a family of subsets S of \mathcal{G} , such that the following conditions are satisfied:

- 1. $\mathcal{D} = \mathcal{B} \cup \mathcal{G}$ and $\mathcal{B} \cap \mathcal{G} = \emptyset$.
- 2. \mathcal{B} satisfies a Carleson packing condition, i.e.

$$\sum_{Q \in \mathcal{B}: Q \subseteq R} \mu(Q) \lesssim_{\eta} \mu(R) \quad \text{for all} \quad R \in \mathcal{D}.$$
(2.21)

- 3. $\mathcal{G} = \bigcup_{S \in \mathsf{Tree}} S$ and the union is disjoint;
- 4. Each $S \in \text{Tree}$ is called a *tree* and is *coherent*: each S has a unique maximal element Q_S , which contains all other elements of S as subsets, i.e.
 - a μ -cube $Q' \in \mathcal{D}$ belongs to S if $Q \subseteq Q' \subseteq Q_S$ for some $Q \in S$;
 - if $Q \in S$ then either all elements of Sons(Q) lie in S or none of them do.
- 5. For each $S \in \text{Tree}$, there exists a (possibly rotated) Lipschitz graph Γ_S with constant smaller than η such that dist $(x, \Gamma_S) \leq \delta \operatorname{diam}(Q)$ whenever $x \in 2Q$ and $Q \in S$.
- 6. The maximal μ -cubes Q_S , for $S \in$ Tree, satisfy the Carleson packing condition

$$\sum_{S \in \mathsf{Tree:} \ Q_S \subseteq R} \mu(Q_S) \lesssim \mu(R) \quad \text{for all} \quad R \in \mathcal{D}.$$

According to [DS1, Section 1, (C4) and (C6)], if μ is uniformly rectifiable, then it admits a corona decomposition for all $\eta, \delta > 0$. Conversely, the existence of a corona decomposition for a single set of η and δ implies that μ is uniformly rectifiable. We now turn to constructing our corona decomposition. Let $\varepsilon > 0$ be some small constant to be chosen at the very end of the construction. From now on, $\mathcal{B}_0(\varepsilon)$ stands for the family of cubes $Q \in \mathcal{D}$ such that $\beta_1(Q) \ge \varepsilon$. Furthermore, $\mathcal{G}_0(\varepsilon) := \mathcal{D} \setminus \mathcal{B}_0(\varepsilon)$. The aim is to show that $\mathcal{B}_0(\varepsilon)$ satisfies a Carleson packing condition.

Note that constants in some inequalities below depend also on the AD-regularity constant and the constants η_1 , η_2 and η_3 related to β -numbers. Since these constants are fixed at some point, we do not indicate this dependence explicitly. On the contrary, we emphasize the dependence of some forthcoming inequalities on the essential parameter ε .

Recall that $\ell(Q) \approx \mu(Q)$. We start with observing that for any $y, z \in \eta_1 Q$ we have

$$\beta_{2}(Q)^{2} \leqslant \frac{1}{\ell(Q)} \int_{\eta_{1}Q} \left(\frac{\operatorname{dist}\left(x, L_{y,z}\right)}{\ell(Q)} \right)^{2} d\mu(x)$$

$$= \frac{1}{\ell(Q)} \left(\int_{\frac{\operatorname{dist}\left(x, L_{y,z}\right)}{\ell(Q)} < \varepsilon^{2}} \left(\frac{\operatorname{dist}\left(x, L_{y,z}\right)}{\ell(Q)} \right)^{2} d\mu(x) + \int_{\frac{\operatorname{dist}\left(x, L_{y,z}\right)}{\ell(Q)} \ge \varepsilon^{2}} \left(\frac{\operatorname{dist}\left(x, L_{y,z}\right)}{\ell(Q)} \right)^{2} d\mu(x) \right)$$

$$\lesssim \varepsilon^{4} + \frac{1}{\ell(Q)} \int_{\frac{\operatorname{dist}\left(x, L_{y,z}\right)}{\ell(Q)} \ge \varepsilon^{2}} \left(\frac{\operatorname{dist}\left(x, L_{y,z}\right)}{\ell(Q)} \right)^{2} d\mu(x).$$

Lemma 2.5. Let $B_1 = B(\zeta_1, r_1)$ and $B_2 = B(\zeta_2, r_2)$ be two balls such that $B_1 \cap \operatorname{spt} \mu \subset \eta_1 Q$, $B_2 \cap \operatorname{spt} \mu \subset \eta_1 Q$, dist $(B_1, B_2) \approx \ell(Q)$ and $r_1 \approx r_2 \approx \ell(Q)$. If $y \in B_1$ and $z \in B_2$, then for ε small enough,

$$\int_{\substack{x \in \eta_1 Q:\\ \frac{\operatorname{dist}(x, L_{y,z})}{\ell(Q)} \geqslant \varepsilon^2}} \operatorname{dist}(x, L_{y,z})^2 d\mu(x) \lesssim_{\varepsilon} \ell(Q)^2 p_{K_t}^{(\varepsilon; Q)}(\mu),$$

where

$$p_{K_t}^{(\varepsilon;Q)}(\mu) := \iiint_{\substack{(x,y,z) \in (\eta_1 Q)^3 \\ |x-y| \ge \varepsilon^2 \ell(Q), \\ |x-z| \ge \varepsilon^2 \ell(Q)}} p_{K_t}(x,y,z) \ d\mu(x) d\mu(y) d\mu(z)$$

Note that the existence of the above-mentioned balls B_1 and B_2 is guaranteed in the AD-regular case.

Proof. Note that the condition dist $(x, L_{y,z}) \ge \varepsilon^2 \ell(Q)$ implies that $|x - y| \ge \varepsilon^2 \ell(Q)$ and $|x - z| \ge \varepsilon^2 \ell(Q)$. Consequently, since $x \in \eta_1 Q$, $y \in B_1$ and $z \in B_2$,

$$|x-z| \approx_{\varepsilon} |x-y| \approx_{\varepsilon} |y-z|.$$
 (2.22)

We now separate two cases.

(1) Suppose that

$$\varepsilon^{10} \leq \theta_V(L_{x,y}) + \theta_V(L_{y,z}) + \theta_V(L_{x,z}) \leq \frac{3}{2}\pi - \varepsilon^{10}$$

Then by the clause (*ii*) of Lemma 2.4, where we put $\alpha_0 = \varepsilon^{10}$ and $\tau = \tau(\varepsilon, \eta_1)$ chosen with respect to the constants in (2.22), we have $c(x, y, z)^2 \lesssim_{\varepsilon} p_{K_t}(x, y, z)$.

(2) Now let

$$\theta_V(L_{x,y}) + \theta_V(L_{y,z}) + \theta_V(L_{x,z}) < \varepsilon^{10}$$

or

$$\theta_V(L_{x,y}) + \theta_V(L_{y,z}) + \theta_V(L_{x,z}) > \frac{3}{2}\pi - \varepsilon^{10}.$$

In this case dist $(x, L_{y,z}) \lesssim \varepsilon^{10} \ell(Q)$. Thus for ε small enough we get a contradiction with the assumption dist $(x, L_{y,z}) \ge \varepsilon^2 \ell(Q)$.

Summarizing,

$$\begin{split} \int_{\frac{dist}{\ell(Q)}} \sum_{\substack{x \in \eta_1 Q:\\ \ell(Q)}} dist (x, L_{y,z})^2 d\mu(x) \\ \lesssim_{\varepsilon} \frac{\ell(Q)^4}{\mu(B_1)\mu(B_2)} \int_{B_2} \int_{B_1} \int_{\frac{dist}{dist}} \frac{\eta_1 Q:}{\ell(Q)} \left(\frac{dist}{|x - y||x - z|} \right)^2 d\mu(x) d\mu(y) d\mu(z) \\ \lesssim_{\varepsilon} \ell(Q)^2 \int_{\eta_1 Q} \int_{\eta_1 Q} \int_{\frac{dist}{\ell(Q)}} \frac{\eta_1 Q:}{\ell(Q)} c(x, y, z)^2 d\mu(x) d\mu(y) d\mu(z) \\ \lesssim_{\varepsilon} \ell(Q)^2 p_{K_t}^{(\varepsilon; Q)}(\mu). \end{split}$$

We used the well-known identity $c(x, y, z) = \text{dist}(x, L_{y,z})/(|x - y||x - z|).$

The estimate for $\beta_2(Q)^2$ that we obtained above and Lemma 2.5 give

$$\beta_2(Q)^2 \lesssim \varepsilon^4 + \frac{C(\varepsilon)}{\ell(Q)} p_{K_t}^{(\varepsilon;Q)}(\mu), \qquad Q \in \mathcal{D}.$$

We now take into account that $\beta_1(Q) \leq \beta_2(Q)$ by Hölder's inequality and, consequently, if $Q \in \mathcal{B}_0(\varepsilon)$, i.e. $\beta_1(Q) \geq \varepsilon$, then $\beta_2(Q) \geq \varepsilon$. From this we deduce for sufficiently small ε that

$$\mu(Q) \lesssim_{\varepsilon} p_{K_t}^{(\varepsilon;Q)}(\mu) \quad \text{for any} \quad Q \in \mathcal{B}_0(\varepsilon).$$

From this we immediately get that

$$\sum_{Q\in\mathcal{B}_0(\varepsilon):\,Q\subseteq R}\mu(Q)\lesssim_{\varepsilon}\sum_{Q\in\mathcal{B}_0(\varepsilon):\,Q\subseteq R}p_{K_t}^{(\varepsilon;Q)}(\mu).$$

To estimate the latter sum, we will use the notation

$$A_j(\varepsilon) := \{ x : \varepsilon^2 2^{-j} \le |x - y| \le c 2^{-j} \} \quad \text{for some sintable } c > 0.$$
 (2.23)

Recall that $\ell(Q) = 2^{-j}$ if $Q \in \mathcal{D}_j$. Clearly, (2.23) are the concentric annuli $B(y, c\ell(Q)) \setminus B(y, \varepsilon^2 \ell(Q))$ contained in the ball $B(y, c\ell(R))$ and having bounded overlap depending on ε and c. These observations lead to the following:

$$\begin{split} \sum_{Q \in \mathcal{B}_{0}(\varepsilon): Q \subseteq R} p_{K_{t}}^{(\varepsilon;Q)}(\mu) \\ &\leqslant \int_{\eta_{1}R} \int_{\eta_{1}R} \left(\sum_{j \ge j_{0}} \sum_{Q \in \mathcal{B}_{0}(\varepsilon) \cap \mathcal{D}_{j}(R)} \int_{\eta_{1}Q \cap A_{j}(\varepsilon)} p_{K_{t}}(x,y,z) \ d\mu(x) \right) d\mu(y) d\mu(z) \\ &\lesssim \int_{\eta_{1}R} \int_{\eta_{1}R} \left(\sum_{j \ge j_{0}} \int_{\eta_{1}R \cap A_{j}(\varepsilon)} p_{K_{t}}(x,y,z) \ d\mu(x) \right) d\mu(y) d\mu(z) \\ &\lesssim_{\varepsilon} p_{K_{t}}(\mu \lfloor (\eta_{1}R)). \end{split}$$

Since T_{K_t} is $L^2(\mu)$ -bounded, we get $p_{K_t}(\mu \lfloor F) \leq \mu(F)$ for any $F \subset \mathbb{C}$. Consequently,

$$p_{K_t}(\mu \lfloor (\eta_1 R)) \lesssim \mu(R) \quad \text{for all} \quad R \in \mathcal{D},$$

and therefore we reach the desired inequality

$$\sum_{Q \in \mathcal{B}_0(\varepsilon): Q \subseteq R} \mu(Q) \lesssim_{\varepsilon} \mu(R) \quad \text{for all} \quad R \in \mathcal{D}.$$

Thus, for any $\varepsilon > 0$, there exists the decomposition

$$\mathcal{D} = \mathcal{B}_0(\varepsilon) \cup \mathcal{G}_0(\varepsilon), \tag{2.24}$$

where $\mathcal{B}_0(\varepsilon)$ satisfies a Carleson packing condition and for any cube $Q \in \mathcal{G}_0(\varepsilon)$ there exists a line L_Q such that dist $(x, L_Q) \lesssim \sqrt{\varepsilon} \ell(Q)$ for all $x \in \frac{1}{2}\eta_1 Q$ (since $\beta_1(Q) < \varepsilon$ for such cubes and $\beta_{\infty}(Q) \lesssim \sqrt{\beta_1(2Q)}$). More details can be found in [DS1, Ch. 6].

Using the decomposition (2.24), we now can apply [DS1, Lemma 7.1] in order to obtain a new decomposition (still depending on ε) but already with a family of stopping time regions. Suppose that θ_0 is small enough and $0 < \varepsilon \ll \theta_0$ to prove the following assertion.

Lemma 2.6. For all sufficiently small $\varepsilon > 0$, there exists a decomposition $\mathcal{D} = \mathcal{B} \cup \mathcal{G}$, where $\mathcal{B} = \mathcal{B}(\varepsilon)$ satisfies a Carleson packing condition (with a constant depending on ε) and $\mathcal{G} = \mathcal{G}(\varepsilon)$ can be partitioned into a family Tree of coherent regions S, satisfying the following. For each $S \in$ Tree denote

$$\alpha(S) := \frac{1}{10}\theta_0 \quad \text{if} \quad \theta_0 \leqslant \theta_V(L_{Q_S}) \leqslant \pi/2 - \theta_0$$

and

$$\alpha(S) := 10\theta_0 \quad if \quad \theta_V(L_{Q_S}) < \theta_0 \quad or \qquad \theta_V(L_{Q_S}) > \pi/2 - \theta_0.$$

Then

- if $Q \in S$, then $\measuredangle(L_Q, L_{Q_S}) \leq \alpha(S)$;
- if Q is a minimal cube of S, then either at least one element of Sons(Q) lies in B or else ∠(L_Q, L_{Q_S}) ≥ ¹/₂α(S).

Here $\mathcal{G} \subseteq \mathcal{G}_0(\varepsilon)$ and therefore for any $Q \in \mathcal{G}$ one has $\beta_1(Q) < \varepsilon$.

Lemma 2.6 is an analogue of [CMPT1, Lemma 8.1] which comes from [DS1, Lemma 7.1]. The main difference between [CMPT1, Lemma 8.1] and [DS1, Lemma 7.1] is that two different values of the parameter $\alpha(S)$ have to be chosen, according to the angle $\theta_V(L_{QS})$. In our case the situations where the angle $\theta_V(L_{QS})$ is close to zero and $\pi/2$ have to be also distinguished.

To obtain the required Lipschitz function, one can follow the proof of [DS1, Proposition 8.2] to deduce the following lemma.

Lemma 2.7. For each $S \in$ Tree from Lemma 2.6, there exists a Lipschitz function A_S : $L_{Q_S} \to L_{Q_S}^{\perp}$ with norm $\leq \alpha(S)$ such that, denoting by Γ_S the graph of A_S ,

dist
$$(x, \Gamma_S) \lesssim \sqrt{\varepsilon} \,\ell(Q)$$

for all $x \in 2Q$ with $Q \in S$.

The proof will be completed if we show that the maximal μ -cubes $Q_S, S \in \mathsf{Tree}$, satisfy the Carleson packing condition

$$\sum_{S \in \mathsf{Tree: } Q_S \subseteq R} \mu(Q_S) \lesssim_{\varepsilon} \mu(R) \quad \text{ for all } \quad R \in \mathcal{D}.$$

To do so, we will distinguish several types of trees.

Here and subsequently, Stop(S) denotes the family of the *minimal* μ -cubes of $S \in Tree$, which may be empty. By Lemma 2.6, we can split Stop(S) as follows:

$$\mathsf{Stop}(S) = \mathsf{Stop}_{\alpha}(S) \cup \mathsf{Stop}_{\beta}(S), \qquad \mathsf{Stop}_{\alpha}(S) \cap \mathsf{Stop}_{\beta}(S) = \varnothing, \tag{2.25}$$

where $\operatorname{Stop}_{\beta}(S)$ contains all minimal μ -cubes Q such that at least one element of $\operatorname{Sons}(Q)$ belongs to \mathcal{B} and $\operatorname{Stop}_{\alpha}(S)$ contains all minimal Q such that $\measuredangle(L_Q, L_{Q_S}) \ge \frac{1}{2}\alpha(S)$.

The first set that we will consider is

$$\Delta_1 := \left\{ S \in \operatorname{Tree} : \mu\left(Q_S \setminus \bigcup_{Q \in \operatorname{Stop}(S)} Q\right) \ge \frac{1}{2} \, \mu(Q_S) \right\}.$$

Clearly, if $S \in \text{Tree} \setminus \Delta_1$, then by (2.25),

$$\frac{1}{2}\mu(Q_S) < \mu\left(\bigcup_{Q \in \mathsf{Stop}_{\beta}(S)}Q\right) = \mu\left(\bigcup_{Q \in \mathsf{Stop}_{\alpha}(S)}Q\right) + \mu\left(\bigcup_{Q \in \mathsf{Stop}_{\beta}(S)}Q\right).$$
(2.26)

Now let

$$\Delta_2 := \left\{ S \in \mathsf{Tree} \setminus \Delta_1 : \mu\left(\bigcup_{Q \in \mathsf{Stop}_\beta(S)} Q\right) \ge \frac{1}{4} \, \mu(Q_S) \right\}.$$

The trees remained are in

$$\Delta_3 := \left\{ S \in \mathsf{Tree} \setminus (\Delta_1 \cup \Delta_2) : \mu\left(\bigcup_{Q \in \mathsf{Stop}_\alpha(S)} Q\right) \geqslant \frac{1}{4}\,\mu(Q_S) \right\}.$$

Indeed, if $S \in \mathsf{Tree} \setminus (\Delta_1 \cup \Delta_2 \cup \Delta_3)$, then (2.26) is valid and moreover

$$\mu\left(\bigcup_{Q\in\mathsf{Stop}_{\alpha}(S)}Q\right) < \frac{1}{4}\,\mu(Q_S) \quad \text{and} \quad \mu\left(\bigcup_{Q\in\mathsf{Stop}_{\beta}(S)}Q\right) < \frac{1}{4}\,\mu(Q_S).$$

This means that Tree $\setminus (\Delta_1 \cup \Delta_2 \cup \Delta_3) = \emptyset$.

We also split Δ_3 in the three disjoint sets:

$$\begin{split} \Delta_3' &:= \left\{ S \in \Delta_3 : \theta_0 \leqslant \theta_V(L_{Q_S}) \leqslant \pi/2 - \theta_0 \right\}, \\ \Delta_3'' &:= \left\{ S \in \Delta_3 : \theta_V(L_{Q_S}) < \theta_0 \right\}, \\ \Delta_3''' &:= \left\{ S \in \Delta_3 : \theta_V(L_{Q_S}) > \pi/2 - \theta_0 \right\}. \end{split}$$

So we have the disjoint union

$$\mathsf{Tree} = \Delta_1 \cup \Delta_2 \cup \Delta_3^{'} \cup \Delta_3^{''} \cup \Delta_3^{'''}.$$

The procedure now is to check the required Carleson packing condition for all components of this union.

For all $S \in$ Tree the sets $Q_S \setminus \bigcup_{Q \in \mathsf{Stop}(S)} Q$ are pairwise disjoint and hence for $S \in \Delta_1$ we get

$$\sum_{S \in \Delta_1: Q_S \subseteq R} \mu(Q_S) \leqslant 2 \sum_{S \in \mathsf{Tree: } Q_S \subseteq R} \mu\left(Q_S \setminus \bigcup_{Q \in \mathsf{Stop}(S)} Q\right) \leqslant 2\mu(R).$$

If $S \in \Delta_2$, then by definition and the fact that $\mu(Q) \approx \mu(Q')$ for $Q' \in \mathsf{Sons}(Q)$,

$$\mu(Q_S) \leqslant 4\mu\left(\bigcup_{Q \in \mathsf{Stop}_\beta(S)} Q\right) \lesssim \sum_{Q \in \mathsf{Stop}(S)} \sum_{Q' \in \mathcal{B} \cap \mathsf{Sons}(Q)} \mu(Q')$$

and consequently by Lemma 2.6,

$$\begin{split} \sum_{S \in \Delta_2: \ Q_S \subseteq R} \mu(Q_S) \lesssim \sum_{S \in \Delta_2: \ Q_S \subseteq R} \sum_{Q \in \mathsf{Stop}(S)} \sum_{Q' \in \mathcal{B} \cap \mathsf{Sons}(Q)} \mu(Q') \\ \leqslant \sum_{Q \in \mathcal{B}: \ Q_S \subseteq R} \mu(Q) \\ \lesssim_{\varepsilon} \mu(R). \end{split}$$

Let us consider the case $S \in \Delta'_3$. We will need β -numbers defined for balls B(x, r):

$$\beta_q(x,r) := \inf_L \left(\frac{1}{r} \int_{B(x,2r)} \left(\frac{\operatorname{dist}(x,L)}{r} \right)^q d\mu(x) \right)^{1/q}, \qquad 1 \leqslant q \leqslant \infty,$$

where the infimum is taken over all affine lines L.

It is claimed in [DS1, Section 12, Inequality (12.2)] that for all $S \in \Delta_3$ there exists $\eta_2 > 1$ such that

$$\mu(Q_S) \lesssim \iint_{X_S} \beta_1(x,\eta_2 r)^2 \frac{d\mu(x)dr}{r},$$

where

$$X_S := \{ (x, r) \in \operatorname{spt} \mu \times \mathbb{R}^+ : x \in \eta_2 Q_S, \frac{1}{\eta_2} d(x) \leqslant r \leqslant \eta_2 \operatorname{diam} (Q_S) \}$$
(2.27)

and

$$d(x) := \inf_{Q \in S} \{ \operatorname{dist} (x, Q) + \operatorname{diam} (Q) \}.$$

By Holder's inequality, $\beta_1(x, \eta_2 r) \leq \beta_2(x, \eta_2 r)$. Moreover, it follows from [Leg, Lemma 2.5 and Proof of Proposition 2.4] (or by the arguments analogous to the ones in the proof of Lemma 2.5) that if μ is AD-regular, then there exists $\eta_3 \geq 2$ such that for any $x \in \operatorname{spt} \mu$,

$$\beta_2(x,\eta_2 r)^2 \lesssim \frac{1}{\eta_2 r} \iiint_{\mathcal{O}_{\eta_3}(x,\eta_2 r)} c(u,v,w)^2 d\mu(u) d\mu(v) d\mu(w),$$

where

$$\mathcal{O}_{\eta_3}(x,\rho) := \left\{ (u,v,w) \in (B(x,\eta_3\rho))^3 : |u-v| \ge \frac{\rho}{\eta_3}, |v-w| \ge \frac{\rho}{\eta_3}, |u-w| \ge \frac{\rho}{\eta_3} \right\}.$$

Note also that for any $(u, v, w) \in \mathcal{O}_{\eta_3}(x, \rho)$ we have $|u - v| \leq 2\eta_3 \rho$, $|v - w| \leq 2\eta_3 \rho$ and $|u - w| \leq 2\eta_3 \rho$, and thus for a fixed η_3 ,

$$|u-v|\approx |v-w|\approx |u-w|\approx \rho$$

Therefore if a triple $(u, v, w) \in \mathcal{O}_{\eta_3}(x, \eta_2 r)$ with $(x, r) \in X_S$, then at least one side of the triangle (u, v, w) makes a big angle with the vertical and horizontal. Indeed, by construction, if η_1 is chosen much bigger than η_2 , then $\beta_{\infty}(x, \eta_2 r) \leq \sqrt{\varepsilon}$, and consequently the angle between one side of (u, v, w) and the best approximating line L_{Q_S} is less than $C(\eta_3)\sqrt{\varepsilon}$ with some $C(\eta_3) > 0$. Furthermore, $\theta_0 \leq \theta_V(L_{Q_S}) \leq \pi/2 - \theta_0$ and thus the angle that one side of (u, v, w) makes with the vertical and horizontal belongs to the interval

$$\left(\frac{9}{10}\theta_0 - C(\eta_3)\sqrt{\varepsilon}, \pi/2 - \frac{9}{10}\theta_0 + C(\eta_3)\sqrt{\varepsilon}\right) \supseteq \left(\frac{1}{2}\theta_0, \pi/2 - \frac{1}{2}\theta_0\right),$$

where ε is chosen sufficiently small. This fact enables us to use the clause (*ii*) of Lemma 2.4 and exchange the curvature for our permutation p_{K_t} :

$$\beta_2(x,\eta_2 r)^2 \lesssim_{\theta_0} \frac{1}{\eta_2 r} \iiint_{\mathcal{O}_{\eta_3}(x,\eta_2 r)} p_{K_t}(u,v,w) \ d\mu(u) d\mu(v) d\mu(w), \qquad (x,r) \in X_S.$$

Summarizing and taking into account that θ_0 is fixed, we get

$$\mu(Q_S) \lesssim \iiint_{X_S} \iiint_{\mathcal{O}_{\eta_3}(x,\eta_2 r)} p_{K_t}(u,v,w) \ d\mu(u) d\mu(v) d\mu(w) \ \frac{d\mu(x)dr}{(\eta_2 r)^2}.$$

What is more, it is shown after [DS1, Lemma 7.9] that the regions X_S (see (2.27)) with $S \in \Delta'_3$ have bounded overlap. By this reason,

$$\sum_{S \in \Delta'_3: Q_S \subseteq R} \mu(Q_S)$$

$$\lesssim \sum_{S \in \Delta'_3: Q_S \subseteq R} \iint_{X_S} \iiint_{\mathcal{O}_{\eta_3}(x,\eta_2 r)} p_{K_t}(u,v,w) \ d\mu(u) d\mu(v) d\mu(w) \ \frac{d\mu(x) dr}{(\eta_2 r)^2}$$

$$\lesssim \int_0^{2\eta_2 \ell(R)} \int_{2\eta_2 R} \iiint_{\mathcal{O}_{\eta_3}(x,\eta_2 r)} p_{K_t}(u,v,w) \ d\mu(u) d\mu(v) d\mu(w) \ \frac{d\mu(x) dr}{(\eta_2 r)^2}$$

$$\lesssim p_{K_t}(\mu \lfloor (2\eta_2 R)).$$

The third inequality is by Fubini's theorem. See the definition of \mathcal{O}_{τ} in (2.17). Finally, by the L^2 -boundedness of T_{K_t} , we get

$$\sum_{S\in \Delta'_3:\;Q_S\subseteq R}\mu(Q_S)\lesssim \mu(R)$$

Suppose now that $S \in \Delta_3''$. If $Q \in \mathsf{Stop}_{\alpha}(S)$, then $\mathsf{Sons}(Q) \cap \mathcal{B} = \emptyset$ and by Lemma 2.6,

$$\measuredangle(L_Q, L_{Q_S}) \leqslant \alpha(S), \qquad \measuredangle(L_Q, L_{Q_S}) \geqslant \frac{1}{2}\alpha(S), \qquad \alpha(S) = 10\theta_0,$$

and thus

$$\theta_V(L_Q) \leqslant \measuredangle(L_Q, L_{Q_S}) + \theta_V(L_{Q_S}) < 10\theta_0 + \theta_0 = 11\theta_0, \\ \theta_V(L_Q) \geqslant \measuredangle(L_Q, L_{Q_S}) - \theta_V(L_{Q_S}) > 5\theta_0 - \theta_0 = 4\theta_0.$$

Since $\beta_1(Q) < \varepsilon$, we can choose ε small enough so that $\measuredangle(L_Q, L_{Q'}) \leq \theta_0, Q' \in \mathsf{Sons}(Q)$, and hence

$$3\theta_0 < \theta_V(L_{Q'}) < 12\theta_0, \qquad Q' \in \mathsf{Sons}(Q).$$

Consequently, any element of Sons(Q) is the maximal μ -cube of a tree belonging either to Δ_1 , Δ_2 or Δ'_3 . Additionally, from the definition of Δ_3 and the fact that minimal cubes for a single tree are pairwise disjoint it follows that

$$\mu(Q_S) \leqslant 4\mu\left(\bigcup_{Q \in \mathsf{Stop}_{\alpha}(S)} Q\right) = 4\sum_{Q \in \mathsf{Stop}_{\alpha}(S)} \mu(Q) = 4\sum_{Q \in \mathsf{Stop}_{\alpha}(S)} \sum_{Q' \in \mathsf{Sons}(Q)} \mu(Q').$$

From the above-mentioned we deduce that

$$\begin{split} \sum_{S \in \Delta_3'': Q_S \subseteq R} \mu(Q_S) \leqslant 4 \sum_{S \in \Delta_3'': Q_S \subseteq R} \sum_{\substack{Q \in \mathsf{Stop}_\alpha(S) \\ S \in \Delta_1 \cup \Delta_2 \cup \Delta_3': Q_S \subseteq R}} \sum_{\substack{Q' \in \mathsf{Sons}(Q) \\ P(Q') \\ Q' \in \mathsf{Sons}(Q)}} \mu(Q') \end{split}$$

Take into account that the maximal cubes of all trees from $\Delta_1 \cup \Delta_2 \cup \Delta'_3$ satisfy a Carleson packing condition (with a constant depending on ε). By this reason,

$$\sum_{S \in \Delta_3'' \colon Q_S \subseteq R} \mu(Q_S) \lesssim_{\varepsilon} \mu(R).$$

£

One can argue for $S \in \Delta_3^{''}$ in the same manner as for $S \in \Delta_3^{''}$. Indeed, if ε is appropriately chosen and $Q \in \text{Stop}_{\alpha}(S)$, then

$$\pi/2 - 12\theta_0 < \theta_V(L_{Q'}) < \pi/2 - 3\theta_0, \qquad Q' \in \mathsf{Sons}(Q),$$

and hence any element of Sons(Q) is the maximal μ -cube of a tree belonging either to Δ_1 , Δ_2 or Δ'_3 .

Summarizing, we proved that maximal cubes of all types of trees satisfy a Carleson packing condition and so the triple $(\mathcal{B}, \mathcal{G}, \mathsf{Tree})$ is a corona decomposition as required.

2.5 Additional remarks

To finish, we would like to mention a corollary of the results from the previous sections.

Let μ be a measure with linear growth and

$$t \in (-\infty, -\sqrt{2}) \cup (0, \infty). \tag{2.28}$$

If the SIO T_{k_t} is $L^2(\mu)$ -bounded, then all 1-dimensional SIOs associated with a wide class of kernels defined around (0.23) are also $L^2(\mu)$ -bounded. Indeed, it follows from (2.8) and (2.12) with (n, N) = (1, 2) that for any t as in (2.28) and any cube $Q \subset \mathbb{C}$, one has

$$\|T_{\kappa_1,\varepsilon}\chi_Q\|_{L^2(\mu \lfloor Q)} \leqslant C(t) \left(\|T_{k_t,\varepsilon}\chi_Q\|_{L^2(\mu \lfloor Q)} + \sqrt{\mu(Q)}\right), \qquad C(t) > 0,$$

where T_{κ_1} , as we have already mentioned before, is the SIO associated with the real part of the Cauchy kernel, i.e. with the Cauchy kernel, up to a constant. Using the T1 Theorem from [Tol5, Theorem 9.40], we conclude that the $L^2(\mu)$ -boundedness of T_{k_t} with t as in (2.28) implies that the Cauchy transform is $L^2(\mu)$ -bounded. It is left to apply the results from [Tol2, Tol4].

Chapter 3

A family of singular integral operators which control the Cauchy transform

3.1 Introduction and Theorems

The exposition in this chapter is based on [CMT2]. We aim to study the behaviour of SIOs associated with the kernels (0.21), i.e. with

$$k_t(z) = \frac{(\operatorname{Re} z)^3}{|z|^4} + t \cdot \frac{\operatorname{Re} z}{|z|^2}, \quad t \in \mathbb{R}, \qquad k_{\infty}(z) = \frac{\operatorname{Re} z}{|z|^2},$$

in the case of small negative t. We will first show that for a measure with linear growth the corresponding L^2 -norm of T_{k_0} controls the L^2 -norm of $T_{k_{\infty}}$ and thus of the Cauchy transform. As a corollary, we prove that the $L^2(\mathcal{H}^1 \lfloor E)$ -boundedness of T_{k_t} with some $t \in (-t_0, 0)$, where $t_0 > 0$ is an absolute constant, implies that E is rectifiable (as above, E is a 1-set). This is so in spite of the fact that the usual curvature method fails to be directly applicable for $t \in (-2, 0)$ as shown in Chapter 1. As we will see, the study of kernels (0.21) for small negative t is more difficult than in the case of $t \in (-2, -\sqrt{2})$ considered in Chapter 2.

We will use the same definitions related to curvature and permutations as given in Introduction and Chapters 1 and 2. Now we will recall some of them. Again we will work with the following simplified version of the permutations P_K given in (0.5),

$$p_K(z_1, z_2, z_3) = K(z_1 - z_2)K(z_1 - z_3) + K(z_2 - z_1)K(z_2 - z_3) + K(z_3 - z_1)K(z_3 - z_2)$$

= $\frac{1}{2}P_K(z_1, z_2, z_3),$

introduced in (1.2) in Chapter 1. The kernel K here is supposed to be odd and real-valued. Then by (1.2), (1.3) and (1.4), we also have

$$p_K(\mu_1,\mu_2,\mu_3) = \frac{1}{2}P_K(\mu_1,\mu_2,\mu_3), \qquad p_{K,\varepsilon}(\mu_1,\mu_2,\mu_3) = \frac{1}{2}P_{K,\varepsilon}(\mu_1,\mu_2,\mu_3)$$

and

$$p_K(\mu) = \frac{1}{2} p_K(\mu, \mu, \mu), \qquad p_{K,\varepsilon}(\mu) = \frac{1}{2} P_{K,\varepsilon}(\mu, \mu, \mu),$$

where μ , μ_1 , μ_2 and μ_3 are measures.

Recall that by a measure we mean a positive locally finite Borel measure on \mathbb{C} .

Some of the identities from Introduction have a slightly different form in these terms. In particular, let K be an odd and real-valued CZ kernel with the permutations (1.2), being non-negative for any $(z_1, z_2, z_3) \in \mathbb{C}^3$. If μ is a finite measure with C_* -linear growth, i.e. there exists a constant $C_* > 0$ such that

$$\mu(B(z,r)) \leqslant C_*r$$
 for all $z \in \operatorname{spt} \mu$,

then the generalised Melnikov-Verdera identity (0.13) takes the form:

$$|T_{K,\varepsilon}1||_{L^{2}(\mu)}^{2} = \frac{1}{3}p_{K,\varepsilon}(\mu) + \mathcal{R}_{K,\varepsilon}(\mu), \qquad |\mathcal{R}_{K,\varepsilon}(\mu)| \leqslant cC_{*}^{2}\mu(\mathbb{C}), \qquad (3.1)$$

where $\varepsilon > 0$ and c > 0 is independent of ε . What is more, (0.15) becomes

$$p_{k_{\infty}}(z_1, z_2, z_3) = \frac{1}{4}c(z_1, z_2, z_3)^2, \qquad (z_1, z_2, z_3) \in \mathbb{C}^3.$$

Clearly, this implies that $p_{k_{\infty}}(\mu) = \frac{1}{4}c^2(\mu)$ for any measure μ .

Recall also that it is shown in Theorem 1.1 and Remark 1.1 in Chapter 1 that

$$\begin{cases} p_{k_t}(z_1, z_2, z_3) \ge 0 \text{ for any } (z_1, z_2, z_3) \in \mathbb{C}^3, \text{ if } t \notin (-2, 0), \\ p_{k_t}(z_1, z_2, z_3) \text{ may change sign for some } (z_1, z_2, z_3) \in \mathbb{C}^3, \text{ if } t \in (-2, 0), \end{cases}$$

see Figure 0.2. Moreover, by (2.5) from Lemma 2.2 proved in Chapter 2,

$$p_{k_0}(z_1, z_2, z_3) \leq 2p_{k_\infty}(z_1, z_2, z_3)$$
 for any $(z_1, z_2, z_3) \in \mathbb{C}^3$.

Now we give several definitions related to β -numbers and densities. For any closed ball B = B(x, r) with center $x \in \mathbb{C}$ and radius r > 0 and $1 \leq p < \infty$, let

$$\beta_{\mu,p}(B) = \inf_{L} \left(\frac{1}{r} \int_{B} \left(\frac{\operatorname{dist}(y,L)}{r} \right)^{p} d\mu(y) \right)^{1/p}, \qquad (3.2)$$

where the infimum is taken over all affine lines $L \subset \mathbb{C}$. The $\beta_{\mu,p}$ coefficients were introduced by David and Semmes [DS1] and are the generalization of the well-known Jones β -numbers [Jon2].

We will mostly deal with $\beta_{\mu,2}(2B_Q)$ and so by L_Q we denote a corresponding best approximating line, i.e. a line where the infimum is reached in (3.2) for $B = 2B_Q$ (see the definition of B_Q below) and p = 2.

Throughout the chapter we also use the following densities:

$$\Theta_{\mu}(B) := \Theta_{\mu}(x, r) = \frac{\mu(B(x, r))}{r}, \quad \text{where} \quad B = B(x, r), \qquad x \in \mathbb{C}, \qquad r > 0.$$

First we prove the following.

Theorem 3.1. There exist absolute constants $t_0 > 0$ and c > 0 such that for any finite measure μ with C_* -linear growth it holds that

$$\sup_{\varepsilon>0} \|T_{k_{\infty},\varepsilon}1\|_{L^{2}(\mu)} \leqslant t_{0}^{-1} \sup_{\varepsilon>0} \|T_{k_{0},\varepsilon}1\|_{L^{2}(\mu)} + cC_{*}\sqrt{\mu(\mathbb{C})}.$$
(3.3)

Roughtly speaking, this means that the L^2 -norm of the Cauchy transform of measure is controlled by the L^2 -norm of the operator T_{k_0} .

Note that by Lemma 2.3 in Chapter 2 under the same assumptions on μ ,

$$||T_{k_0,\varepsilon}1||_{L^2(\mu)} \leqslant \sqrt{2} ||T_{k_\infty,\varepsilon}1||_{L^2(\mu)} + cC_*\sqrt{\mu(\mathbb{C})},$$

where $\varepsilon > 0$ and c > 0 is independent of ε . With respect to the proof of Lemma 2.3, the proof of (3.3) is more difficult as we will see in this chapter.

As a corollary of Theorem 3.1, using the same t_0 , we obtain the following result.

Theorem 3.2. Let k_t be a kernel of the form (0.21), where $t \in (-t_0, 0)$, and E a 1-set. If the SIO T_{k_t} is $L^2(\mathcal{H}^1 | E)$ -bounded, then E is rectifiable.

This theorem complements Theorems B, D, E, F and Theorems 1.2 and 2.2 so that we have the overall picture as in Figure 0.2. It is clear from Theorem F that necessarily $t_0 \in (0, 3/4)$. What is more, it is very important here that the pointwise permutations, corresponding to $t \in (-t_0, 0)$, also change sign as in Theorem 2.2 so that the curvature method cannot be applied directly but L^2 -boundedness still implies rectifiability.

Remark 3.1. By simple analysis one can show that the kernel k_t has

 $\left\{ \begin{array}{l} \text{ one zero line if } t \in (-\infty, -1) \cup [0, \infty], \\ \text{ two zero lines if } t = -1, \\ \text{ three zero lines if } t \in (-1, 0). \end{array} \right.$

By a zero line we mean a straight line $L \subset \mathbb{C}$ such that $k_t(z) = 0$ for $z \in L$.

In this sense, it is interesting to compare Theorem 3.2 with Theorem F. Observing Figure 0.2, one can see that the number of zero lines along is not determinant.

Remark 3.2. Let t_1 and t_2 be such that $-\sqrt{2} \leq t_1 < t_2 \leq -t_0$. If there exist finite *purely unrectifiable* (i.e. concentrated on purely unrectifiable sets) measures μ_1 and μ_2 with linear growth such that $T_{k_{t_1}}$ is $L^2(\mu_1)$ -bounded and $T_{k_{t_2}}$ is $L^2(\mu_2)$ -bounded, then μ_1 is different from μ_2 .

Indeed, let μ be a finite purely unrectifiable measure with linear growth such that $T_{k_{\tilde{t}}}$ is $L^2(\mu)$ -bounded for a fixed $\tilde{t} \in [-\sqrt{2}, -t_0]$. By the triangle inequality for any real t,

$$||T_{k_t}1||_{L^2(\mu)} = ||(T_{k_0} + (t - \tilde{t}) \cdot T_{k_\infty} + \tilde{t} \cdot T_{k_\infty})1||_{L^2(\mu)} \ge |t - \tilde{t}|||T_{k_\infty}1||_{L^2(\mu)} - ||T_{k_t}1||_{L^2(\mu)} -$$

Consequently, $||T_{k_t}1||_{L^2(\mu)} = \infty$ for all $t \neq \tilde{t}$ as $||T_{k_\infty}1||_{L^2(\mu)} = \infty$ since μ is purely unrectifiable. Thus an example of a purely unrectifiable measure μ such that $T_{k_{\tilde{t}}}$ is $L^2(\mu)$ -bounded for a fixed $\tilde{t} \in [-\sqrt{2}, -t_0]$ does not work for $t \neq \tilde{t}$.

3.2 Main Lemma and proofs of Theorems

Theorem 3.1 is implied by the following lemma.

Main Lemma. There exist absolute constants $t_0 > 0$ and c > 0 such that for any finite measure μ with C_* -linear growth it holds that

$$p_{k_{\infty}}(\mu) \leq t_0^{-2} p_{k_0}(\mu) + c C_*^2 \mu(\mathbb{C}).$$
 (3.4)

The proof of this result is long and technical and actually takes the biggest part of this chapter. Note that (3.4) is a counterpart to the inequality $p_{k_0}(\mu) \leq 2p_{k_{\infty}}(\mu)$ that follows from (2.5).

3.2.1 Proof of Theorem 3.1

Suppose that Main Lemma holds. Then the identity (3.1) and inequality (3.4) yield

$$\begin{split} \sup_{\varepsilon > 0} \| T_{k_{\infty},\varepsilon} 1 \|_{L^{2}(\mu)}^{2} &\leq \frac{1}{3} p_{k_{\infty}}(\mu) + c C_{*}^{2} \mu(\mathbb{C}) \\ &\leq \frac{1}{3} t_{0}^{-2} p_{k_{0}}(\mu) + c C_{*}^{2} \mu(\mathbb{C}) \\ &\leq t_{0}^{-2} \sup_{\varepsilon > 0} \| T_{k_{0},\varepsilon} 1 \|_{L^{2}(\mu)}^{2} + c C_{*}^{2} \mu(\mathbb{C}), \end{split}$$

where c > 0 is an absolute constant. Applying the inequality $\sqrt{ax^2 + b} \leq \sqrt{ax} + \sqrt{b}$ that is valid for $a, b, x \geq 0$, gives Theorem 3.1.

3.2.2 Proof of Theorem 3.2

We now apply the perturbation method from Chapter 2. By the triangle inequality and Theorem 3.1,

$$\sup_{\varepsilon > 0} \|T_{k_t,\varepsilon} 1\|_{L^2(\mu)} = \sup_{\varepsilon > 0} \|(T_{k_0,\varepsilon} + t \cdot T_{k_\infty,\varepsilon}) 1\|_{L^2(\mu)}$$

$$\geq \sup_{\varepsilon > 0} \|T_{k_0,\varepsilon} 1\|_{L^2(\mu)} - |t| \sup_{\varepsilon > 0} \|T_{k_\infty,\varepsilon} 1\|_{L^2(\mu)}$$

$$\geq (t_0 - |t|) \sup_{\varepsilon > 0} \|T_{k_\infty,\varepsilon} 1\|_{L^2(\mu)} - cC_* \sqrt{\mu(\mathbb{C})}.$$

Consequently,

$$\sup_{\varepsilon > 0} \|T_{k_{\infty},\varepsilon} 1\|_{L^{2}(\mu)} \leqslant \frac{\sup_{\varepsilon > 0} \|T_{k_{t},\varepsilon} 1\|_{L^{2}(\mu)} + cC_{*}\sqrt{\mu(\mathbb{C})}}{t_{0} - |t|}, \qquad |t| < t_{0}$$

and therefore for any cube $Q \subset \mathbb{C}$ we have

$$\sup_{\varepsilon > 0} \|T_{k_{\infty},\varepsilon}\chi_Q\|_{L^2(\mu \lfloor Q)} \leqslant \frac{\sup_{\varepsilon > 0} \|T_{k_t,\varepsilon}\chi_Q\|_{L^2(\mu \lfloor Q)} + cC_*\sqrt{\mu(Q)}}{t_0 - |t|}, \qquad |t| < t_0.$$
(3.5)

Applying a variant of the T1 Theorem of Nazarov, Treil and Volberg from [Tol5, Theorem 9.40], we infer from (3.5) with $\mu = \mathcal{H}^1 \lfloor E$ and 1-set E that the $L^2(\mathcal{H}^1 \lfloor E)$ -boundedness of T_{k_t} with a fixed t such that $|t| < t_0$ implies that $T_{k_{\infty}}$ and thus the Cauchy transform is $L^2(\mathcal{H}^1 \lfloor E)$ -bounded. Finally, Theorem B gives the desired result.

3.3 Several corollaries

Recall that a measure μ is Ahlfors-David regular (AD-regular) if it satisfies (0.4), i.e.

$$C^{-1}r \leq \mu(B(z,r)) \leq Cr$$
, where $z \in \operatorname{spt} \mu$, $0 < r < \operatorname{diam}(\operatorname{spt} \mu)$,

and C > 1 is some fixed constant. A measure μ is called *uniformly rectifiable* if it is ADregular and spt μ is contained in an AD-regular curve. One can summarise all up-to-date results characterising uniformly rectifiable measures via $L^2(\mu)$ -bounded SIOs T_{k_t} as follows.

Corollary 3.1. Let μ be an AD-regular measure and k_t a kernel of the form (0.21) with $t \in (-\infty, -\sqrt{2}) \cup (-t_0, \infty]$. The measure μ is uniformly rectifiable if and only if the SIO T_{k_t} is $L^2(\mu)$ -bounded.

The part of Corollary 3.1 for $t = \infty$, i.e. for the Cauchy transform, was proved in [MMV]; for t = 0 in [CMPT1]; and for $t \in (-\infty, -\sqrt{2}) \cup (0, \infty)$ in Chapters 1 and 2.

Furthermore, one can formulate the following general result.

Corollary 3.2. Let μ be a measure with linear growth and k_t a kernel of the form (0.21) with $t \in (-\infty, -\sqrt{2}) \cup (-t_0, \infty]$. If the SIO T_{k_t} is $L^2(\mu)$ -bounded, then so are all 1-dimensional SIOs associated with a wide class of kernels.

The above-mentioned "wide class of kernels" was defined around (0.23) in Introduction. The part of Corollary 3.2 for $t = \infty$, i.e. for the Cauchy transform, was proved in [Tol2, Tol4] (see also [Gir]) and for $t \in (-\infty, -\sqrt{2}) \cup (0, \infty)$ in Chapters 1 and 2.

3.4 Plan of the proof of Main Lemma

To prove Main Lemma, we will use a corona decomposition that is similar, for example, to the ones from [Tol6] and [AT]: it splits the David-Mattila dyadic lattice into some collections of cubes, which we will call "trees", where the density of μ does not oscillate too much and most of the measure is concentrated close to a graph of a Lipschitz function. To construct this function we will use a variant of the Whitney extension theorem adapted to the David-Mattila dyadic lattice. Further, we will show that the family of trees of the corona decomposition satisfies a packing condition by arguments inspired by some of the techniques used in [AT] and earlier in [Tol3] to prove the bilipschitz "invariance" of analytic capacity. More precisely, we will deduce Main Lemma from the two-sided estimate

$$p_{k_{\infty}}(\mu) \lesssim \sum_{R \in \mathsf{Top}} \Theta_{\mu}(2B_R)^2 \mu(R) \lesssim p_{k_0}(\mu) + C_*^2 \mu(\mathbb{C}),$$
(3.6)

where Top is the family of top cubes for the above-mentioned trees. Note that the left hand side inequality in (3.6) in essentially contained in [Tol6] and verifying the right hand side inequality is actually the main objective in the proof.

It is worth mentioning that the structure of our trees is more complicated than in [AT]. This is because we deal with permutations which are not comparable to curvature in some cases and this leads to additional technical difficulties. What is more, we are not able to use a nice theorem by David and Toro [DT] which shortens the proof in [AT] considerably. Indeed, this theorem would be useful to construct a chordal curve such that most of the measure μ is concentrated close to it. However, in our situation we need to control slope and therefore we have to deal with and to construct a graph of a Lipschitz function with well-controlled Lipschitz constant instead.

The plan of the proof of Main Lemma is the following. In Section 3.5 we recall the properties of the David-Mattila dyadic lattice. We construct the trees and establish their properties in Sections 3.6–3.12. The main properties are summarized in Section 3.13, where they are further used for constructing the corona type decomposition. The end of the proof of Main Lemma is given in Section 3.13.6.

Finally, in Sections 3.14 and 3.15 we show how one can slightly change the proof of Main Lemma in order to give another proof of a certain result from [AT] and also to extend the results in this chapter for a more general class of kernels.

Remark 3.3. In the forthcoming proof, we will usually write p_t instead of p_{k_t} in order to simplify notation.

Remark 3.4. The measure μ considered below is under assumptions of Main Lemma, i.e. μ is a finite measure with C_* -linear growth. Moreover, without loss of generality we additionally suppose that μ has compact support.

3.5 The David-Mattila lattice

We use the dyadic lattice of cubes with small boundaries constructed by David and Mattila [DM]. The properties of this lattice are summarized in the next lemma (for the case of \mathbb{C}).

Lemma 3.1 (Theorem 3.2 in [DM]). Let μ be a measure, $E = \operatorname{spt} \mu$, and consider two constants $C_0 > 1$ and $A_0 > 5000 C_0$. Then there exists a sequence of partitions of E into Borel subsets $Q, Q \in \mathcal{D}_k$, with the following properties:

- For each integer $k \ge 0$, E is the disjoint union of the "cubes" $Q, Q \in \mathcal{D}_k$, and if $k < l, Q \in \mathcal{D}_l$, and $R \in \mathcal{D}_k$, then either $Q \cap R = \emptyset$ or else $Q \subset R$.
- The general position of the cubes Q can be described as follows. For each $k \ge 0$ and each cube $Q \in \mathcal{D}_k$, there is a ball $B(Q) = B(z_Q, r(Q))$ such that

$$z_Q \in Q, \qquad A_0^{-k} \leqslant r(Q) \leqslant C_0 A_0^{-k},$$
$$E \cap B(Q) \subset Q \subset E \cap 28 B(Q) = E \cap B(z_Q, 28r(Q)),$$

and

the balls
$$5B(Q)$$
, $Q \in \mathcal{D}_k$, are disjoint.

• The cubes $Q \in \mathcal{D}_k$ have small boundaries. That is, for each $Q \in \mathcal{D}_k$ and each integer $l \ge 0$, set

$$N_l^{ext}(Q) = \{ x \in E \setminus Q : \text{ dist}(x, Q) < A_0^{-k-l} \},\$$
$$N_l^{int}(Q) = \{ x \in Q : \text{ dist}(x, E \setminus Q) < A_0^{-k-l} \},\$$

and

$$N_l(Q) = N_l^{ext}(Q) \cup N_l^{int}(Q).$$

Then

$$\mu(N_l(Q)) \leqslant (C^{-1}C_0^{-7}A_0)^{-l}\,\mu(90B(Q))$$

• Denote by \mathcal{D}_k^{db} the family of cubes $Q \in \mathcal{D}_k$ for which

$$\mu(100B(Q)) \leqslant C_0 \,\mu(B(Q)). \tag{3.7}$$

If
$$Q \in \mathcal{D}_k \setminus \mathcal{D}_k^{db}$$
, then $r(Q) = A_0^{-k}$ and
 $\mu(100B(Q)) \leqslant C_0^{-l} \mu(100^{l+1}B(Q))$ for all $l \ge 1$ such that $100^l \leqslant C_0$.

We use the notation $\mathcal{D} = \bigcup_{k \ge 0} \mathcal{D}_k$. For $Q \in \mathcal{D}$, we set $\mathcal{D}(Q) = \{P \in \mathcal{D} : P \subset Q\}$. Observe that

 $r(Q) \approx \operatorname{diam}(Q).$

Also we call z_Q the center of Q. We set $B_Q = 28 B(Q) = B(z_Q, 28 r(Q))$, so that

$$E \cap \frac{1}{28} B_Q \subset Q \subset B_Q.$$

We denote $\mathcal{D}^{db} = \bigcup_{k \ge 0} \mathcal{D}_k^{db}$ and $\mathcal{D}^{db}(Q) = \mathcal{D}^{db} \cap \mathcal{D}(Q)$. Note that, in particular, from (3.7) it follows that

$$\mu(100B(Q)) \leqslant C_0 \,\mu(2B_Q) \qquad \text{if } Q \in \mathcal{D}^{db}.$$
(3.8)

For this reason we will call the cubes from \mathcal{D}^{db} doubling.

As shown in [DM], any cube $Q \in \mathcal{D}$ can be covered μ -a.e. by doubling cubes.

Lemma 3.2 (Lemma 5.28 in [DM]). Let $Q \in \mathcal{D}$. Suppose that the constants A_0 and C_0 in Lemma 3.1 are chosen suitably. Then there exists a family of doubling cubes $\{Q_i\}_{i \in I} \subset \mathcal{D}^{db}$, with $Q_i \subset Q$ for all i, such that their union covers μ -almost all Q.

We denote by J(Q) the number k such that $Q \in \mathcal{D}_k$.

Lemma 3.3 (Lemma 5.31 in [DM]). Let $P \in \mathcal{D}$ and let $Q \subsetneq P$ be a cube such that all the intermediate cubes $S, Q \subsetneq S \subsetneq P$, are non-doubling (i.e. not in \mathcal{D}^{db}). Then

$$\mu(100B(Q)) \leqslant A_0^{-20(J(Q) - J(P) - 1)} \mu(100B(P)).$$

Recall that $\Theta_{\mu}(B) = \mu(B(x,r))/r$. From Lemma 3.3 one can easily deduce¹

Lemma 3.4 (Lemma 2.4 in [AT]). Let $Q, P \in \mathcal{D}$ be as in Lemma 3.3. Then

$$\Theta_{\mu}(100B(Q)) \leqslant C_0 A_0^{-19(J(Q)-J(P)-1)+1} \Theta_{\mu}(100B(P)) \leqslant C_0 A_0 \Theta_{\mu}(100B(P))$$

and

$$\sum_{S \in \mathcal{D}: Q \subset S \subset P} \Theta_{\mu}(100B(S)) \leqslant c \,\Theta_{\mu}(100B(P)), \qquad c = c(C_0, A_0).$$

We will assume that all implicit constants in the inequalities that follow may depend on C_0 and A_0 . Moreover, we will assume that C_0 and A_0 are some big *fixed* constants so that the results stated in the lemmas below hold.

3.6 Balanced cubes and control on beta numbers through permutations

We first recall the properties of the so called balanced balls introduced in [AT].

Lemma 3.5 (Lemma 3.3 and Remark 3.2 in [AT]). Let μ be a measure and consider the dyadic lattice \mathcal{D} associated with μ from Lemma 3.1. Let $0 < \gamma < 1$ be small enough (with respect to some absolute constant), then there exist $\rho' = \rho'(\gamma) > 0$ and $\rho'' = \rho''(\gamma) > 0$ such that one of the following alternatives holds for every $Q \in \mathcal{D}^{db}$:

(a) There are balls $B_k = B(\xi_k, \rho' r(Q)), k = 1, 2$, where $\xi_1, \xi_2 \in B(Q)$, such that

$$\mu(B_k \cap B(Q)) \ge \rho'' \mu(Q), \qquad k = 1, 2,$$

and for any $y_k \in B_k \cap Q$, k = 1, 2,

$$\operatorname{dist}\left(y_1, y_2\right) \geqslant \gamma \, r(B_Q)$$

(b) There exists a family of pairwise disjoint cubes $\{P\}_{P \in I_Q} \subset \mathcal{D}^{db}(Q)$ so that diam $(P) \gtrsim \gamma \operatorname{diam}(Q)$ and $\Theta_{\mu}(2B_P) \gtrsim \gamma^{-1} \Theta_{\mu}(2B_Q)$ for each $P \in I_Q$, and

$$\sum_{P \in I_Q} \Theta_\mu (2B_P)^2 \,\mu(P) \gtrsim \gamma^{-2} \,\Theta_\mu (2B_Q)^2 \,\mu(Q). \tag{3.9}$$

Let us mention that the densities in the latter inequality in the original Lemma 3.3 in [AT] are not squared. However, a slight variation of the proof of [AT, Lemma 3.3] gives (3.9) as stated.

Moreover, notice that in Lemma 3.5 the cubes Q and P, with $P \in I_Q$, are doubling. If the alternative (a) holds for a doubling cube Q with some γ , $\rho'(\gamma)$ and $\rho''(\gamma)$, then the corresponding ball B(Q) is called γ -balanced. Otherwise, it is called γ -unbalanced. If B(Q) is γ -balanced, then the cube Q is also called γ -balanced.

We are going to show now that the beta numbers $\beta_{\mu,2}(2B_Q)$ (see (3.2)) for γ -balanced cubes Q are controlled by a truncated version of the permutations $p_0(\mu \lfloor 2B_Q)$. To do so, we introduce some additional notation.

Given two distinct points $z, w \in \mathbb{C}$, we denote by $L_{z,w}$ the line passing through z and w. Given three pairwise distinct points $z_1, z_2, z_3 \in \mathbb{C}$, we denote by $\measuredangle(z_1, z_2, z_3)$ the smallest

¹Note that there is an inaccuracy with constants in the original Lemma 2.4 in [AT].

angle formed by the lines L_{z_1,z_2} and L_{z_1,z_3} and belonging to $[0, \pi/2]$. If L and L' are lines, let $\measuredangle(L, L')$ be the smallest angle between them. This angle belongs to $[0, \pi/2]$, too. Also, we set $\theta_V(L) = \measuredangle(L, V)$, where V is the vertical line.

First we recall the following result of Chousionis and Prat [CP]. We say that a triple $(z_1, z_2, z_3) \in \mathbb{C}^3$ is in the class $V_{\mathsf{Far}}(\theta)$ if it satisfies

$$\theta_V(L_{z_1, z_2}) + \theta_V(L_{z_1, z_3}) + \theta_V(L_{z_2, z_3}) \ge \theta > 0.$$
(3.10)

Lemma 3.6 (Proposition 3.3 in [CP]). If $(z_1, z_2, z_3) \in V_{\mathsf{Far}}(\theta)$, then

$$p_0(z_1, z_2, z_3) \ge c_1(\theta) \cdot p_\infty(z_1, z_2, z_3), \quad where \quad 0 < c_1(\theta) \le 2.$$
 (3.11)

Note that the inequality $c_1(\theta) \leq 2$ follows from (2.5) that was proved in [CMT1]. For measures μ_1 , μ_2 and μ_3 and a cube Q we set

$$p_0^{[\delta,Q]}(\mu_1,\mu_2,\mu_3) := \iiint_{\delta r(Q) \leqslant |z_1 - z_2| \leqslant \delta^{-1} r(Q)} p_0(z_1,z_2,z_3) \ d\mu_1(z_1) d\mu_2(z_2) d\mu_3(z_3).$$

The parameter $\delta > 0$ will be chosen later to be small enough for our purposes. If $\mu_1 = \mu_2 = \mu_3 = \mu$, then we write $p_0^{[\delta,Q]}(\mu)$ instead of $p_0^{[\delta,Q]}(\mu,\mu,\mu)$, for short.

Now we are ready to state the above mentioned estimate of $\beta_{\mu,2}(2B_Q)$ for γ -balanced cubes Q via the truncated version of $p_0(\mu \lfloor 2B_Q)$. Pay attention that the first term in the estimate is a "non-summable" part which makes a big difference with the case of curvature or p_{∞} (see Section 3.14).

Lemma 3.7. If Q is γ -balanced, then for any $\varepsilon \in (0, 1)$,

$$\beta_{\mu,2}(2B_Q)^2 \Theta_{\mu}(2B_Q) \leqslant 4\varepsilon^2 \Theta_{\mu}(2B_Q)^2 + C(\varepsilon,\gamma) \frac{p_0^{[\delta,Q]}(\mu \lfloor 2B_Q)}{\mu(Q)}, \qquad 0 < \delta \leqslant \gamma.$$
(3.12)

Moreover, for any $\varepsilon_0 > 0$, there exist $\varepsilon = \varepsilon(\varepsilon_0) > 0$ and $\tilde{\varepsilon} = \tilde{\varepsilon}(\varepsilon_0, \gamma) > 0$ such that if

$$\frac{p_0^{[\delta,Q]}(\mu\lfloor 2B_Q)}{\Theta_\mu(2B_Q)^2\mu(Q)} \leqslant \tilde{\varepsilon},\tag{3.13}$$

then

$$\beta_{\mu,2}(2B_Q)^2 \leqslant \varepsilon_0^2 \Theta_\mu(2B_Q).$$

Proof. By Lemma 3.5, there exist balls $B_k = B(\xi_k, \rho' r(Q)), k = 1, 2$, where $\xi_k \in B(Q)$, such that $\mu(B_k \cap B(Q)) \ge \rho'' \mu(Q)$ and dist $(y_1, y_2) \ge \gamma r(B_Q)$ for any $y_k \in B_k \cap Q, k = 1, 2$. From (3.2) it follows that

$$\beta_{\mu,2}(2B_Q)^2 \leqslant \frac{1}{2r(B_Q)} \int_{2B_Q} \left(\frac{\operatorname{dist}(w, L_{y_1, y_2})}{2r(B_Q)}\right)^2 d\mu(w).$$

We separate triples (w, y_1, y_2) that are in and not in $V_{\mathsf{Far}}(\varepsilon)$. Clearly,

$$\operatorname{dist}(w, L_{y_1, y_2}) \leqslant \operatorname{diam}(2B_Q) \operatorname{sin} \varepsilon \leqslant 4 \varepsilon r(B_Q) \qquad \text{if } (w, y_1, y_2) \notin \operatorname{V}_{\mathsf{Far}}(\varepsilon).$$

Thus

$$\begin{split} &\beta_{\mu,2}(2B_Q)^2 \\ &\leqslant \frac{4\varepsilon^2}{2r(B_Q)} \int_{2B_Q} d\mu(w) + \frac{1}{2r(B_Q)} \int_{2B_Q, \ (w,y_1,y_2) \in \mathcal{V}_{\mathsf{Far}}(\varepsilon)} \left(\frac{\operatorname{dist}(w,L_{y_1,y_2})}{2r(B_Q)}\right)^2 d\mu(w) \\ &\leqslant 4\varepsilon^2 \Theta_\mu(2B_Q) + 8r(B_Q) \int_{2B_Q, \ (w,y_1,y_2) \in \mathcal{V}_{\mathsf{Far}}(\varepsilon)} \left(\frac{2\operatorname{dist}(w,L_{y_1,y_2})}{|w-y_1||w-y_2|}\right)^2 d\mu(w) \\ &= 4\varepsilon^2 \Theta_\mu(2B_Q) + 8r(B_Q) \int_{2B_Q, \ (w,y_1,y_2) \in \mathcal{V}_{\mathsf{Far}}(\varepsilon)} c(w,y_1,y_2)^2 d\mu(w). \end{split}$$

We used that $|w - y_k| \leq \text{diam}(2B_Q) = 4r(B_Q)$ as $w, y_1, y_2 \in 2B_Q$ and that

$$c(w, y_1, y_2) = \frac{2 \operatorname{dist} (w, L_{y_1, y_2})}{|w - y_1| |w - y_2|}$$

Recall that $r(B_Q) = 28r(Q)$ by definition. By (0.15) and (3.11),

$$\begin{split} \int_{2B_Q, \ (w,y_1,y_2)\in \mathcal{V}_{\mathsf{Far}}(\varepsilon)} c(w,y_1,y_2)^2 d\mu(w) \\ \leqslant \frac{2}{\mathsf{c}_1(\varepsilon)} \int_{2B_Q, \ (w,y_1,y_2)\in \mathcal{V}_{\mathsf{Far}}(\varepsilon)} p_0(w,y_1,y_2) d\mu(w). \end{split}$$

Recall that $|y_1 - y_2| \ge \gamma r(Q)$ for any $y_k \in B_k \cap Q$, k = 1, 2. Furthermore, for any δ such that $0 < \delta \le \gamma$ we can find $y_1 \in B_1$ and $y_2 \in B_2$ so that

$$\int_{2B_Q} p_0(w, y_1, y_2) d\mu(w) \leqslant \frac{p_0^{[\delta, Q]}(\mu \lfloor 2B_Q)}{\mu(B_1)\mu(B_2)} \leqslant \frac{p_0^{[\delta, Q]}(\mu \lfloor 2B_Q)}{(\rho'')^2 \mu(Q)^2}.$$

By (3.7) and the fact that $E \cap B(Q) \subset Q$, we deduce that

$$\mu(Q) \ge C_0^{-1}\mu(100B(Q)) \ge C_0^{-1}\mu(56B(Q)) = C_0^{-1}\mu(2B_Q).$$

Consequently,

$$\beta_{\mu,2}(2B_Q)^2 \leqslant 4\varepsilon^2 \Theta_{\mu}(2B_Q) + \frac{16r(B_Q)p_0^{[\delta,Q]}(\mu\lfloor 2B_Q)}{\mathsf{c}_1(\varepsilon)(\rho'')^2\mu(Q)C_0^{-1}\mu(2B_Q)}$$
$$= 4\varepsilon^2 \Theta_{\mu}(2B_Q) + C(\varepsilon,\gamma)\frac{p_0^{[\delta,Q]}(\mu\lfloor 2B_Q)}{\Theta_{\mu}(2B_Q)\mu(Q)}.$$

Multiplying both sides by $\Theta_{\mu}(2B_Q)$ finishes the proof of (3.12). Note that $\rho'' = \rho''(\gamma)$.

Let us prove the second statement. By the assumption (3.13),

$$\beta_{\mu,2}(2B_Q)^2 \leqslant (4\varepsilon^2 + C(\varepsilon,\gamma)\tilde{\varepsilon})\Theta_{\mu}(2B_Q).$$

For any $\varepsilon_0 > 0$, we put $\varepsilon = \frac{\sqrt{2}}{4}\varepsilon_0$ and choose $\tilde{\varepsilon}$ so that $\tilde{\varepsilon} \leq \frac{1}{2}\varepsilon_0^2/C(\varepsilon,\gamma)$.

3.7 Parameters and thresholds

Recall that we work everywhere with the David-Mattila dyadic lattice \mathcal{D} associated with the measure μ .

In what follows we will use many parameters and thresholds. Some of them depend on each other, some are independent. Let us give a list of the parameters: • τ is the threshold for cubes with low density:

$$0<\tau\ll 1.$$

• A is the threshold for cubes with high density:

$$0 < A^{-1} \le \tau^2 \ll 1$$
, i.e. $A \gg 1$.

• θ_0 is the threshold for the angle between best approximating lines associated to some cubes:

$$0 < \theta_0 \ll 1.$$

• γ is the parameter controlling unbalanced cubes:

$$0 < \gamma \leqslant \tau^3 \ll 1.$$

• ε_0 is the threshold controlling the $\beta_{2,\mu}$ -numbers:

$$0 < \varepsilon_0 = \varepsilon_0(\gamma, \tau, A, \theta_0) \ll 1.$$

• α is the threshold controlling permutations of intermediate cubes:

$$0 < \alpha = \alpha(\tau, A, \varepsilon_0, \gamma, \theta_0) \ll 1.$$

• δ is the parameter controlling the truncation of permutations:

$$0 < \delta = \delta(\gamma, \varepsilon_0, \tau, A) \ll 1.$$

All the parameters and thresholds are supposed to be chosen (and fixed at the very end) so that the forthcoming results hold true. In what follows, we will again indicate step by step how the choice should be made.

3.8 Stopping cubes and trees

3.8.1 Stopping cubes

Let $R \in \mathcal{D}^{db}$. We use the parameters and thresholds given in Section 3.7. We denote by $\mathsf{Stop}(R)$ the family of the maximal cubes $Q \subset R$ for which one of the following holds:

(S1) $Q \in HD(R) \cup LD(R) \cup UB(R)$, where

• HD(R) is the family of high density doubling cubes $Q \in \mathcal{D}^{db}$ satisfying

$$\Theta_{\mu}(2B_Q) > A \Theta_{\mu}(2B_R);$$

• LD(R) is the family of *low density* cubes Q satisfying

$$\Theta_{\mu}(2B_Q) < \tau \,\Theta_{\mu}(2B_R);$$

• $\mathsf{UB}(R)$ is the family of *unbalanced* cubes $Q \in \mathcal{D}^{db} \setminus (\mathsf{HD}(R) \cup \mathsf{LD}(R))$ such that Q is γ -unbalanced;

(S2) $Q \in \mathsf{BP}(R)$ ("big permutations"), meaning $Q \notin \mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{UB}(R)$ and

$$\sum_{Q \subset \tilde{Q} \subset R} \mathsf{perm}(\tilde{Q})^2 > \alpha^2, \qquad \mathsf{perm}(\tilde{Q})^2 := \frac{p_0^{[\delta,Q]}(\mu \lfloor 2B_{\tilde{Q}}, \mu \lfloor 2B_R, \mu \lfloor 2B_R)}{\Theta_{\mu}(2B_R)^2 \mu(\tilde{Q})}$$

(S3) $Q \in \mathsf{BS}(R)$ ("big slope"), meaning $Q \notin \mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{UB}(R) \cup \mathsf{BP}(R)$ and $Q \in \mathcal{D}^{db}$ so that

$$\measuredangle(L_Q, L_R) > \theta(R),$$

where $\theta(R)$ depends on some geometric properties of R and is comparable with the parameter $\theta_0 > 0$ mentioned in Section 3.7. The more precise description will be given in Section 3.12.

(S4) $Q \in F(R)$ ("big part of Q is far from best approximating lines for the doubling ancestors of Q"), meaning $Q \notin HD(R) \cup LD(R) \cup UB(R) \cup BP(R) \cup BS(R)$ and

$$\mu(Q \setminus 2B_Q^{\mathsf{Cl}}) > \sqrt{\alpha}\,\mu(Q),$$

where

$$2B_Q^{\mathsf{CI}} := \{ x \in R \cap 2B_Q : \text{ dist} (x, L_{\tilde{Q}}) \leq 5\sqrt{\varepsilon_0} r(B_{\tilde{Q}}) \quad \forall \tilde{Q} \in \mathcal{D}^{db}(R) :$$
$$2B_Q \subset 2B_{\tilde{Q}} \text{ and } \tilde{Q} \text{ is not contained in any cube from}$$
(3.14)
$$\mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{UB}(R) \cup \mathsf{BP}(R) \cup \mathsf{BS}(R) \}.$$

Let $\mathsf{Tree}(R)$ be the subfamily of the cubes from $\mathcal{D}(R)$ which are not *strictly* contained in any cube from $\mathsf{Stop}(R)$. We also set

$$\mathsf{DbTree}(R) := \mathcal{D}^{db} \cap (\mathsf{Tree}(R) \setminus \mathsf{Stop}(R)).$$

Note that all cubes in Stop(R) are disjoint.

Remark 3.5. It may happen that $\mathsf{Stop}(R)$ is empty. In this case there is no need to estimate the measure of stopping cubes and we may immediately go to Section 3.10. In the lemmas below related to estimating the measure of stopping cubes we naturally suppose that $\mathsf{Stop}(R)$ is not empty.

Generally speaking it is possible that $R \in \text{Stop}(R)$ (and then DbTree(R) is empty). Clearly, $R \notin \text{HD}(R) \cup \text{LD}(R) \cup \text{BS}(R)$ by definition but it may occur that $R \in \text{UB}(R) \cup \text{BP}(R) \cup \text{F}(R)$. Firstly, we will not work with the family UB(R) before Section 3.13 so we may assume before that section that $R \notin \text{UB}(R)$. Secondly, if $R \in \text{BP}(R)$, then we may directly go to Lemma 3.14 and use the same estimate for the measure of stopping cubes from BP(R). Thirdly, it will follow from Lemmas 3.12 and 3.13 (see Remark 3.6) that if $R \notin \text{UB}(R) \cup \text{BP}(R)$, then $R \notin \text{F}(R)$, i.e. the case $R \in \text{F}(R)$ may be skipped.

It is also worth mentioning that if $R \in \text{Stop}(R)$, then the Lipschitz function mentioned in Section 3.4 may be chosen identically zero and its graph is just L_R .

3.8.2 Properties of cubes in trees

Below, we will collect main properties of cubes from Tree(R) that readily follow from the stopping conditions. Before it we prove an additional result.

Lemma 3.8. For any $Q \in \text{Tree}(R)$, we have

$$\Theta_{\mu}(2B_Q) \lesssim A \Theta_{\mu}(2B_R).$$

The implicit constant depends only on C_0 and A_0 .

Proof. Let $Q \in \text{Tree}(R)$. If $Q \in \mathcal{D}^{db}$, then there is nothing to prove. If not, then denote by $\tilde{Q} \in \mathcal{D}^{db}$ the first doubling ancestor of Q. Such a cube \tilde{Q} exists and $\tilde{Q} \subset R$ because $R \in \mathcal{D}^{db}$ by construction. Since the intermediate cubes $P, Q \subsetneq P \subsetneq \tilde{Q}$, do not belong to \mathcal{D}^{db} , by Lemma 3.4 we have

$$\Theta_{\mu}(2B_Q) \lesssim \Theta_{\mu}(100B(Q)) \lesssim C_0 A_0 \Theta_{\mu}(100B(Q)).$$

Using that $\tilde{Q} \in \mathcal{D}^{db}$, namely, the inequality (3.8), we get

$$\Theta_{\mu}(2B_Q) \lesssim C_0^2 A_0 \,\Theta_{\mu}(2B_{\tilde{Q}}) \lesssim C_0^2 A_0 \,A \,\Theta_{\mu}(2B_R),$$

and we are done.

Lemma 3.9. The following properties hold:

$$\tau \,\Theta_{\mu}(2B_R) \leqslant \Theta_{\mu}(2B_Q) \lesssim A \,\Theta_{\mu}(2B_R), \quad \forall Q \in \mathsf{Tree}(R) \setminus (\mathsf{LD}(R) \cup \mathsf{HD}(R)). \tag{3.15}$$

$$Q \in \mathcal{D}^{db} \cap (\mathsf{Tree}(R) \setminus (\mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{UB}(R))) \implies Q \text{ is } \gamma \text{-balanced.}$$
(3.16)

$$\sum_{Q \subset \tilde{Q} \subset R} \operatorname{perm}(\tilde{Q})^2 < \alpha^2 \quad \forall Q \in \operatorname{Tree}(R) \setminus (\operatorname{HD}(R) \cup \operatorname{LD}(R) \cup \operatorname{UB}(R) \cup \operatorname{BP}(R)).$$
(3.17)

$$\beta_{\mu,2}(2B_Q)^2 \leqslant \varepsilon_0^2 \Theta_\mu(2B_Q) \quad if \ \alpha = \alpha(\gamma, \tau, \varepsilon_0) \ is \ small \ enough \ and Q \in \mathcal{D}^{db} \cap (\mathsf{Tree}(R) \setminus (\mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{UB}(R) \cup \mathsf{BP}(R))).$$
(3.18)

$$\measuredangle(L_Q, L_R) \leqslant \theta(R) \qquad \forall Q \in \mathsf{DbTree}(R). \tag{3.19}$$

$$\mu(Q \setminus 2B_Q^{\mathsf{Cl}}) \leqslant \sqrt{\alpha} \,\mu(Q) \qquad \forall Q \in \mathsf{Tree}(R) \setminus \mathsf{Stop}(R). \tag{3.20}$$

Proof. The statement (3.15) follows from Lemma 3.8 and the stopping condition (S1). The statements (3.16), (3.17), (3.19) and (3.20) immediately follow from the construction of $\mathsf{Stop}(R)$ and $\mathsf{Tree}(R)$, while (3.18) is implied by Lemma 3.7 and the stopping conditions (S1) and (S2).

The following property of γ -balanced cubes will be used many times below.

Lemma 3.10. Let $\varepsilon_0 = \varepsilon_0(\gamma)$ be chosen small enough. Then for any $Q \in \mathcal{D}^{db} \cap (\mathsf{Tree}(R) \setminus (\mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{UB}(R) \cup \mathsf{BP}(R)))$ there exist two sets $\mathcal{Z}_k \subset Q$, k = 1, 2, such that

$$\mu(Q) \lesssim_{\gamma} \mu(\mathcal{Z}_k) \leqslant \mu(Q) \quad and \quad \operatorname{dist} (\mathcal{Z}_1, \mathcal{Z}_2) \geqslant \gamma r(B_Q),$$

and moreover for any $z_1 \in \mathcal{Z}_1$ and $z_2 \in \mathcal{Z}_2$ we have

dist
$$_H(L_{z_1,z_2} \cap 2B_Q, L_Q \cap 2B_Q) \leq \sqrt{\varepsilon_0} r(B_Q).$$

Proof. Since $Q \in \mathcal{D}^{db} \cap (\mathsf{Tree}(R) \setminus (\mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{UB}(R))), Q \text{ is } \gamma\text{-balanced by (3.16)}.$ Furthermore, by Lemma 3.5 there exist balls $B_k = B(\xi_k, \rho' r(Q)), k = 1, 2$, where $\xi_k \in B(Q)$, such that

$$\mu(B_k \cap B(Q)) \ge \rho'' \mu(Q)$$
 and $\operatorname{dist}(y_1, y_2) \ge \gamma r(B_Q)$ for any $y_k \in B_k, \ k = 1, 2,$

where ρ' and ρ'' depend on γ . Due to the estimate $\beta_{\mu,2}(2B_Q)^2 \leq \varepsilon_0^2 \Theta_{\mu}(2B_Q)$ (see (3.18)), by Chebyshev's inequality there exist $\mathcal{Z}_k \subset B_k \cap Q$ such that

$$\mu(Q) \lesssim_{\gamma} \mu(B_k) \lesssim \mu(\mathcal{Z}_k) \leqslant \mu(Q) \text{ and } \sup_{z \in \mathcal{Z}_k} \operatorname{dist}(z, L_Q) \lesssim_{\gamma} \varepsilon_0 r(B_Q), \quad k = 1, 2.$$

Thus for any $z_1 \in \mathcal{Z}_1$ and $z_2 \in \mathcal{Z}_2$ we have

dist
$$(z_k, L_Q) \lesssim_{\gamma} \varepsilon_0 r(B_Q), \quad k = 1, 2, \qquad \text{dist} (z_1, z_2) \gtrsim_{\gamma} r(B_Q).$$

This implies that $\measuredangle(L_{z_1,z_2},L_Q) \lesssim_{\gamma} \varepsilon_0$ and therefore the following estimate for the Hausdorff distance holds:

dist $_{H}(L_{z_1,z_2} \cap 2B_Q, L_Q \cap 2B_Q) \lesssim_{\gamma} \varepsilon_0 r(B_Q).$

Choosing ε_0 small enough with respect to the implicit constant depending on γ , we obtain the required result.

Clearly, it may happen that not all cubes in Tree(R) are γ -balanced as there may be undoubling cubes. However, for any cube in Tree(R), there is always an ancestor in DbTree(R) close by. Namely, the following result holds.

Lemma 3.11 (Lemma 6.3 in [AT]). For any cube $Q \in \text{Tree}(R)$ there exists a cube $\tilde{Q} \supset Q$ such that $\tilde{Q} \in \text{DbTree}(R)$ and diam $(\tilde{Q}) \leq \lambda \operatorname{diam}(Q)$ with some $\lambda = \lambda(A, \tau)$.

Now we want to show that the measure of the set of points from R which are far from the best approximation lines for cubes in $\{R\} \cup (\text{Tree}(R) \setminus \text{Stop}(R))$ is small. Set

$$p_0^{[\delta,Q]}(x,\mu,\mu) := \iint_{\delta r(Q) \leqslant |x-y| \leqslant \delta^{-1} r(Q)} p_0(x,y,z) \ d\mu(y) d\mu(z)$$

and consider

$$\begin{split} R_{\mathsf{Far}} &:= \{ x \in R : \quad \frac{p_0^{[\delta,Q]}(x,\mu\lfloor 2B_R,\mu\lfloor 2B_R)}{\Theta_\mu(2B_R)^2} \geqslant \mathsf{c}_2(\gamma,\tau,\varepsilon_0) \\ & \text{for some } Q \in \{R\} \cup (\mathsf{Tree}(R) \setminus \mathsf{Stop}(R)) \text{ such that } x \in 2B_Q \} \end{split}$$

where $c_2(\gamma, \tau, \varepsilon_0) > 0$ will be defined precisely in the proof of Lemma 3.13.

Lemma 3.12. If $R \notin UB(R) \cup BP(R)$ and $\alpha = \alpha(\gamma, \tau, \varepsilon_0)$ is chosen small enough, then

$$\mu(R_{\mathsf{Far}}) \leqslant \alpha \mu(R).$$

Proof. By Chebyshev's inequality,

$$\begin{split} \mathbf{c}_{2}(\gamma,\tau,\varepsilon_{0})\,\mu(R_{\mathsf{Far}}) \\ &\leqslant \int_{R} \sum_{Q\in\{R\}\cup(\mathsf{Tree}(R)\backslash\mathsf{Stop}(R)):\,x\in 2B_{Q}} \frac{p_{0}^{[\delta,Q]}(x,\mu\lfloor 2B_{R},\mu\lfloor 2B_{R})}{\Theta_{\mu}(2B_{R})^{2}}\,d\mu(x) \\ &\leqslant \sum_{Q\in\{R\}\cup(\mathsf{Tree}(R)\backslash\mathsf{Stop}(R))} \frac{p_{0}^{[\delta,Q]}(\mu\lfloor 2B_{Q},\mu\lfloor 2B_{R},\mu\lfloor 2B_{R})}{\Theta_{\mu}(2B_{R})^{2}} \\ &= \sum_{Q\in\{R\}\cup(\mathsf{Tree}(R)\backslash\mathsf{Stop}(R))} \frac{p_{0}^{[\delta,Q]}(\mu\lfloor 2B_{Q},\mu\lfloor 2B_{R},\mu\lfloor 2B_{R})}{\Theta_{\mu}(2B_{R})^{2}\mu(Q)}\int \chi_{Q}(x)\,d\mu(x). \end{split}$$

Changing the order of summation yields

$$\begin{split} \mathbf{c}_{2}(\gamma,\tau,\varepsilon_{0}) & \mu(R_{\mathsf{Far}}) \\ &\leqslant \int_{R} \left(\frac{p_{0}^{[\delta,R]}(\mu\lfloor 2B_{R})}{\Theta_{\mu}(2B_{R})^{2}\mu(R)} + \sum_{Q\in\mathsf{Tree}(R)\backslash\mathsf{Stop}(R):\,x\in Q} \frac{p_{0}^{[\delta,Q]}(\mu\lfloor 2B_{Q},\mu\lfloor 2B_{R},\mu\lfloor 2B_{R})}{\Theta_{\mu}(2B_{R})^{2}\mu(Q)} \right) \, d\mu(x) \\ &\leqslant \int_{R} \left(\mathsf{perm}(R)^{2} + \sum_{Q\in\mathsf{Tree}(R)\backslash\mathsf{Stop}(R):\,x\in Q} \mathsf{perm}(Q)^{2} \right) \, d\mu(x) \\ &\leqslant 2\alpha^{2} \, \mu(R). \end{split}$$

Supposing that $2\alpha \leq c_2(\gamma, \tau, \varepsilon_0)$ gives the required result.

Recall the definition (3.14).

Lemma 3.13. Let $\delta = \delta(\varepsilon_0)$ be chosen small enough. If $x \in (R \cap 2B_{\tilde{Q}}) \setminus 2B_{\tilde{Q}}^{\mathsf{Cl}}$ for some $\tilde{Q} \in \mathsf{Tree}(R)$, i.e. in particular there exists $Q \in \mathcal{D}^{db}(R)$ such that $2B_Q \supset 2B_{\tilde{Q}}$ and Q is not contained in any cube from $\mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{UB}(R) \cup \mathsf{BP}(R) \cup \mathsf{BS}(R)$, then $x \in R_{\mathsf{Far}}$.

Proof. Clearly, $x \in 2B_Q$ and $Q \in \mathcal{D}^{db} \cap (\mathsf{Tree}(R) \setminus (\mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{UB}(R) \cup \mathsf{BP}(R)))$. Therefore, by Lemma 3.10 we can find $\mathcal{Z}_k \subset Q$, k = 1, 2, such that for any $z_1 \in \mathcal{Z}_1$ and $z_2 \in \mathcal{Z}_2$ we have

dist
$$_H(L_{z_1,z_2} \cap 2B_Q, L_Q \cap 2B_Q) \leq \sqrt{\varepsilon_0} r(B_Q)$$
.

Consider triangle (x, z_1, z_2) which is wholly contained in $2B_Q$. It is easily seen that

$$\operatorname{dist}(x, L_{z_1, z_2}) \ge \operatorname{dist}(x, L_Q) - \operatorname{dist}_H(L_{z_1, z_2} \cap 2B_Q, L_Q \cap 2B_Q) \ge 4\sqrt{\varepsilon_0} r(B_Q). \quad (3.21)$$

This implies that one of the angle of the triangle (x, z_1, z_2) is at least

$$\frac{4\sqrt{\varepsilon_0} r(B_Q)}{\operatorname{diam}(2B_Q)} = \sqrt{\varepsilon_0},$$

and thus $(x, z_1, z_2) \in V_{\mathsf{Far}}(\sqrt{\varepsilon_0})$ for any $z_1 \in \mathbb{Z}_1$ and $z_2 \in \mathbb{Z}_2$. Note also that (3.21) implies that $|x - z_1| \ge \delta r(Q)$ if $\delta = \delta(\varepsilon_0)$ is chosen small enough. Consequently, by the identity (0.15) and Lemma 3.6,

$$\begin{split} p_0^{[\delta,Q]}(x,\mu\lfloor 2B_R,\mu\lfloor 2B_R) &\ge \iint_{z_1\in\mathcal{Z}_1,\,z_2\in\mathcal{Z}_2} p_0(x,z_1,z_2) \,\,d\mu(z_1)d\mu(z_2) \\ &\ge \frac{1}{2}\mathsf{c}_1(\sqrt{\varepsilon_0}) \iint_{z_1\in\mathcal{Z}_1,\,z_2\in\mathcal{Z}_2} c(x,z_1,z_2)^2 \,\,d\mu(z_1)d\mu(z_2) \\ &= \frac{1}{2}\mathsf{c}_1(\sqrt{\varepsilon_0}) \iint_{z_1\in\mathcal{Z}_1,\,z_2\in\mathcal{Z}_2} \left(\frac{2\operatorname{dist}\left(x,L_{z_1,z_2}\right)}{|x-z_1||x-z_2|}\right)^2 \,\,d\mu(z_1)d\mu(z_2), \end{split}$$

where the constant c_1 is from Lemma 3.6. Furthermore, we apply (3.21) and the fact that $|x - z_k| \leq \text{diam}(2B_Q) = 4r(B_Q)$ for k = 1, 2 to obtain the following:

$$p_0^{[\delta,Q]}(x,\mu\lfloor 2B_R,\mu\lfloor 2B_R) \geqslant \frac{\varepsilon_0 \operatorname{c}_1(\sqrt{\varepsilon_0})}{8r(B_Q)^2} \,\mu(\mathcal{Z}_1)\mu(\mathcal{Z}_2).$$

Since $\mu(\mathcal{Z}_k) \gtrsim_{\gamma} \mu(Q)$ by Lemma 3.10, $\mu(Q) \gtrsim \mu(2B_Q)$ as $Q \in \mathcal{D}^{db}$ and $\Theta_{\mu}(2B_Q) \geq \tau \Theta_{\mu}(2B_R)$ by (3.15), we finally get

$$p_0^{[\delta,Q]}(x,\mu\lfloor 2B_R,\mu\lfloor 2B_R)\gtrsim_{\gamma}\varepsilon_0\,\mathsf{c}_1(\sqrt{\varepsilon_0})\,\tau^2\Theta_\mu(2B_R)^2=\mathsf{c}_2(\gamma,\tau,\varepsilon_0)\,\Theta_\mu(2B_R)^2.$$

Consequently, $x \in R_{\mathsf{Far}}$ by definition.

Remark 3.6. Suppose that $R \in \mathsf{F}(R)$ and thus $\mu(R \setminus 2B_R^{\mathsf{Cl}}) > \sqrt{\alpha}\,\mu(R)$ by definition. Then it is clear that $R \notin \mathsf{UB}(R) \cup \mathsf{BP}(R)$ (and furthermore $R \notin \mathsf{HD}(R) \cup \mathsf{LD}(R) \cup \mathsf{BS}(R)$, see Remark 3.5) and so $\mu(R_{\mathsf{Far}}) \leq \alpha \,\mu(R)$ by Lemma 3.12. Furthermore, $R \setminus 2B_R^{\mathsf{Cl}} \subset R_{\mathsf{Far}}$ by Lemma 3.13 (where one takes R for both Q and \tilde{Q}) and thus $\mu(R \setminus 2B_R^{\mathsf{Cl}}) \leq \alpha \mu(R)$ which contradicts the fact that $R \in \mathsf{F}(R)$ as $\alpha \ll 1$.

3.9 Measure of stopping cubes from BP(R) and F(R)

Lemma 3.14. It holds that

$$\sum_{Q \in \mathsf{BP}(R)} \mu(Q) \leqslant \frac{1}{\alpha^2 \, \Theta_{\mu}(2B_R)^2} \sum_{\tilde{Q} \in \mathsf{Tree}(R)} p_0^{[\delta, \tilde{Q}]}(\mu \lfloor 2B_{\tilde{Q}}, \mu \lfloor 2B_R, \mu \lfloor 2B_R).$$

What is more, if $\alpha = \alpha(\tau)$ is small enough, then

$$\sum_{Q\in \mathsf{F}(R)} \mu(Q) \leqslant \sqrt{\alpha}\, \mu(R) \leqslant \tfrac{1}{3} \sqrt{\tau}\, \mu(R).$$

Proof. All the cubes in $\mathsf{Stop}(R)$ are disjoint and so are the cubes in $\mathsf{BP}(R)$ and $\mathsf{F}(R)$. From **(S3)** we get

$$\begin{split} \sum_{Q \in \mathsf{BP}(R)} \mu(Q) &\leqslant \frac{1}{\alpha^2} \sum_{Q \in \mathsf{BP}(R)} \sum_{Q \subset \tilde{Q} \subset R} \frac{p_0^{[\delta,Q]}(\mu \lfloor 2B_{\tilde{Q}}, \mu \lfloor 2B_R, \mu \lfloor 2B_R)}{\Theta_{\mu}(2B_R)^2 \mu(\tilde{Q})} \, \mu(Q) \\ &= \frac{1}{\alpha^2 \Theta_{\mu}(2B_R)^2} \sum_{\tilde{Q} \in \mathsf{Tree}(R)} p_0^{[\delta,\tilde{Q}]}(\mu \lfloor 2B_{\tilde{Q}}, \mu \lfloor 2B_R, \mu \lfloor 2B_R) \sum_{Q \in \mathsf{BP}(R): \ Q \subset \tilde{Q}} \frac{\mu(Q)}{\mu(\tilde{Q})} \\ &\leqslant \frac{1}{\alpha^2 \Theta_{\mu}(2B_R)^2} \sum_{\tilde{Q} \in \mathsf{Tree}(R)} p_0^{[\delta,\tilde{Q}]}(\mu \lfloor 2B_{\tilde{Q}}, \mu \lfloor 2B_R, \mu \lfloor 2B_R). \end{split}$$

By (S4) and Lemmas 3.12 and 3.13, we obtain

$$\sum_{Q \in \mathsf{F}(R)} \mu(Q) \leqslant \frac{1}{\sqrt{\alpha}} \sum_{Q \in \mathsf{F}(R)} \mu(Q \setminus 2B_Q^{\mathsf{CI}}) \leqslant \frac{1}{\sqrt{\alpha}} \, \mu(R_{\mathsf{Far}}) \leqslant \sqrt{\alpha} \, \mu(R),$$

which finishes the proof.

3.10 Construction of a Lipschitz function

We aim to construct a Lipschitz function $F : L_R \to L_R^{\perp}$ whose graph Γ_R is close to Rup to the scale of cubes from Stop(R). We will mostly use the properties mentioned in Lemma 3.9. This task is quite technical and so we start with a bunch of auxiliary results. Note that, although we follow some of the methods from [Leg] and [Tol5, Chapter 7] quite closely, we need to adapt the whole construction to the David-Mattila lattice used in the current chapter (instead of the balls with controlled density used in [Leg] and [Tol5]).

Let us mention again that we may suppose that $R \notin \text{Stop}(R)$ as otherwise we choose $F \equiv 0$ and the graph Γ_R of F is just L_R .

3.10.1 Auxiliary results

As before, we denote by L_Q a best approximating line for the ball $2B_Q$ in the sense of the beta numbers (3.2). We need now to estimate the angles between the best approximating lines corresponding to cubes that are near each other. This task is carried out in the next two lemmas. The first one is a well known result from [DS1, Section 5]. We formulate it for lines in the complex plane.

Lemma 3.15 ([DS1]). Let $L_1, L_2 \subset \mathbb{C}$ be lines and $z_1, z_2 \in Z \subset \mathbb{C}$ be points so that

- (a) $d_1 = \operatorname{dist}(z_1, z_2) / \operatorname{diam}(Z) \in (0, 1),$
- (b) dist $(z_i, L_j) < d_2 \operatorname{diam}(Z)$ for i = 1, 2 and j = 1, 2, where $d_2 < d_1/4$.

Then for any $z \in L_2$,

dist
$$(z, L_1) \leq d_2 \left(\frac{4}{d_1} \operatorname{dist}(z, Z) + \operatorname{diam}(Z) \right).$$
 (3.22)

We will use the preceding lemma to prove the following result.

Lemma 3.16. Let $\varepsilon_0 = \varepsilon_0(\gamma)$ be chosen small enough. If $Q_1, Q_2 \in \mathsf{DbTree}(R)$ are such that $r(Q_1) \approx r(Q_2)$ and dist $(Q_1, Q_2) \leq r(Q_j)$ for j = 1, 2, then

$$\operatorname{dist}(w, L_{Q_2}) \lesssim \sqrt{\varepsilon_0} \left(\operatorname{dist}(w, Q_1) + r(Q_1) \right), \qquad w \in L_{Q_1}, \tag{3.23}$$

$$\operatorname{dist}(w, L_{Q_1}) \lesssim \sqrt{\varepsilon_0} \left(\operatorname{dist}(w, Q_2) + r(Q_2) \right), \qquad w \in L_{Q_2}, \tag{3.24}$$

$$\measuredangle(L_{Q_1}, L_{Q_2}) \lesssim \sqrt{\varepsilon_0}. \tag{3.25}$$

Proof. Let $Q \in \mathsf{DbTree}(R)$ be the smallest cube such that $2B_Q \supset B_{Q_1} \cap B_{Q_2}$. Clearly, $r(Q) \gtrsim r(Q_j), j = 1, 2$. Moreover, we can also guarantee that

$$r(Q) \lesssim \operatorname{dist}(Q_1, Q_2) + \sum_{j=1}^2 r(Q_j) \lesssim r(Q_j)$$

Now we use arguments similar to those in Lemma 3.10. Since $Q_j \in \mathsf{DbTree}(R)$ for j = 1, 2, by (3.16) and Lemma 3.5 there are balls $B_{k,j} = B(\xi_{k,j}, \rho' r(Q_j)), k = 1, 2$, where $\xi_{k,j} \in B(Q_j)$, such that $\mu(B_{k,j} \cap B(Q_j)) \ge \rho'' \mu(Q_j)$ and $\operatorname{dist}(y_{1,j}, y_{2,j}) \ge \gamma r(B_{Q_j}) \ge \gamma r(Q_j)$ for all $y_{k,j} \in B_{k,j} \cap Q_j$, where ρ' and ρ'' depend on γ . Consequently, by (3.18) and the fact that $r(B_{k,j}) \approx_{\gamma} r(Q_j)$ we get

$$\frac{1}{r(B_{k,j})} \int_{B_{k,j}} \left(\frac{\operatorname{dist}(w, L_{Q_j})}{r(B_{k,j})} \right)^2 d\mu(w) \lesssim_{\gamma} \beta_{\mu,2} (2B_{Q_j})^2 \lesssim_{\gamma} \varepsilon_0^2 \Theta_{\mu}(2B_{Q_j}).$$

Since $r(Q) \approx r(Q_j)$, we analogously obtain

$$\frac{1}{r(B_{k,j})} \int_{B_{k,j}} \left(\frac{\operatorname{dist}\left(w, L_Q\right)}{r(B_{k,j})} \right)^2 d\mu(w) \lesssim_{\gamma} \beta_{\mu,2} (2B_Q)^2 \lesssim_{\gamma} \varepsilon_0^2 \Theta_{\mu}(2B_Q)$$

Therefore, using Chebyshev's inequality and again the relation $r(Q) \approx r(Q_j)$, we can find $z_{k,j} \in B_{k,j} \cap Q_j$ such that

$$\max\{\operatorname{dist}(z_{k,j}, L_{Q_j}), \operatorname{dist}(z_{k,j}, L_Q)\} \lesssim_{\gamma} \varepsilon_0 r(Q).$$

Since dist $(z_{1,j}, z_{2,j}) \ge \gamma r(Q_j) \ge \gamma r(Q)$, it follows by Lemma 3.15 that

dist
$$(w, L_Q) \lesssim_{\gamma} \varepsilon_0($$
dist $(w, Q_j) + r(Q_j))$ for all $w \in L_{Q_j}, \qquad j = 1, 2,$

and

dist
$$(w, L_{Q_j}) \lesssim_{\gamma} \varepsilon_0(\text{dist}(w, Q_j) + r(Q_j))$$
 for all $w \in L_Q$, $j = 1, 2$

From this, by the triangle inequality, choosing ε_0 small enough with respect to the implicit constant depending on γ , we obtain (3.23) and (3.24).

The inequality (3.25) follows from (3.23) and (3.24) by elementary geometry.

Lemma 3.17. Let $\alpha = \alpha(\gamma)$ and $\varepsilon_0 = \varepsilon_0(\gamma)$ be chosen small enough. If $Q_1, Q_2 \in DbTree(R)$ are such that $2B_{Q_1} \subset 2B_{Q_2}$ and $x \in L_{Q_1} \cap 2B_{Q_1}$, then

$$\operatorname{dist}\left(x, L_{Q_2}\right) \lesssim \varepsilon_0^{1/3} r(Q_2).$$

Proof. By Lemma 3.5 there exists a family of balls $B_k = B(\xi_k, \rho' r(Q_1))$, where $\xi_k \in B(Q_1)$, such that $\mu(B_k \cap B(Q_1)) \ge \rho'' \mu(Q_1)$ and dist $(y_1, y_2) \ge \gamma r(B_{Q_1}) \ge \gamma r(Q_1)$ for any $y_k \in B_k \cap Q_1$, k = 1, 2. Recall that ρ' and ρ'' depend on γ . Furthermore, we can choose $\alpha = \alpha(\gamma)$ in (3.20) small enough to guarantee that $B_k \cap B(Q_1) \cap 2B_{Q_1}^{\mathsf{Cl}} \ne \emptyset$. This and the definition of $2B_{Q_1}^{\mathsf{Cl}}$ imply that there exist $z_k \in B_k \cap B(Q_1) \cap 2B_{Q_1}^{\mathsf{Cl}}$, k = 1, 2, such that

dist
$$(z_k, L_{Q_j}) \lesssim \sqrt{\varepsilon_0} r(Q_j), \quad k = 1, 2, \quad j = 1, 2.$$

Let z'_k be the orthogonal projection of z_k onto L_{Q_1} . We easily get from the previous inequality that

dist
$$(z'_k, L_{Q_2}) \lesssim \sqrt{\varepsilon_0} r(Q_2), \quad k = 1, 2.$$
 (3.26)

Moreover, dist $(z_1, z_2) \gtrsim_{\gamma} r(Q_1)$ implies that dist $(z'_1, z'_2) \gtrsim_{\gamma} r(Q_1)$ and $z'_k \in 2B_{Q_1}$, if $\varepsilon_0 = \varepsilon_0(\gamma)$ is small enough. Having this and (3.26) in mind and taking into account that $x \in L_{Q_1} \cap 2B_{Q_1}$, by elementary geometry we get the required estimate for dist (x, L_{Q_2}) , assuming again that $\varepsilon_0 = \varepsilon_0(\gamma)$ is small enough.

3.10.2 Lipschitz function F for the good part of R

For each given $R \in \mathcal{D}^{db}$, we first construct the required function F on the projection of the "good part" of R onto L_R and then extend it onto the whole L_R . In what follows, we will work a lot with the function

$$d(z) := \inf_{Q \in \mathsf{DbTree}(R)} \{ \operatorname{dist}(z, Q) + \operatorname{diam}(Q) \}, \qquad z \in \mathbb{C}.$$
(3.27)

Let us mention that $\theta(R)$ is supposed to be comparable with the parameter θ_0 , i.e. $\theta(R) \approx \theta_0$, where the implicit constants will be defined in Section 3.12.

Lemma 3.18. Let $\varepsilon_0 = \varepsilon_0(\tau, A, \theta_0)$ and θ_0 be small enough. For any $z_1, z_2 \in cB_R$ we have

$$|\Pi^{\perp}(z_1) - \Pi^{\perp}(z_2)| \lesssim \theta(R) |\Pi(z_1) - \Pi(z_2)| + c(\tau, A) \left(d(z_1) + d(z_2) \right),$$

where $\Pi(z)$ and $\Pi^{\perp}(z)$ are the projections of z onto L_R and L_R^{\perp} , correspondingly, and $c(\tau, A) > 0$.

Proof. Everywhere in the proof k = 1, 2. For a fixed h > 0 and any $z_k \in cB_R$ one can always find $Q_k \in \mathsf{DbTree}(R)$ such that

$$\operatorname{dist}(z_k, Q_k) + \operatorname{diam}(Q_k) \leqslant d(z_k) + h, \qquad k = 1, 2.$$

Choose $z'_k \in Q_k$. Clearly, $|z_k - z'_k| \leq d(z_k) + h$. Let $\tilde{Q}_k \in \mathsf{DbTree}(R)$ be the smallest cube such that $2B_{\tilde{Q}_k} \supset B_{Q_k}$ and

$$r(\tilde{Q}_k) \approx_{\tau,A} \varepsilon_0 |z_1 - z_2| + \sum_k \operatorname{diam}(Q_k).$$

Now let $\tilde{Q} \in \mathsf{DbTree}(R)$ be the smallest cube such that $2B_{\tilde{Q}} \supset B_{\tilde{Q}_1} \cup B_{\tilde{Q}_2}$ and

$$r(\tilde{Q}) \approx_{\tau,A} |z_1 - z_2| + \sum_k \operatorname{diam}(Q_k).$$

Note that $|z_1 - z_2| \leq r(R)$ as $z_k \in cB_R$ and thus the cubes \tilde{Q}_k and \tilde{Q} are well defined. Furthermore, we easily get that $\varepsilon_0 r(\tilde{Q}) \lesssim_{\tau,A} r(\tilde{Q}_k)$. Consequently, the way how \tilde{Q}_k and \tilde{Q} are chosen and the inequalities (3.15) and (3.18) in Lemma 3.9 imply that

$$\frac{1}{\mu(B_{\tilde{Q}_k})} \int_{B_{\tilde{Q}_k}} \left(\frac{\operatorname{dist}\left(w, L_{\tilde{Q}}\right)}{r(\tilde{Q})} \right)^2 d\mu(w) \lesssim_{\tau,A} \frac{r(\tilde{Q}) \beta_{\mu,2} (2B_{\tilde{Q}})^2}{\mu(2B_{\tilde{Q}_k})} \lesssim_{\tau,A} \varepsilon_0^2 \frac{r(\tilde{Q}) \Theta_{\mu}(2B_{\tilde{Q}})}{\mu(2B_{\tilde{Q}_k})} \\ \lesssim_{\tau,A} \varepsilon_0 \frac{\Theta_{\mu}(2B_{\tilde{Q}_k})}{\Theta_{\mu}(2B_{\tilde{Q}_k})} \lesssim_{\tau,A} \varepsilon_0 \lesssim \varepsilon_0^{3/4},$$

if $\varepsilon_0 = \varepsilon_0(\tau, A)$ is chosen properly. Recall again that $r(B_Q) = 28r(Q)$ by definition.

From the inequality just obtained we deduce by Chebyshev's inequality that there exist $z_k'' \in R \cap B_{\tilde{Q}_k}, \ k = 1, 2$, such that

$$|z_k'' - \pi(z_k'')| \lesssim \varepsilon_0^{3/8} r(\tilde{Q}) \lesssim \sqrt[4]{\varepsilon_0} \left(|z_1 - z_2| + \sum_k \operatorname{diam} (Q_k) \right),$$

where $\pi(z_k'')$ stands for the orthogonal projection of z_k'' onto $L_{\tilde{Q}}$ and $\varepsilon_0 = \varepsilon_0(\tau, A)$ is small enough. Note also that

$$|z'_k - z''_k| \lesssim r(\tilde{Q}_k) \lesssim \sqrt[4]{\varepsilon_0} |z_1 - z_2| + c(\tau, A) \sum_k \operatorname{diam}(Q_k),$$

if $\varepsilon_0 = \varepsilon_0(\tau, A)$ is small enough. Summarizing, we obtain the inequality

$$|z'_k - \pi(z''_k)| \le |z'_k - z''_k| + |z''_k - \pi(z''_k)| \le \sqrt[4]{\varepsilon_0} |z_1 - z_2| + c(\tau, A) \sum_k \operatorname{diam}(Q_k).$$

Furthermore, the triangle inequality yields

$$\begin{aligned} |\Pi^{\perp}(z_1') - \Pi^{\perp}(z_2')| &\leq |\Pi^{\perp}(\pi(z_1'')) - \Pi^{\perp}(\pi(z_2''))| + \sum_k |\Pi^{\perp}(z_k') - \Pi^{\perp}(\pi(z_k''))| \\ &\leq |\Pi^{\perp}(\pi(z_1'')) - \Pi^{\perp}(\pi(z_2''))| + \sum_k |z_k' - \pi(z_k'')|, \end{aligned}$$

and therefore we immediately obtain

$$|\Pi^{\perp}(z_1') - \Pi^{\perp}(z_2')| \lesssim |\Pi^{\perp}(\pi(z_1'')) - \Pi^{\perp}(\pi(z_2''))| + \sqrt[4]{\varepsilon_0}|z_1 - z_2| + c(\tau, A) \sum_k \operatorname{diam}(Q_k).$$
From (3.19) in Lemma 3.9 applied to \tilde{Q} and the triangle inequality we deduce that

$$\begin{aligned} |\Pi^{\perp}(\pi(z_{1}'')) - \Pi^{\perp}(\pi(z_{2}''))| \\ &\lesssim \theta(R) |\Pi(\pi(z_{1}')) - \Pi(\pi(z_{2}''))| \\ &\lesssim \theta(R) \left(|\Pi(z_{1}) - \Pi(z_{2})| + \sum_{k} |\Pi(z_{k}) - \Pi(\pi(z_{k}''))| \right) \\ &\lesssim \theta(R) \left(|\Pi(z_{1}) - \Pi(z_{2})| + \sum_{k} |z_{k} - \pi(z_{k}'')| \right) \\ &\lesssim \theta(R) \left(|\Pi(z_{1}) - \Pi(z_{2})| + \sum_{k} \left(|z_{k} - z_{k}'| + |z_{k}' - \pi(z_{k}'')| \right) \right). \end{aligned}$$

Recall the estimates for $|z_k - z'_k|$ and $|z'_k - \pi(z''_k)|$ and take into account that diam $(Q_k) \leq d(z_k) + h$ and that ε_0 and θ_0 (and thus $\theta(R)$) are small. Consequently,

$$|\Pi^{\perp}(z_1') - \Pi^{\perp}(z_2')| \lesssim \theta(R) |\Pi(z_1) - \Pi(z_2)| + \sqrt[4]{\varepsilon_0} |z_1 - z_2| + c(\tau, A) \sum_k (d(z_k) + h).$$

Additionally, the triangle inequality and the estimate for $|z_k - z'_k|$ lead to

$$|\Pi^{\perp}(z_1) - \Pi^{\perp}(z_2)| \leq |\Pi^{\perp}(z_1') - \Pi^{\perp}(z_2')| + \sum_k (d(z_k) + h),$$

and thus

$$|\Pi^{\perp}(z_1) - \Pi^{\perp}(z_2)| \lesssim \theta(R) |\Pi(z_1) - \Pi(z_2)| + \sqrt[4]{\varepsilon_0} |z_1 - z_2| + c(\tau, A) \sum_k (d(z_k) + h).$$

Take into account that $|z_1 - z_2| \leq |\Pi(z_1) - \Pi(z_2)| + |\Pi^{\perp}(z_1) - \Pi^{\perp}(z_2)|$ and choose ε_0 small enough with respect to θ_0 (and thus to $\theta(R)$) and to the implicit absolute constant in the latter inequality. Finally,

$$|\Pi^{\perp}(z_1) - \Pi^{\perp}(z_2)| \lesssim \theta(R) |\Pi(z_1) - \Pi(z_2)| + c(\tau, A) \sum_k (d(z_k) + h).$$

Letting $h \to 0$ finishes the proof.

We will also use the following notation:

$$G_R = \{ x \in \mathbb{C} : d(x) = 0 \}.$$
(3.28)

Lemma 3.18 implies that the map $\Pi : G_R \to L_R$ is injective and we can define the function F on $\Pi(G_R)$ by setting

$$F(\Pi(x)) = \Pi^{\perp}(x), \qquad x \in G_R.$$
(3.29)

Moreover, this F is Lipschitz with constant $\leq \theta(R)$.

We are now aimed to extend F onto the whole line L_R using a variant of the Whitney extension theorem. This approach is quite standard and is used, for example, in [DS1, Section 8], [Leg, Section 3.2] and [Tol5, Section 7.5]. Therefore we will skip some details and mostly give the results related to the adaptation of the scheme to the David-Mattila lattice that we use. These results will then imply the extension of F onto the whole L_R by repeating the "partition of unity" arguments presented in [Tol5, Section 7.5].

Let us define the function

$$D(z) := \inf_{x \in \Pi^{-1}(z)} d(x) = \inf_{Q \in \mathsf{DbTree}(R)} \{ \operatorname{dist}(z, \Pi(Q)) + \operatorname{diam}(Q) \}, \qquad z \in L_R.$$
(3.30)

For each $z \in L_R$ such that D(z) > 0, i.e. $z \in L_R \setminus \Pi(G_R)$, we call J_z the largest dyadic interval from L_R containing z such that

$$\ell(J_z) \leqslant \frac{1}{20} \inf_{u \in J_z} D(u).$$

Let J_i , $i \in I$, be a relabelling of the set of all these intervals J_z , $z \in L_R \setminus \Pi(G_R)$, without repetition. Some properties of $\{J_i\}$ are summarized in the following lemma.

Lemma 3.19 (Analogue of Lemma 7.20 in [Tol5]). The intervals in $\{J_i\}_{i \in I}$ have disjoint interiors in L_R and satisfy the properties:

- (a) If $z \in 15J_i$, then $5\ell(J_i) \leq D(z) \leq 50\ell(J_i)$.
- (b) There exists an absolute constant c > 1 such that if $15J_i \cap 15J_{i'} \neq \emptyset$, then

$$c^{-1}\ell(J_i) \leqslant \ell(J_{i'}) \leqslant c\,\ell(J_i).$$

- (c) For each $i \in I$, there are at most N intervals $J_{i'}$ such that $15J_i \cap 15J_{i'} \neq \emptyset$, where N is some absolute constant.
- (d) $L_R \setminus \Pi(G_R) = \bigcup_{i \in I} J_i = \bigcup_{i \in I} 15J_i.$

Now we construct the function F on

$$U_0 = L_R \cap B_0, \qquad B_0 = B(\Pi(x_0), 10 \operatorname{diam}(R)),$$

where $x_0 \in R$ is such that

dist
$$(x_0, \Pi(x_0)) = \operatorname{dist}(x_0, L_R) \leqslant \operatorname{diam}(R)$$

This x_0 exists due to the inequality (3.18) in Lemma 3.9. Note that by construction

$$R \subset B(\Pi(x_0), 2\operatorname{diam}(R)) \quad \text{and} \quad \Pi(R) \subset L_R \cap B(\Pi(x_0), 2\operatorname{diam}(R)).$$
(3.31)

We also define the following set of indexes:

$$I_0 = \{ i \in I : J_i \cap U_0 \neq \emptyset \}.$$

Lemma 3.20. The following holds.

- (a) If $i \in I_0$, then $\ell(J_i) \leq \operatorname{diam}(R)$ and $3J_i \subset L_R \cap B(\Pi(x_0), 12\operatorname{diam}(R))$.
- (b) If $J_i \cap B(\Pi(x_0), 3 \operatorname{diam}(R)) = \emptyset$ (in particular if $i \notin I_0$), then

$$\ell(J_i) \approx \operatorname{dist} (\Pi(x_0), J_i) \approx |\Pi(x_0) - z| \quad for \ all \quad z \in J_i.$$

Proof. For (a), take J_i with $i \in I_0$ so that $J_i \cap U_0 \neq \emptyset$. Then we have

$$3J_i \subset L_R \cap B(\Pi(x_0), 10 \operatorname{diam}(R) + 2\ell(J_i)).$$

It is necessary to estimate $\ell(J_i)$. Recall that

$$\ell(J_i) \leqslant \frac{1}{20} \inf_{u \in J_i} D(u).$$

Definitely, $\inf_{u \in J_i} D(u) \leq \max_{u \in U_0} D(u)$ in our case so we will estimate this maximum instead. To do so, we first notice that the definition (3.27) of d and the inequality (3.31) give

$$d(x) \leq \operatorname{dist}(x, R) + \operatorname{diam}(R) \leq 13 \operatorname{diam}(R), \quad x \in B_0.$$

This yields

$$\max_{u \in U_0} D(u) \le \max_{x \in B_0} d(x) \le 13 \operatorname{diam}(R),$$

if we take into account the connection between d and D in (3.30). Thus

$$\ell(J_i) \leq \frac{13}{20} \operatorname{diam}(R)$$

and therefore

$$3J_i \subset L_R \cap B(\Pi(x_0), (10 + \frac{13}{10}) \operatorname{diam}(R)).$$

Now let us prove (b). Let $z \in J_i$. Clearly, diam $(R) \leq \frac{1}{3} |\Pi(x_0) - z|$. Furthermore, we infer from this and the definition (3.30) that

$$D(z) \leq (|\Pi(x_0) - z| + 2 \operatorname{diam}(R)) + \operatorname{diam}(R) \leq 2|\Pi(x_0) - z|.$$

From another side, by (3.30) and (3.31),

$$D(z) \ge \operatorname{dist}(z, \Pi(R)) \ge |\Pi(x_0) - z| - 2\operatorname{diam}(R) \ge \frac{1}{3}|\Pi(x_0) - z|.$$

Thus

$$\frac{1}{3}|\Pi(x_0) - z| \le D(z) \le 2|\Pi(x_0) - z|, \qquad z \in J_i.$$

Together with Lemma 3.19(a) this gives

$$\frac{5}{2}\ell(J_i) \leqslant |\Pi(x_0) - z| \leqslant 150\ell(J_i).$$

Moreover, since

$$|\Pi(x_0) - z| - \ell(J_i) \leqslant \operatorname{dist} (\Pi(x_0), J_i) \leqslant |\Pi(x_0) - z|, \qquad z \in J_i,$$

we get

$$\frac{3}{2}\ell(J_i) \leq \operatorname{dist}\left(\Pi(x_0), J_i\right) \leq 150\ell(J_i)$$

which finishes the proof.

Lemma 3.21. Given $i \in I_0$, there exists a cube $Q_i \in \mathsf{DbTree}(R)$ such that

- (a) $\ell(J_i) \lesssim \operatorname{diam}(Q_i) \lesssim_{\tau,A} \ell(J_i);$
- (b) dist $(J_i, \Pi(Q_i)) \lesssim \ell(J_i).$

Proof. From the definition (3.30) of D it follows that there exists a cube $Q \in \mathsf{DbTree}(R)$ such that

dist
$$(z, \Pi(Q))$$
 + diam $(Q) \leq 2D(z) \approx \ell(J_i), \qquad z \in J_i,$

where the comparability is due to Lemma 3.19(*a*). This immediately gives (*b*) and the right hand side inequality in (*a*) for $Q_i = Q$. If the left hand side inequality in (*a*) does not hold, we can replace Q by its smallest doubling ancestor Q' satisfying diam $(Q') \gtrsim \ell(J_i)$ so that all other inequalities are valid (recall Lemma 3.11). We rename Q' by Q_i then.

For $i \in I_0$, let F_i be the affine function $L_R \to L_R^{\perp}$ whose graph is the line L_{Q_i} . Moreover, F_i are Lipschitz functions with constant $\leq \theta(R)$ as $\measuredangle(L_{Q_i}, L_R) \leq \theta(R)$ by (3.19) in Lemma 3.9 taking into account that all $Q_i \in \mathsf{DbTree}(R)$. On the other hand, for $i \notin I_0$, we set $F_i \equiv 0$, i.e. the graph of F_i is just L_R in this case.

Lemma 3.22. If $10J_i \cap 10J_{i'} \neq \emptyset$ for some $i, i' \in I$, then

- (a) dist $(Q_i, Q_{i'}) \lesssim_{\tau, A} \ell(J_i)$ if moreover $i, i' \in I_0$;
- (b) $|F_i(z) F_{i'}(z)| \lesssim \varepsilon_0^{1/3} \ell(J_i) \text{ for } z \in 100J_i;$
- (c) $|F'_i F'_{i'}| \lesssim \varepsilon_0^{1/3}$.

Proof. For $i, i' \in I_0$, Lemmas 3.19(b) and 3.21(b) ensure that $\ell(Q_i) \approx \ell(Q_{i'})$ and

$$\operatorname{dist}\left(\Pi(Q_i), \Pi(Q_{i'})\right) \leqslant \operatorname{dist}\left(\Pi(Q_i), J_i\right) + \ell(J_i) + \operatorname{dist}\left(J_i, J_{i'}\right) + \ell(J_{i'}) + \operatorname{dist}\left(J_{i'}, \Pi(Q_{i'})\right) \lesssim \ell(J_i) + \ell(J_i) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_{i'}, \Pi(Q_{i'})\right) \leq \ell(J_i) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_{i'}, \Pi(Q_{i'})\right) \leq \ell(J_i) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_i, \Pi(Q_{i'})\right) \leq \ell(J_i) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_i, \Pi(Q_{i'})\right) \leq \ell(J_i) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_i, \Pi(Q_{i'})\right) \leq \ell(J_i) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_i, \Pi(Q_{i'})\right) \leq \ell(J_i) + \operatorname{dist}\left(J_i, \Pi(Q_{i'})\right) \leq \ell(J_i) + \operatorname{dist}\left(J_i, J_{i'}\right) + \operatorname{dist}\left(J_i, \Pi(Q_{i'})\right) \leq \ell(J_i) + \operatorname{dist}\left($$

Keeping this in mind, we continue. For any $z_1 \in Q_i$ and $z_2 \in Q_{i'}$ by the triangle inequality and Lemma 3.18 we have

dist
$$(Q_i, Q_{i'}) \leq dist(z_1, z_2) \leq |\Pi^{\perp}(z_1) - \Pi^{\perp}(z_2)| + |\Pi(z_1) - \Pi(z_2)|$$

 $\leq |\Pi(z_1) - \Pi(z_2)| + c(\tau, A)(d(z_1) + d(z_2)).$

Since $z_1 \in Q_i$ and $z_2 \in Q_{i'}$, we have $d(z_1) \leq \text{diam}(Q_i)$ and $d(z_2) \leq \text{diam}(Q_{i'})$. Moreover, if z_1 and z_2 are chosen so that

$$|\Pi(z_1) - \Pi(z_2)| \leq 2 \operatorname{dist} (\Pi(Q_i), \Pi(Q_{i'})),$$

then dist $(Q_i, Q_{i'}) \lesssim \operatorname{dist}(\Pi(Q_i), \Pi(Q_{i'})) + \operatorname{diam}(Q_i) + \operatorname{diam}(Q_{i'}) \lesssim_{\tau, A} \ell(J_i)$ as in (a).

For $i, i' \in I_0$ the properties (b) and (c) follow from (a) and Lemma 3.16. Indeed, in this case

diam
$$(Q_i) \approx \text{diam}(Q_{i'}) \approx_{\tau,A} \ell(J_i) \approx \ell(J_{i'})$$
 and $\text{dist}(Q_i, Q_{i'}) \lesssim_{\tau,A} \ell(J_i)$.

Taking into account that L_{Q_i} and $L_{Q_{i'}}$ are the graphs of F_i and $F_{i'}$, correspondingly, by Lemma 3.16 we have

$$|F_i(z) - F_{i'}(z)| \lesssim_{\tau,A} \sqrt{\varepsilon_0} \,\ell(J_i) \lesssim \varepsilon_0^{1/3} \,\ell(J_i), \qquad z \in 100 J_i,$$

if $\varepsilon_0 = \varepsilon_0(\tau, A)$ is chosen small enough. Moreover, by the same lemma $\measuredangle(L_{Q_i}, L_{Q_{i'}}) \lesssim_{\tau, A} \sqrt{\varepsilon_0}$ and thus

$$\begin{aligned} |F'_{i} - F'_{i'}| &= |\arctan\measuredangle(L_{Q_{i}}, L_{R}) - \arctan\measuredangle(L_{Q_{i'}}, L_{R})| \\ &= |\arctan\measuredangle(L_{Q_{i}}, L_{R}) - \arctan(\measuredangle(L_{Q_{i}}, L_{R}) \pm \measuredangle(L_{Q_{i}}, L_{Q_{i'}}))| \\ &\lesssim |\arctan\measuredangle(L_{Q_{i}}, L_{Q_{i'}})| \\ &\lesssim \tau_{\tau,A} \sqrt{\varepsilon_{0}} \\ &\lesssim \varepsilon_{0}^{1/3}, \end{aligned}$$

if $\varepsilon_0 = \varepsilon_0(\tau, A)$ is small enough.

For $i, i' \notin I_0$, $F_i \equiv F_{i'} \equiv 0$, and so (b) and (c) are trivial.

Finally, let $i \in I_0$ and $i' \notin I_0$. From the assumption $10J_i \cap 10J_{i'} \neq \emptyset$ and Lemma 3.19(b) we know that $\ell(J_i) \approx \ell(J_{i'})$. Moreover, by Lemma 3.20(a) we have $\ell(J_i) \leqslant \text{diam}(R)$ as $i \in I_0$. From another side, by Lemma 3.20(b)

$$\ell(J_{i'}) \approx \operatorname{dist}\left(\Pi(x_0), J_{i'}\right)$$

and additionally dist $(\Pi(x_0), J_{i'}) \ge 10 \operatorname{diam}(R)$ as $i' \notin I_0$, i.e. $J_{i'} \cap U_0 = \emptyset$. From all these facts we conclude that

$$\ell(J_i) \approx \ell(J_{i'}) \approx_{\tau,A} \operatorname{diam}(R) \quad \text{and} \quad \operatorname{dist}(J_i, J_{i'}) \lesssim_{\tau,A} \operatorname{diam}(R).$$

Recall that $F_{i'} \equiv 0$ and $J_{i'} \subset L_R$. Then, using Lemma 3.21 and arguments close to those in the proof of Lemmas 3.10 and 3.16, one can show that L_{Q_i} is very close to L_R in cB_0 , which yields (b) and (c) in this case if $\varepsilon_0 = \varepsilon_0(\tau, A)$ is chosen small enough.

3.10.3 Extension of F to the whole L_R

We are now ready to finish the definition of F on the whole L_R . Recall that F has already been defined on $\Pi(G_R)$ (see (3.29)). So it remains to define it only on $L_R \setminus \Pi(G_R)$. To this end, we first introduce a partition of unity on $L_R \setminus \Pi(G_R)$. For each $i \in I$, we can find a function $\tilde{\varphi}_i \in C^{\infty}(L_R)$ such that $\chi_{2J_i} \leq \tilde{\varphi}_i \leq \chi_{3J_i}$, with

$$|\tilde{\varphi}'_i| \leqslant rac{c}{\ell(J_i)} \qquad ext{and} \qquad |\tilde{\varphi}''_i| \leqslant rac{c}{\ell(J_i)^2}.$$

Then, for each $i \in I$, we set

$$\varphi_i = \frac{\tilde{\varphi}_i}{\sum_{j \in I} \tilde{\varphi}_j}.$$
(3.32)

It is clear that the family $\{\varphi_i\}_{i \in I}$ is a partition of unity subordinated to the sets $\{3J_i\}_{i \in I}$, and each function φ_i satisfies

$$|\varphi_i'| \leqslant \frac{c}{\ell(J_i)}$$
 and $|\varphi_i''| \leqslant \frac{c}{\ell(J_i)^2}$,

taking into account Lemma 3.19.

Recall that $L_R \setminus \Pi(G_R) = \bigcup_{i \in I} J_i = \bigcup_{i \in I} 3J_i$. For $z \in L_R \setminus \Pi(G_R)$, we define

$$F(z) := \sum_{i \in I_0} \varphi_i(z) F_i(z).$$

Observe that in the preceding sum we can replace I_0 by I as $F_i \equiv 0$ for $i \in I \setminus I_0$.

We denote by Γ_R the graph $\{(z, F(z)) : z \in L_R\}$.

Using the lemmas proved above, one can undeviatingly follow the "partition of unity" arguments in [Tol5, Section 7.5] to prove the following.

Lemma 3.23. The function $F : L_R \to L_R^{\perp}$ is supported on $L_R \cap B(\Pi(x_0), 12 \operatorname{diam}(R))$ and is $C_F \theta(R)$ -Lipschitz, where $C_F > 0$ is absolute. Also, if $z \in 15J_i$, $i \in I$, then

$$|F''(z)| \lesssim \frac{\sqrt[4]{\varepsilon_0}}{\ell(J_i)}.$$

Recall that we suppose of course that the parameters and thresholds mentioned in Section 3.7 are chosen properly.

3.10.4 Γ_R and R are close to each other

Lemma 3.24. There exists a constant $c_3(\tau, A) > 0$ such that

dist
$$(x, \Gamma_R) \leq c_3(\tau, A) \cdot d(x)$$
 for any $x \in B_0$. (3.33)

Proof. Let $y = (\Pi(x), F(\Pi(x)))$. By Lemma 3.18,

dist
$$(x, \Gamma_R) \leq |x - y| = |\Pi^{\perp}(x) - \Pi^{\perp}(y)| \lesssim_{\tau, A} d(x) + d(y).$$
 (3.34)

If $\Pi(x) \in \Pi(G_R)$, then $y \in G_R$ and thus $d(y) \equiv 0$, which proves the lemma.

If $\Pi(x) \notin \Pi(G_R)$, let J_i , $i \in I$, be such that $\Pi(x) \in J_i$. Since $\Pi(x) \in J_i \cap B_0 \neq \emptyset$, $i \in I_0$ and therefore there exists a cube $Q_i \in \mathsf{DbTree}(R)$ described in Lemma 3.21. This gives

$$d(y) \leq \operatorname{dist}(y, Q_i) + \operatorname{diam}(Q_i) \leq_{\tau, A} \operatorname{dist}(y, Q_i) + \ell(J_i).$$

Let us estimate dist (y, Q_i) . One can deduce from the definition of F that there exist $y' \in L_{Q_i}$ such that $\Pi(y') = \Pi(y)$ and dist $(y, y') \leq \ell(J_i)$ (recall that L_{Q_i} is the graph of F_i and $\Pi(y) \in J_i$, see some details in [Tol5, Proof of Lemma 7.24]). Moreover, it follows in a similar way as in the proof of Lemmas 3.10 and 3.16 that there exist $\zeta \in Q_i$ and $\zeta' \in L_{Q_i}$ such that dist $(\zeta, \zeta') \leq \sqrt{\varepsilon_0} \operatorname{diam}(Q_i)$. We know from Lemma 3.21 that dist $(\Pi(y'), \Pi(\zeta)) \leq \ell(J_i)$. Furthermore, it holds that $\measuredangle(L_{Q_i}, L_R) \leq \theta(R)$ by (3.19) in Lemma 3.9 taking into account that all $Q_i \in \mathsf{DbTree}(R)$. These facts imply that dist $(y', \zeta') \leq \ell(J_i)$. Summarizing, we obtain

dist
$$(y, Q_i) \leq dist(y, y') + dist(y', \zeta') + dist(\zeta', \zeta) \leq \ell(J_i).$$

From this by Lemma 3.19(a) and the definition of D (see (3.30)), we conclude that

$$d(y) \lesssim_{\tau,A} \ell(J_i) \lesssim_{\tau,A} D(\Pi(x)) \lesssim_{\tau,A} d(x).$$

This fact together with (3.34) proves the lemma.

Lemma 3.25. Let $\varepsilon_0 = \varepsilon_0(A, \tau)$ be small enough. If $Q \in \mathsf{DbTree}(R)$ and $z \in \Gamma_R \cap 2B_Q$, then

dist
$$(z, L_Q) \lesssim \sqrt[4]{\varepsilon_0} r(Q).$$
 (3.35)

Proof. Let $z \in G_R$. Then there exists $Q' \in \mathsf{DbTree}(R)$ such that $z \in Q', Q' \subset Q$ and $r(Q') \leq \varepsilon_0^{1/3} r(Q)$. By Lemma 3.10 there is $z' \in Q'$ such that dist $(z', z'') \leq \sqrt{\varepsilon_0} r(Q')$, where $z'' \in L_{Q'} \cap 2B_{Q'}$. Furthermore, it is clear that dist $(z, z') \leq r(Q') \leq \varepsilon_0^{1/3} r(Q)$. Using that $Q' \subset Q$, by Lemma 3.17 we get dist $(z'', L_Q) \leq \varepsilon_0^{1/3} r(Q)$. Consequently,

$$\operatorname{dist}(z, L_Q) \leqslant \operatorname{dist}(z, z') + \operatorname{dist}(z', z'') + \operatorname{dist}(z'', L_Q) \lesssim \varepsilon_0^{1/3} r(Q).$$

Now let $z \notin G_R$ and $\zeta = \Pi(z)$. In this case

$$F(\zeta) = \sum_{i \in I_0} \varphi_i(\zeta) F_i(\zeta).$$

Now take into account (3.32) and distinguish two cases. Suppose first that

$$\sum_{i\in I_0}\varphi_i(\zeta)=1$$

In this case $(\zeta, F(\zeta))$ is a convex combination of the points $(\zeta, F_i(\zeta))$ for *i* such that $\varphi_i(\zeta) \neq 0$ (we will write $i \in I_0$ for these *i* s, $I_0 \subset I_0$). Therefore (3.35) follows if

dist
$$((\zeta, F_i(\zeta)), L_Q) \lesssim \varepsilon_0^{1/3} r(Q)$$
 for all $i \in \tilde{I}_0$. (3.36)

To prove this estimate, notice that since $z \in 2B_Q$,

$$D(\zeta) \leqslant d(z) \lesssim r(Q)$$

Let $J_{i'}$, where $i' \in I_0$, be the interval that contains ζ . Then

$$\ell(J_{i'}) \leqslant \frac{1}{20} D(\zeta) \lesssim r(Q). \tag{3.37}$$

Recall that φ_i is supported on $3J_i$. Consequently, we necessarily have $3J_i \cap J_{i'} \neq \emptyset$ if $i \in I_0$. Therefore by Lemma 3.19(b) and 3.21(a),

$$\ell(J_i) \approx_{\tau,A} \operatorname{diam}(Q_i) \approx_{\tau,A} \operatorname{diam}(Q_{i'}) \approx_{\tau,A} \ell(J_{i'}) \lesssim_{\tau,A} r(Q).$$

Moreover, by Lemma 3.22(a),

dist
$$(\Pi(Q_i), \Pi(Q_{i'})) \leq \operatorname{dist}(Q_i, Q_{i'}) \leq_{\tau, A} \ell(J_i).$$

Taking into account that

$$\operatorname{dist}\left(\Pi(Q_{i'}),\Pi(Q)\right) \leqslant \operatorname{dist}\left(\Pi(Q_{i'}),J_{i'}\right) + \operatorname{diam}\left(J_{i'}\right) + \operatorname{dist}\left(J_{i'},\Pi(Q)\right) \lesssim \ell(J_{i'}) \lesssim_{\tau,A} r(Q),$$

we get

$$\operatorname{dist}\left(\Pi(Q_i), \Pi(Q)\right) \leqslant \operatorname{dist}\left(\Pi(Q_i), \Pi(Q_{i'})\right) + \operatorname{diam}\left(\Pi(Q_{i'})\right) + \operatorname{dist}\left(\Pi(Q_{i'}), \Pi(Q)\right) \lesssim_{\tau, A} r(Q).$$

From Lemma 3.18, applied for $z_1 \in Q_i$ and $z_2 \in Q$, we deduce that

dist
$$(Q_i, Q) \lesssim \operatorname{dist}(\Pi(Q_i), \Pi(Q)) + \operatorname{diam}(Q_i) + \operatorname{diam}(Q) \lesssim_{\tau, A} r(Q).$$

This means that $2B_{Q_i} \subset cB_Q$ with some $c = c(\tau, A) > 1$. Consequently, by Lemmas 3.11 and 3.16, we can find $Q' \in \mathsf{DbTree}(R)$ such that $cB_Q \subset 2B_{Q'}$, diam $(Q') \approx_{A,\tau} \operatorname{diam}(Q)$ and

dist
$$(w, L_Q) \lesssim_{A,\tau} \sqrt{\varepsilon_0} (\operatorname{dist}(w, Q') + \operatorname{diam}(Q')), \qquad w \in L_{Q'}.$$

Choosing $\varepsilon_0 = \varepsilon_0(A, \tau)$ small enough, we get

$$\operatorname{dist}(w, L_Q) \lesssim \varepsilon_0^{1/3} (\operatorname{dist}(w, Q') + \operatorname{diam}(Q')), \qquad w \in L_{Q'}.$$
(3.38)

Recall that $(\zeta, F_i(\zeta)) \in L_{Q_i} \cap cB_{Q_i}$ and $2B_{Q_i} \subset 2B_{Q'}$ so Lemma 3.17 gives

dist
$$((\zeta, F_i(\zeta)), L_{Q'}) \lesssim \varepsilon_0^{1/3} r(Q').$$

Note that the parameters and thresholds in Lemma 3.17 are also supposed to be properly chosen. Together with (3.38) applied to $w = \operatorname{proj}_{L_{\Omega'}}(\zeta, F_i(\zeta))$, this yields (3.36) as required.

Suppose now that

$$\sum_{i\in I_0}\varphi_i(\zeta)<1.$$

In this case, there exists some $J_{i'}$ with $i' \notin I_0$ such that $\zeta \in 3J_{i'}$ (as from (3.32) it follows that $\sum_{i \in I \setminus I_0} \varphi_i(\zeta) > 0$) and by Lemma 3.20(b),

diam
$$(R) \lesssim \ell(J_{i'}) \approx \text{dist} (\Pi(x_0), J_{i'}).$$

Moreover, if J_i is the interval that contains $\zeta = \Pi(z), z \in 2B_Q$, then

$$\ell(J_i) \lesssim D(\Pi(z)) \lesssim d(z) \lesssim \operatorname{dist}(z, Q) + \operatorname{diam}(Q) \lesssim \operatorname{diam}(R),$$

where we used the definition of D, see (3.30).

By Lemma 3.19(b), $\ell(J_i) \approx \ell(J_{i'})$ as $J_i \cap 3J_{i'} \neq \emptyset$. That is why $\ell(J_{i'}) \approx \text{diam}(R)$. This also implies that $\ell(J_m) \approx \text{diam}(R)$ for any $m \in I_0$ such that $\zeta \in 3J_m$. By Lemma 3.21(*a*), it means that $\text{diam}(Q_m) \approx_{\tau,A} \text{diam}(R)$. Furthermore, it is clear that $\text{dist}(Q_m, R) \equiv 0$ and so the assumptions of Lemma 3.16 are satisfied for Q_m and R. Consequently, L_{Q_m} and L_R are very close in cB_R for some c > 1 if the corresponding parameters are chosen properly, namely,

dist
$$_{H}(L_{Q_{m}} \cap cB_{R}, L_{R} \cap cB_{R}) \lesssim_{\tau, A} \sqrt{\varepsilon_{0}} \operatorname{diam}(R).$$
 (3.39)

On the other hand, arguing as in (3.37), one deduces that $\ell(J_m) \lesssim_{\tau,A} r(Q)$, and from this we conclude that $r(Q) \approx_{\tau,A} \operatorname{diam}(R)$. By (3.39) then we get

$$|F_m(\zeta)| = \operatorname{dist}\left((\zeta, F_m(\zeta)), L_R\right) \lesssim_{\tau, A} \sqrt{\varepsilon_0} \operatorname{diam}\left(R\right) \lesssim_{\tau, A} \sqrt{\varepsilon_0} r(Q) \lesssim \varepsilon_0^{1/3} r(Q)$$

for all above-mentioned ms is $\varepsilon_0 = \varepsilon_0(\tau, A)$ is chosen small enough. Recall that we only need to sum up $i \in I_0$ such that $\zeta \in 3J_i$ and these are our $m \in I_0$. Thus

dist
$$((\zeta, F(\zeta)), L_R) \leq \sum_{i \in I_0} \varphi_i(\zeta) |F_i(\zeta)| = \sum_{m \in I_0} \varphi_m(\zeta) |F_m(\zeta)|$$

 $\leq \max_{m \in I_0} |F_m(\zeta)| \sum_{m \in I_0} \varphi_m(\zeta) \lesssim \varepsilon_0^{1/3} r(Q).$

Due to the fact that $r(Q) \approx \operatorname{diam}(R)$, by Lemma 3.16 lines L_R and L_Q are very close to each other in $2B_Q$, and thus

dist
$$((\zeta, F(\zeta)), L_Q) \lesssim \varepsilon_0^{1/3} r(Q)$$

as desired.

Lemma 3.26. For all $x \in R \setminus R_{\mathsf{Far}}$,

dist
$$(x, \Gamma_R) \lesssim \sqrt[4]{\varepsilon_0} d(x).$$
 (3.40)

Proof. Recall that if d(x) = 0, then $x \in \Gamma_R$ and we are done.

By Lemmas 3.13 and 3.25 any point $x \in R \setminus R_{\mathsf{Far}}$ is very close to L_R and (3.40) clearly holds if $d(x) \approx \operatorname{diam}(R)$. Hence, we may suppose below that d(x) is small with respect to $\operatorname{diam}(R)$, say, $d(x) \ll (\mathsf{c}_3(\tau, A) + 2) \operatorname{diam}(R)$, where $\mathsf{c}_3(\tau, A) > 0$ is from Lemma 3.24.

Given $x \in R \setminus R_{\mathsf{Far}}$ with d(x) > 0, take a cube $Q \in \mathsf{DbTree}(R)$ such that

$$\operatorname{dist}(x,Q) + \operatorname{diam}(Q) \leq 2d(x)$$

Take any $z \in Q$ (note that dist $(z, x) \leq 2d(x)$) and find $Q' \in \mathsf{DbTree}(R)$ such that

$$B(z, 2(c_3(\tau, A) + 2)d(x)) \subset \frac{3}{2}B_{Q'}.$$

Recall that d(x) is small with respect to diam (R) and thus Q' can be found. We can also guarantee that $r(Q') \approx_{\tau,A} d(x)$. Furthermore, it is clear that $x \in B(z, 2(c_3(\tau, A) + 2)d(x))$ and thus $x \in \frac{3}{2}B_{Q'}$. Moreover, Lemma 3.24 gives

dist
$$(z, \Gamma_R) \leq dist(z, x) + dist(x, \Gamma_R) \leq (2 + c_3(\tau, A))d(x),$$

which yields that $B(z, 2(c_3(\tau, A) + 2)d(x)) \cap \Gamma_R \neq \emptyset$ and therefore

$$\frac{3}{2}B_{Q'}\cap\Gamma_R\neq\emptyset.$$

Take into account that $x \in \frac{3}{2}B_{Q'} \cap R \setminus R_{\mathsf{Far}} \subset 2B_{Q'}^{\mathsf{Cl}}$, i.e. dist $(x, L_{Q'}) \lesssim \sqrt{\varepsilon_0} r(Q')$ and thus there is $x' \in L_{Q'} \cap 2B_{Q'}$ such that dist $(x, x') \lesssim \sqrt{\varepsilon_0} r(Q')$. Furthermore, Lemma 3.25 says that dist $(y, L_{Q'}) \leqslant c\varepsilon_0^{-1/3} r(Q')$ for any $y \in \Gamma_R \cap 2B_{Q'}$ and some c > 0. In other words,

$$\Gamma_R \cap 2B_{Q'} \subset \mathcal{U}_{c\varepsilon_0^{1/3} r(Q')}(L_{Q'}),$$

and thus dist $(x', \Gamma_R) \lesssim \varepsilon_0^{1/3} r(Q')$. Summarising, we get

dist
$$(x, \Gamma_R) \leq dist(x, x') + dist(x', \Gamma_R) \lesssim \varepsilon_0^{1/3} r(Q').$$

It is left to remember that $r(Q') \approx_{\tau,A} d(x)$ by construction and to choose $\varepsilon_0 = \varepsilon_0(\tau, A)$ small enough.

Lemma 3.27. For each $i \in I_0$,

dist
$$(Q_i, \Gamma_R \cap \Pi^{-1}(J_i)) \lesssim_{\tau, A} \ell(J_i).$$

Proof. Let $x \in Q_i \subset B_0$. Then by Lemmas 3.24 and 3.21(a) we have

dist
$$(Q_i, \Gamma_R) \leq dist(x, \Gamma_R) \lesssim_{\tau, A} d(x) \lesssim_{\tau, A} diam(Q_i) \approx_{\tau, A} \ell(J_i).$$

Moreover, dist $(J_i, \Pi(Q_i)) \leq \ell(J_i)$ by Lemma 3.21(b). From these two inequalities and Lemma 3.23, the required result follows.

We finish this section with one more result which can be easily deduced from Lemmas 3.23 (look at spt F) and 3.25.

Lemma 3.28. For any $z \in \Gamma_R$, it holds that

dist
$$(z, L_R) \lesssim \sqrt[4]{\varepsilon_0} r(R)$$
.

3.11 Small measure of the cubes from LD(R)

In what follows we show that the measure of the low-density cubes is small.

Lemma 3.29. If $\varepsilon_0 = \varepsilon_0(\tau, A)$ and τ are small enough, then

$$\sum_{Q \in \mathsf{LD}(R)} \mu(Q) \leqslant \frac{1}{3} \sqrt{\tau} \, \mu(R). \tag{3.41}$$

Proof. Recall that the parameters and thresholds from Section 3.7 are supposed to be chosen so that all above-stated results hold true. Taking this into account, note that by Lemma 3.12 with $\alpha = \alpha(\tau)$, being small enough, we have

$$\mu(R_{\mathsf{Far}}) \leqslant \frac{1}{6}\sqrt{\tau}\mu(R),$$

thus for obtaining (3.41) it suffices to show that

$$\mu(\mathcal{S}_{\mathsf{LD}}) \leq \frac{1}{6}\sqrt{\tau}\mu(R), \quad \text{where } \mathcal{S}_{\mathsf{LD}} := \bigcup_{Q \in \mathsf{LD}(R)} Q \setminus R_{\mathsf{Far}}.$$
(3.42)

By the Besicovitch covering theorem, there exist a countable collection of points $x_i \in S_{LD}$ such that

$$\mathcal{S}_{\mathsf{LD}} \subset \bigcup_{i} B(x_i, r(Q_i)) \quad \text{and} \quad \sum_{i} \chi_{B(x_i, r(Q_i))} \leqslant N,$$

where $Q_i \in \mathsf{LD}(R)$ is such that $x_i \in Q_i$, and N is some fixed constant. Note that $B(x_i, r(Q_i)) \subset 2B_{Q_i}$. From this it follows that

$$\mu(\mathcal{S}_{\mathsf{LD}}) \leqslant \sum_{i} \mu(B(x_i, r(Q_i))) \leqslant \sum_{i} \mu(2B_{Q_i}) \lesssim \sum_{i} \Theta_{\mu}(2B_{Q_i}) r(Q_i).$$

Since $Q_i \in \mathsf{LD}(R)$, we have $\Theta_{\mu}(2B_{Q_i}) < \tau \Theta_{\mu}(2B_R)$ by definition. Furthermore, each $x_i \in \mathcal{S}_{\mathsf{LD}}$ satisfies Lemma 3.26 and moreover $d(x_i) \lesssim_{\tau,A} \operatorname{diam}(Q_i)$ (as x_i also belongs to the first doubling ancestor of Q_i with a comparable diameter with comparability constant $\lambda = \lambda(\tau, A)$, see Lemma 3.11) so that

dist
$$(x_i, \Gamma_R) \lesssim_{\tau, A} \sqrt[4]{\varepsilon_0} r(Q_i) \lesssim \sqrt[8]{\varepsilon_0} r(Q_i),$$

if $\varepsilon_0 = \varepsilon_0(\tau, A)$ is small enough. This means that Γ_R passes very close to the center of $B(x_i, r(Q_i))$ in terms of $r(Q_i)$. Consequently,

$$r(Q_i) \lesssim \mathcal{H}^1(\Gamma_R \cap B(x_i, r(Q_i)))$$

as Γ_R is a connected graph of a Lipschitz function. Thus we get

$$\mu(\mathcal{S}_{\mathsf{LD}}) \lesssim \tau \Theta_{\mu}(2B_R) \sum_{i} \mathcal{H}^1(\Gamma_R \cap B(x_i, r(Q_i))).$$

Since $\sum_{i} \chi_{B(x_i, r(Q_i))} \leq N$ with an absolute constant N, we get by Lemma 3.23 that

$$\sum_{i} \mathcal{H}^{1}(\Gamma_{R} \cap B(x_{i}, r(Q_{i}))) \lesssim \mathcal{H}^{1}\left(\Gamma_{R} \cap \bigcup_{i} B(x_{i}, r(Q_{i}))\right) \lesssim \mathcal{H}^{1}(\Gamma_{R} \cap 2B_{R}) \lesssim r(B_{R}).$$

From this we deduce that

$$\mu(\mathcal{S}_{\mathsf{LD}}) \lesssim \tau \Theta_{\mu}(2B_R) r(B_R) \lesssim \tau \mu(2B_R) \lesssim \tau \mu(R),$$

where the latter inequality is due to the fact that $R \in \mathcal{D}^{db}$ by construction. Finally, we obtain (3.42) if τ is chosen small enough.

3.12 Small measure of the cubes from BS(R) for R whose best approximation line is far from the vertical

3.12.1 Auxiliaries and the key estimate for the measure of cubes from $\mathsf{BS}(R)$

Given some $\theta_0 > 0$, we say that

$$R \in \mathsf{T}_{VF}(\theta_0) \text{ and } \theta(R) = \theta_0, \qquad \text{if} \quad \theta_V(L_R) \ge (1 + C_F) \,\theta_0;$$

$$R \notin \mathsf{T}_{VF}(\theta_0) \text{ and } \theta(R) = 2(1 + C_F) \,\theta_0, \qquad \text{if} \quad \theta_V(L_R) < (1 + C_F) \,\theta_0.$$

Note that $C_F > 0$ is an absolute constant from Lemma 3.23 where it is stated that the function F is $C_F \theta(R)$ -Lipschitz. Recall that θ_0 and $\theta(R)$ were first introduced and used in Sections 3.7 and 3.8.1.

Let $R \in \mathsf{T}_{VF}(\theta_0)$. From the definition of the family $\mathsf{BS}(R)$ it follows that in this case we have

$$\measuredangle(L_Q, L_R) > \theta_0 \qquad \forall Q \in \mathsf{BS}(R). \tag{3.43}$$

On the other hand, if $Q \in \mathsf{DbTree}(R)$, then $\measuredangle(L_Q, L_R) \leq \theta_0$ and thus

$$\theta_V(L_Q) \ge (1+C_F) \theta_0 - \measuredangle(L_Q, L_R) \ge C_F \theta_0 \qquad \forall Q \in \mathsf{DbTree}(R)$$

In this section we are going to deal with $R \in \mathsf{T}_{VF}(\theta_0)$ only. Our aim is to prove the following assertion.

Lemma 3.30. For any $R \in \mathsf{T}_{VF}(\theta_0)$, if $\varepsilon_0 = \varepsilon_0(\tau)$ is chosen small enough, then

$$\sum_{Q \in \mathsf{BS}(R)} \mu(Q) \leqslant \frac{1}{3} \sqrt{\tau} \mu(R).$$

The rest of this section is devoted to the proof of this lemma.

Remark 3.7. It is natural to suppose in this section that BS(R) is not empty. This and Remark 3.5 imply that $R \notin Stop(R)$ and thus $Tree(R) \setminus Stop(R)$ is not empty.

3.12.2 The measure of cubes from BS(R) is controlled by the permutations of the Hausdorff measure restricted to Γ_R

Recall that the parameters and thresholds from Section 3.7 are supposed to be chosen so that all above-stated results hold. Taking this into account, note that by Lemma 3.12 with $\alpha = \alpha(\tau)$, being small enough, we have

$$\mu(R_{\mathsf{Far}}) \leqslant \frac{1}{6}\sqrt{\tau}\mu(R),$$

thus to prove Lemma 3.30, it suffices to show that

$$\mu(\mathcal{S}_{\mathsf{BS}}) \leq \frac{1}{6}\sqrt{\tau}\mu(R), \quad \text{where } \mathcal{S}_{\mathsf{BS}} := \bigcup_{Q \in \mathsf{BS}(R)} Q \setminus R_{\mathsf{Far}}.$$
(3.44)

The following results is the first step in proving (3.44). (Recall the identity (0.15).)

Lemma 3.31. If θ_0 and $\varepsilon_0 = \varepsilon_0(\theta_0, \tau, A)$ are chosen small enough, then

$$\mu(\mathcal{S}_{\mathsf{BS}}) \lesssim_A \frac{p_{\infty}(\Theta_{\mu}(2B_R) \,\mathcal{H}_{\Gamma_R}^1)}{\theta_0^2 \,\Theta_{\mu}(2B_R)^2}$$

Proof. For every $x \in S_{BS}$ take the ball $B(x, r(Q_x))$, where $Q_x \in BS(R)$ and is such that $x \in Q_x$. By the 5*r*-covering theorem there exists a subfamily of pairwise disjoint balls $\{B(x_i, r(Q_i))\}_{i \in \hat{I}}$, where $Q_i = Q_{x_i}$, such that

$$\mathcal{S}_{\mathsf{BS}} \subset R \cap \bigcup_{i \in \hat{I}} B(x_i, 5r(Q_i)).$$

Let $B_i = B(x_i, \frac{1}{2}r(Q_i)), i \in \hat{I}$. Clearly, $B_i \subset B_{Q_i}$. Moreover, take into account that $Q_i \in \mathcal{D}^{db}$ by the stopping condition (S3) and that $\mathcal{S}_{\mathsf{BS}} \cap R_{\mathsf{Far}} = \emptyset$ by definition. Therefore, by Lemma 3.26,

dist
$$(x_i, \Gamma_R) \lesssim \sqrt[4]{\varepsilon_0} d(x_i) \lesssim \sqrt[4]{\varepsilon_0} r(Q_i) < \frac{1}{4} r(Q_i),$$

if ε_0 is small enough. Thus $\Gamma_R \cap \frac{1}{2}B_i \neq \emptyset$ and therefore there exist $y_1, y_2 \in \Gamma_R \cap B_i$ such that

$$cr(Q_i) \leq |y_1 - y_2| \leq |\Pi(y_1) - \Pi(y_2)|$$

with some small fixed constant c > 0, where in the latter inequality we took into account that Γ_R is a graph of a Lipschitz function F (see Lemma 3.23).

Now, by Lemma 3.11, there exists $\tilde{Q}_i \in \mathsf{DbTree}(R)$ such that $Q_i \subset \tilde{Q}_i$ and moreover diam $(Q_i) \approx_{\tau,A} \operatorname{diam}(\tilde{Q}_i)$. By Lemma 3.25,

dist
$$(y_k, L_{\tilde{Q}_i}) \lesssim \varepsilon_0^{1/3} r(\tilde{Q}_i), \qquad y_k \in \Gamma_R \cap B_i, \quad k = 1, 2.$$

At the same time, $\angle(L_{Q_i}, L_{\tilde{Q}_i}) \lesssim_{\tau,A} \sqrt[4]{\varepsilon_0}$ by arguments similar to those in the proof of Lemma 3.16 (this lemma cannot be applied directly as $Q_i \notin \mathsf{DbTree}(R)$ but the arguments can still be adapted if one of the cubes is in $\mathsf{BS}(R)$). Therefore, if $\varepsilon_0 = \varepsilon_0(\tau, A)$ is small enough, then one can show that

dist
$$(y_k, L_{Q_i}) \lesssim \sqrt[8]{\varepsilon_0} r(Q_i), \qquad y_k \in \Gamma_R \cap B_i \subset \Gamma_R \cap 2B_{Q_i}, \quad k = 1, 2.$$

Consequently, denoting by y'_k the orthogonal projections of y_k onto L_{Q_i} , we get

$$|y_k - y'_k| \lesssim \sqrt[8]{\varepsilon_0} r(Q_i), \qquad k = 1, 2.$$

Since $\measuredangle(L_{Q_i}, L_R) > \theta_0$ by (3.43) and $\varepsilon_0 = \varepsilon_0(\theta_0)$ is small enough, it holds that

$$\begin{split} |F(\Pi(y_1)) - F(\Pi(y_2))| \\ &= |\Pi^{\perp}(y_1) - \Pi^{\perp}(y_2)| \ge |\Pi^{\perp}(y_1') - \Pi^{\perp}(y_2')| - \sum_k |y_k - y_k'| \\ &\gtrsim \theta_0 |\Pi(y_1') - \Pi(y_2')| - \sum_k |y_k - y_k'| \gtrsim \theta_0 |\Pi(y_1) - \Pi(y_2)| - 2\sum_k |y_k - y_k'| \\ &\gtrsim \theta_0 r(Q_i) - \sqrt[8]{\varepsilon_0} r(Q_i) \gtrsim \theta_0 r(Q_i), \end{split}$$

where k = 1, 2. Thus,

$$\int_{\Pi(B_i)} |F'(z)| dz \ge \left| \int_{\Pi(y_1)}^{\Pi(y_2)} F'(z) dz \right| = |F(\Pi(y_1)) - F(\Pi(y_2))| \ge \theta_0 r(Q_i).$$

This and Hölder's inequality yield

$$\theta_0 r(Q_i) \lesssim \sqrt{r(B_i)} \|F'\|_{2,\Pi(B_i)} \approx \sqrt{r(Q_i)} \|F'\|_{2,\Pi(B_i)},$$

and finally

$$r(Q_i) \lesssim \theta_0^{-2} \|F'\|_{2,\Pi(B_i)}^2.$$

Since the balls $2B_i$, $i \in \hat{I}$, are pairwise disjoint by construction, so are the intervals $\Pi(B_i) \subset L_R$, $i \in \hat{I}$, if θ_0 is chosen small enough. This is a consequence of the fact that x_i , the centres of B_i , lie very close to Γ_R , namely, dist $(x_i, \Gamma_R) \leq \sqrt[4]{\varepsilon_0} r(B_i)$, and moreover $\measuredangle(L_{x_i,x_j}, L_R) \leq \theta_0$ for all $i, j \in \hat{I}$ as Γ_R is Lipschits with constant $\leq \theta_0$, see Lemma 3.23. By this reason we have

$$\begin{split} \mu(\mathcal{S}_{\mathsf{BS}}) &\leqslant \sum_{i \in \hat{I}} \mu(B(x_i, 5r(Q_i))) \lesssim \sum_{i \in \hat{I}} \Theta_{\mu}(2B_{Q_i})r(Q_i) \\ &\lesssim_A \theta_0^{-2} \Theta_{\mu}(2B_R) \sum_{i \in \hat{I}} \|F'\|_{2,\Pi(B_i)}^2 \lesssim_A \theta_0^{-2} \Theta_{\mu}(2B_R) \|F'\|_2^2. \end{split}$$

Now take into account that under the assumption that $||F'||_{\infty} \leq 1/10$ (which is satisfied if θ_0 is sufficiently small) by [Tol5, Lemma 3.9] we have

$$||F'||_2^2 \approx p_\infty(\mathcal{H}^1_{\Gamma_R}) \approx \Theta_\mu(2B_R)^{-3} \, p_\infty(\Theta_\mu(2B_R) \, \mathcal{H}^1_{\Gamma_R})$$

with some absolute constants.

We claim that $p_{\infty}(x, y, z)$ is well controlled by $p_0(x, y, z)$ for any $x, y \in \Gamma_R$ if $R \in \mathsf{T}_{VF}(\theta_0)$.

Lemma 3.32. If $R \in \mathsf{T}_{VF}(\theta_0)$, then

$$p_{\infty}(x, y, z) \lesssim_{\theta_0} p_0(x, y, z)$$
 for any $x, y \in \Gamma_R$.

Proof. The fact that the function F (whose graph is Γ_R) is $C_F \theta(R)$ -Lipschitz by Lemma 3.23 and the definitions at the beginning of Subsection 3.12.1 yield

$$\measuredangle(L_{xy}, L_R) \leqslant C_F \,\theta_0 \qquad \text{and} \qquad \theta_V(L_R) \geqslant (1 + C_F) \,\theta_0.$$

Consequently,

$$\theta_V(L_{xy}) \ge \theta_V(L_R) - \measuredangle(L_{xy}, L_R) \ge (1 + C_F) \,\theta_0 - C_F \,\theta_0 = \theta_0.$$

Therefore $(x, y, z) \in V_{\mathsf{Far}}(\theta_0)$ and it is left to use Lemma 3.6.

For $x \in \mathbb{C}$ such that $\Pi(x) \notin \Pi(G_R)$, set

$$J_x = J_i, \quad i \in I, \quad \text{such that } \Pi(x) \in J_i,$$

and

$$\ell_x = \ell(J_x)$$

If $\Pi(x) \in \Pi(G_R)$, we write

$$J_x = \Pi(x)$$
 and $\ell_x = 0$,

i.e. one should think that in this case the point $\Pi(x)$ is a degenerate interval J_x with zero side length. To simplify notation, throughout this section we also write

$$x_1 = \Pi(x)$$
 and $x_2 = \Pi^{\perp}(x)$.

Recall that the intervals $\{J_i\}$, $i \in I_0$, are the ones from $\{J_i\}$, $i \in I$, that intersect the ball $B_0 = B(\Pi(x_0), 10 \operatorname{diam}(R))$, where $x_0 \in R$ is such that $\operatorname{dist}(x_0, L_R) \leq r(R)$ (see (3.31)). Observe that if $z \in U_0 = L_R \cap B_0$, then $D(z) \leq r(R)$. Thus $\ell(J_i) \leq r(R)$ for all $i \in I_0$. Thus, setting

$$\Gamma_{B_0} = G_R \cup \bigcup_{i \in I_0} \Gamma_R \cap \Pi^{-1}(J_i),$$

we deduce that $\Gamma_{B_0} \subset c'B_0$ with some fixed c' > 0. It is also true that $B_0 \subset c''B_R$ with some c'' > 0 and thus

$$\Gamma_{B_0} \subset cB_R$$
 with some $c > 0$.

One can actually tune constants to guarantee that

$$\Gamma_{B_0} \subset \bigcup_{Q \in \mathsf{Tree}(R)} 2B_Q \subset 2B_R,$$

so we will suppose this in what follows.

Clearly, $\Pi(\Gamma_{B_0})$ is an interval on L_R and therefore Γ_{B_0} is a connected subset of Γ_R . We also set

$$\Gamma_{\mathbf{Ext}(B_0)} = \Gamma_R \setminus \Gamma_{B_0}$$

First we will show that the part of the permutations of $\mathcal{H}^1_{\Gamma_R}$ that involves $\Gamma_{\mathbf{Ext}(B_0)}$ is very small.

Lemma 3.33. We have

$$p_{\infty}(\Theta_{\mu}(2B_{R}) \mathcal{H}^{1}_{\Gamma_{\mathbf{Ext}(B_{0})}}, \Theta_{\mu}(2B_{R}) \mathcal{H}^{1}_{\Gamma_{R}}, \Theta_{\mu}(2B_{R}) \mathcal{H}^{1}_{\Gamma_{R}}) \lesssim \sqrt[8]{\varepsilon_{0}} \Theta_{\mu}(2B_{R})^{2} \mu(R).$$

Proof. The proof is analogous (up to constants) to the proof of [Tol5, Lemma 7.36], where we should use our Lemmas 3.23 and 3.28 instead of [Tol5, Lemma 7.27 and Lemma 7.32].

What is more, it can be easily seen that

$$p_{\infty}(\Theta_{\mu}(2B_R) \mathcal{H}^{1}_{\Gamma_R}) \leq p_{\infty}(\Theta_{\mu}(2B_R) \mathcal{H}^{1}_{\Gamma_{B_0}}) + 3p_{\infty}(\Theta_{\mu}(2B_R) \mathcal{H}^{1}_{\Gamma_{\mathbf{Ext}(B_0)}}, \Theta_{\mu}(2B_R) \mathcal{H}^{1}_{\Gamma_R}, \Theta_{\mu}(2B_R) \mathcal{H}^{1}_{\Gamma_R}).$$
(3.45)

Consequently, taking into account Lemmas 3.31 and 3.33, we are now able to reduce the proof of Lemma 3.30 to the proof of a proper estimate for $p_{\infty}(\Theta_{\mu}(2B_R) \mathcal{H}^1_{\Gamma_{B_0}})$, where $\Gamma_{B_0} \subset cB_R$ with some c > 0. For short, we will write

$$\sigma := \Theta_{\mu}(2B_R) \,\mathcal{H}^1_{\Gamma_{B_0}}.$$

Thus, using this notation, we are aimed to prove the following lemma in the forthcoming subsections.

Lemma 3.34. It holds that

$$p_{\infty}(\sigma) \lesssim \varepsilon_0^{1/40} \Theta_{\mu} (2B_R)^2 \ \mu(R).$$

3.12.3 Estimates for the permutations of the Hausdorff measure restricted to Γ_R

Recall that, for $x \in \mathbb{C}$, we set $\ell_x = \ell(J_x)$. Let $x, y \in \Gamma_R$. We say that x and y are

• very close and write

$$(x,y) \in \mathbf{VC}, \quad \text{if } |x_1 - y_1| \leq \ell_x + \ell_y;$$

• close and write

$$(x,y) \in \mathbf{C},$$
 if $|x_1 - y_1| \leq \varepsilon_0^{-1/20} (\ell_x + \ell_y);$

• far and write

$$(x,y) \in \mathbf{F},$$
 if $|x_1 - y_1| > \varepsilon_0^{-1/20} (\ell_x + \ell_y).$

Notice that the relations are symmetric with respect to x and y.

Given $(x, y, z) \in \Gamma^3_{B_0}$, there are three possibilities: either two of the points in the triple are very close, or no pair of points is very close but there is at least one pair that is close, or all the pairs of points are far. So we can split $p_{\infty}(\sigma)$ as follows:

$$p_{\infty}(\sigma) \leq 3 \iiint_{(x,y)\in\mathbf{VC}} p_{\infty}(x,y,z) \, d\sigma(x) \, d\sigma(y) \, d\sigma(z) + 3 \iiint_{\substack{(x,y)\in\mathbf{C}\setminus\mathbf{VC}\\(x,z)\notin\mathbf{VC}\\(y,z)\notin\mathbf{VC}}} p_{\infty}(x,y,z) \, d\sigma(x) \, d\sigma(y) \, d\sigma(z) + \iiint_{\substack{(x,y)\in\mathbf{F}\\(y,z)\in\mathbf{F}\\(y,z)\in\mathbf{F}}} p_{\infty}(x,y,z) \, d\sigma(x) \, d\sigma(y) \, d\sigma(z) =: p_{\infty,\mathbf{VC}}(\sigma) + p_{\infty,\mathbf{C}\setminus\mathbf{VC}}(\sigma) + p_{\infty,\mathbf{F}}(\sigma).$$

$$(3.46)$$

A straightforward adaptation of the arguments from [Tol5, Section 7.8.2, Lemmas 7.38 and 7.39] to our settings gives the following.

Lemma 3.35. If $\varepsilon_0 = \varepsilon_0(\tau, A)$ and $\alpha = \alpha(\theta_0, \varepsilon_0, \tau, A)$ are chosen small enough, then

$$p_{\infty,\mathbf{VC}}(\sigma) + p_{\infty,\mathbf{C}\setminus\mathbf{VC}}(\sigma) \lesssim \varepsilon_0^{1/40} \Theta_\mu (2B_R)^2 \mu(R).$$

Now we are going to prove the following result that actually finishes the proof of Lemma 3.34 and therefore Lemma 3.30 taking into account (3.46) and Lemma 3.35.

Lemma 3.36. If $\varepsilon_0 = \varepsilon_0(\tau, A)$, $\alpha = \alpha(\theta_0, \varepsilon_0, \tau, A)$ and $\delta = \delta(\varepsilon_0)$ are small enough, then

$$p_{\infty,\mathbf{F}}(\sigma) \lesssim \varepsilon_0^{1/40} \,\Theta_\mu (2B_R)^2 \,\mu(R). \tag{3.47}$$

The proof of Lemma 3.36 is similar to the one of [Tol5, Lemma 7.40] but necessary changes are not straightforward so we give details. First we need to approximate the measure σ by another measure absolutely continuous with respect to μ , of the form $g\mu$, with some $g \in L^{\infty}(\mu)$. This is carried out by the next lemma, where we say that

$$i \in I'_0$$
 if $i \in I_0$ and $\mu(Q_i \setminus R_{\mathsf{Far}}) \ge \frac{3}{4}\mu(Q_i),$ (3.48)

for the cubes $Q_i \in \mathsf{DbTree}(R)$ from Lemma 3.21 associated with the intervals J_i , $i \in I_0$. Recall the definition of R_{Far} in Section 3.8.2 and Lemmas 3.12 and 3.13. In what follows we will also write

$$\widehat{J_i} := \Gamma_R \cap \Pi^{-1}(J_i).$$

Lemma 3.37. For each $i \in I'_0$ there exists a non-negative function $g_i \in L^{\infty}(\mu)$, supported on $A_i \subset Q_i \setminus R_{\mathsf{Far}}$, where $Q_i \in \mathsf{DbTree}(R)$ are associated with the intervals J_i by Lemma 3.21, and such that

$$\int g_i d\mu = \Theta_\mu(2B_R) \mathcal{H}^1(\widehat{J}_i) = \sigma(\widehat{J}_i), \qquad (3.49)$$

and

$$\sum_{i \in I'_0} g_i \lesssim_{\tau,A} 1. \tag{3.50}$$

Proof. Assume first that the family $\{J_i\}_{i \in I'_0}$ is finite. Suppose also that $\ell(J_i) \leq \ell(J_{i+1})$ for all $i \in I'_0$. We will construct

$$g_i = \alpha_i \chi_{A_i}, \quad \text{where} \quad \alpha_i \geqslant 0 \quad \text{and} \quad A_i \subset Q_i \setminus R_{\mathsf{Far}}.$$

We set

$$\alpha_1 = \frac{\sigma(J_1)}{\mu(A_1)}$$
 and $A_1 = Q_1 \setminus R_{\mathsf{Far}},$

so that $\int g_1 d\mu = \sigma(\hat{J}_1)$. Furthermore, by (3.15) in Lemma 3.9, Lemmas 3.21 and 3.23 and the condition (3.48) we get

$$\|g_1\|_{\infty} = \alpha_1 \lesssim_{\tau,A} \frac{\Theta_{\mu}(2B_R)\ell(J_1)}{\mu(Q_1)} \approx_{\tau,A} \frac{\Theta_{\mu}(2B_R) \operatorname{diam}(Q_1)}{\mu(2B_{Q_1})} \leqslant b' \quad \text{with } b' = b'(\tau,A) > 0.$$

Furthermore, we define g_k , $k \ge 2$, by induction. Suppose that g_1, \ldots, g_{k-1} have been constructed, satisfy (3.49) and the inequality $\sum_{i=1}^{k-1} g_i \le b$ with some $b = b(\tau, A) > 0$ to be chosen later.

If Q_k is such that $Q_k \cap \bigcup_{i=1}^{k-1} Q_i = \emptyset$, then we set

$$lpha_k = rac{\sigma(\widehat{J}_k)}{\mu(A_k)} \qquad ext{and} \qquad A_k = Q_k \setminus R_{\mathsf{Far}},$$

so that $\int g_k d\mu = \sigma(\widehat{J}_k)$. Moreover, similarly to the case of α_1 , we have

$$||g_k||_{\infty} = \alpha_k \leqslant b',$$

where $b' = b'(\tau, A)$ is obviously independent of k. Since $A_k \cap \bigcup_{i=1}^{k-1} A_i = \emptyset$, we have

$$g_k + \sum_{i=1}^{k-1} g_i \leqslant \max\{b, b'\}.$$

We choose $b = b'(\tau, A)$ in order to have (3.50).

Now suppose that $Q_k \cap \bigcup_{i=1}^{k-1} Q_i \neq \emptyset$ and let Q_{s_1}, \ldots, Q_{s_m} be the subfamily of Q_1, \ldots, Q_{k-1} such that $Q_{s_j} \cap Q_k \neq \emptyset$. Since $\ell(J_{s_j}) \leq \ell(J_k)$ (because of the non-decreasing sizes of $\ell(J_i), i \in I'_0$), we deduce that dist $(J_{s_j}, J_k) \leq \ell(J_k)$, and thus $J_{s_j} \subset c'J_k$, for some constant c' > 0. Using (3.49) for $i = s_j$, we get by (3.15) in Lemma 3.9, Lemmas 3.21 and 3.23 that

$$\begin{split} \sum_{j} \int g_{s_{j}} \, d\mu &= \sum_{j} \sigma(\widehat{J}_{s_{j}}) \leqslant \sigma(\Pi^{-1}(c'J_{k})) \\ &\lesssim \Theta_{\mu}(2B_{R})\ell(J_{k}) \lesssim \Theta_{\mu}(2B_{R}) \operatorname{diam}\left(Q_{k}\right) \leqslant c'' \, \mu(Q_{k}) \end{split}$$

with some $c'' = c''(\tau, A) > 0$. Therefore, by Chebyshev's inequality,

$$\mu\left(\left\{\sum_{j}g_{s_{j}}>2c''\right\}\right)\leqslant\frac{1}{2}\,\mu(Q_{k}).$$

So we set

$$A_k = \left(Q_k \cap \left\{\sum\nolimits_j g_{s_j} \leqslant 2c''\right\}\right) \setminus R_{\mathsf{Far}},$$

and then $\mu(A_k) \ge \frac{1}{4}\mu(Q_k)$. As above, we put $\alpha_k = \sigma(\widehat{J}_k)/\mu(A_k)$ so that $g_k = \alpha_k \chi_{A_k}$ satisfies $\int g_k d\mu = \sigma(\widehat{J}_k)$. Consequently,

$$\alpha_k \leqslant \frac{\sigma(\widehat{J}_k)}{\frac{1}{4}\mu(Q_k)} \leqslant b'' \quad \text{with some } b'' = b''(\tau, A) > 0,$$

which yields

$$g_k + \sum_j g_{s_j} \leqslant b'' + 2c''.$$

Recall that s_j are such that $Q_{s_j} \cap Q_k \neq \emptyset$. The latter inequality implies that

$$g_k + \sum_{i=1}^{k-1} g_i \leq \max\{b, b'' + 2c''\}.$$

In this case, we choose b = b'' + 2c'' and (3.50) follows. Clearly, this bound is independent of the number of functions.

Suppose now that $\{J_i\}_{i \in I'_0}$ is not finite. For each fixed M we consider a family of intervals $\{J_i\}_{1 \leq i \leq M}$. As above, we construct functions g_1^M, \ldots, g_M^M with spt $(g_i^M) \subset Q_i \setminus R_{\mathsf{Far}}$ satisfying

$$\int g_i^M d\mu = \sigma(\widehat{J}_i) \quad \text{and} \quad \sum_{i=1}^M g_i^M \leqslant b = b(\tau, A).$$

Then there is a subsequence $\{g_1^k\}_{k\in I_1}$ which is convergent in the weak * topology of $L^{\infty}(\mu)$ to some function $g_1 \in L^{\infty}(\mu)$. Now we take another convergent subsequence $\{g_2^k\}_{k\in I_2}$, $I_2 \subset I_1$, in the weak * topology of $L^{\infty}(\mu)$ to another function $g_2 \in L^{\infty}(\mu)$, etc. We have spt $(g_i) \subset Q_i \setminus R_{\mathsf{Far}}$. Furthermore, (3.49) and (3.50) also hold due to the weak * convergence.

Recall that $G_R = \{z \in \mathbb{C} : d(z) = 0\}$ (see (3.28)) and clearly $G_R \subset R$. We will need the following result which can be proved analogously to [Tol5, Lemma 7.18] taking into account that the density $\Theta_{\mu}(2B_R)$ is involved in our case. Lemma 3.38. We have

$$\mu \lfloor G_R = \rho_{G_R} \Theta_\mu(2B_R) \mathcal{H}^1_{G_R} = \rho_{G_R} \sigma \lfloor G_R,$$

where ρ_{G_R} is a function such that $c \leq \rho_{G_R} \leq c^{-1}$ with some constant $c = c(\tau, A) > 0$.

Let us mention now the following technical result proved in [Tol5, Subsection 4.6.1].

Lemma 3.39. Let $x, y, z \in \mathbb{C}$ be pairwise distinct points, and let $x' \in \mathbb{C}$ be such that

$$a^{-1}|x-y| \leq |x'-y| \leq a|x-y|,$$

where a > 0 is some constant. Then

$$|c(x, y, z) - c(x', y, z)| \leq (4 + 2a) \frac{|x - x'|}{|x - y||x - z|}$$

Take into account that $p_{\infty}(x, y, z) = \frac{1}{2}c(x, y, z)^2$ by (0.15). Recall that

$$\Gamma_{B_0} = G_R \cup \bigcup_{i \in I_0} \widehat{J}_i$$
 and $\widehat{J}_i = \Gamma_R \cap \Pi^{-1}(J_i).$

In Lemma 3.37 we showed how $\sigma \lfloor \hat{J}_i$ can be approximated by a measure supported on $Q_i \setminus R_{\mathsf{Far}}$, for each $i \in I'_0$, where I'_0 is defined in (3.48). Notice that, by Lemma 3.27,

dist
$$(Q_i, J_i) \lesssim_{\tau, A} \ell(J_i), \quad i \in I_0.$$
 (3.51)

Now we consider the measures

$$\nu_i := g_i \, \mu, \qquad i \in I'_0,$$

with g_i as in Lemma 3.37, and set

$$\nu := \sigma \lfloor G_R + \sum_{i \in I'_0} \nu_i = \rho_{G_R}^{-1} \mu \lfloor G_R + \sum_{i \in I'_0} g_i \mu.$$
(3.52)

This measure should be understood as an approximation of $\sigma = \Theta_{\mu}(2B_R)\mathcal{H}^1_{\Gamma_{B_0}}$, which coincides with σ on G_R due to Lemma 3.38 ($g_i \equiv 0$ in this case).

Using the measure ν , we will actually prove the inequality (3.47) in Lemma 3.36. This will be done in the forthcoming subsection.

3.12.4 Estimates for the permutations of the Hausdorff measure restricted to Γ_R in the case when points are far from each other

To proceed, we need to introduce some additional notation. Given measures τ_1, τ_2, τ_3 , set

$$p_t(\tau_1, \tau_2, \tau_3) := \iiint p_t(x, y, z) \, d\tau_1(x) \, d\tau_2(y) \, d\tau_3(z), \quad \text{where } t = 0 \text{ or } t = \infty.$$

We denote by $p_{t,\mathbf{F}}(\tau_1,\tau_2,\tau_3)$ the triple integral above restricted to (x,y,z) such that

$$|x_1 - y_1| \ge \varepsilon_0^{-1/20} (\ell_x + \ell_y), |x_1 - z_1| \ge \varepsilon_0^{-1/20} (\ell_x + \ell_z), |y_1 - z_1| \ge \varepsilon_0^{-1/20} (\ell_y + \ell_z).$$
(3.53)

So we have

$$p_{\infty,\mathbf{F}}(\sigma) = p_{\infty,\mathbf{F}}(\sigma \lfloor G_R) + p_{\infty,\mathbf{F}}(\sigma \lfloor \Gamma_{B_0} \setminus G_R) + 3 p_{\infty,\mathbf{F}}(\sigma \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R) + 3 p_{\infty,\mathbf{F}}(\sigma \lfloor G_R, \sigma \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R).$$

$$(3.54)$$

1. Consider the term $p_{\infty,\mathbf{F}}(\sigma \lfloor G_R)$. In this case $\ell_x = \ell_y = \ell_z \equiv 0$ and the subscript **F** may be skipped. Moreover, using Lemmas 3.32 and 3.38, we get

$$p_{\infty,\mathbf{F}}(\sigma \lfloor G_R) \lesssim_{\theta_0} p_0(\sigma \lfloor G_R) \approx_{\theta_0,\tau,A} p_0(\mu \lfloor G_R).$$

Now we proceed very similarly to the proof of Lemma 3.12. For $\delta > 0$ from Lemma 3.7 (see also Section 3.7), taking into account Remark 3.7, we get

$$p_{0}(\mu \lfloor G_{R}) \leq \int_{G_{R}} \sum_{Q \in \mathsf{Tree}(R) \setminus \mathsf{Stop}(R): x \in 2B_{Q}} p_{0}^{[\delta,Q]}(x,\mu \lfloor G_{R},\mu \lfloor G_{R}) \ d\mu(x)$$

$$\leq \sum_{Q \in \mathsf{Tree}(R) \setminus \mathsf{Stop}(R)} p_{0}^{[\delta,Q]}(\mu \lfloor 2B_{Q},\mu \lfloor 2B_{R},\mu \lfloor 2B_{R})$$

$$= \sum_{Q \in \mathsf{Tree}(R) \setminus \mathsf{Stop}(R)} \frac{p_{0}^{[\delta,Q]}(\mu \lfloor 2B_{Q},\mu \lfloor 2B_{R},\mu \lfloor 2B_{R})}{\mu(Q)} \int \chi_{Q}(x) \ d\mu(x).$$

Changing the order of summation and the inequality (3.17) yield

$$\begin{split} \frac{p_0(\mu \lfloor G_R)}{\Theta_{\mu}(2B_R)^2} &\leqslant \int_R \sum_{Q \in \mathsf{Tree}(R) \backslash \mathsf{Stop}(R): \, x \in Q} \frac{p_0^{[b,Q]}(\mu \lfloor 2B_Q, \mu \lfloor 2B_R, \mu \lfloor 2B_R)}{\Theta_{\mu}(2B_R)^2 \mu(Q)} \, d\mu(x) \\ &= \int_R \sum_{Q \in \mathsf{Tree}(R) \backslash \mathsf{Stop}(R): \, x \in Q} \mathsf{perm}(Q)^2 \, d\mu(x). \end{split}$$

From this and the inequality (3.17) in Lemma 3.9 we deduce that

$$p_0(\mu \lfloor G_R) \leqslant \alpha^2 \Theta_\mu (2B_R)^2 \mu(R).$$

Finally, if $\alpha = \alpha(\theta_0, \varepsilon_0, \tau, A)$ is chosen small enough, then

$$p_{\infty,\mathbf{F}}(\sigma \lfloor G_R) \lesssim \varepsilon_0^{1/40} \Theta_\mu (2B_R)^2 \mu(R).$$

2. Let us study $p_{\infty,\mathbf{F}}(\sigma \lfloor \Gamma_{B_0} \setminus G_R)$. In this case ℓ_x , ℓ_y and ℓ_z are strictly positive and so are the lengths of the associated doubling cubes from Lemma 3.21. We set

$$p_{\infty,\mathbf{F}}(\sigma \lfloor \Gamma_{B_0} \setminus G_R) = \sum_{i,j,k \in I_0} p_{\infty,\mathbf{F}} \left(\sigma \lfloor \widehat{J}_i, \sigma \lfloor \widehat{J}_j, \sigma \lfloor \widehat{J}_k \right).$$

First let us consider the case when at least one of the indices i, j or k is in $I_0 \setminus I'_0$, i.e. $\mu(Q_h \cap R_{\mathsf{Far}}) > \frac{3}{4}\mu(Q_h)$ for h being i, j or k, according to (3.48). By symmetry, we may consider just the case $i \in I_0 \setminus I'_0$. Moreover, then the required estimate will follow from a proper one for

$$p_{\infty}\left(\sigma\lfloor\widehat{J'},\sigma\lfloor\Gamma_{B_{0}},\sigma\lfloor\Gamma_{B_{0}}
ight), \quad \text{where } \widehat{J'}:=igcup_{i\in I_{0}\setminus I'_{0}}\widehat{J_{i}}.$$

Recall that

$$\sigma = \Theta_{\mu}(2B_R)\mathcal{H}^1_{\Gamma_{B_0}} \quad \text{and} \quad \Gamma_{B_0} = G_R \cup \bigcup_{i \in I_0} \widehat{J}_i.$$

Lemma 3.40. We have

$$\mathcal{H}^1(\widehat{J}') \leqslant \sqrt{\alpha} \operatorname{diam}(R)$$

Proof. Notice that for $i \in I_0 \setminus I'_0$ we have

$$\Theta_{\mu}(2B_R)\mathcal{H}^1(\widehat{J}_i) \lesssim \Theta_{\mu}(2B_R)\ell(J_i) \lesssim_{\tau,A} \Theta_{\mu}(2B_R) \operatorname{diam}\left(Q_i\right) \approx_{\tau,A} \mu(Q_i) \lesssim_{\tau,A} \mu(Q_i \cap R_{\mathsf{Far}}),$$

where $Q_i \in \mathsf{DbTree}(R)$ is the cube associated to the interval J_i by Lemma 3.21. By Vitali's covering lemma, there exists a subfamily of balls $2B_{Q_i}$, $i \in J \subset I_0 \setminus I'_0$, such that

- the balls $2B_{Q_i}$, $i \in J$, are disjoint,
- $\bigcup_{i \in I_0 \setminus I'_0} 2B_{Q_i} \subset \bigcup_{i \in J} 10B_{Q_i}$.

Then, taking into account that $\mu(10B_{Q_i} \cap R) \approx_{\tau,A} \mu(2B_{Q_i}) \approx \mu(Q_i)$, we get

$$\begin{split} \Theta_{\mu}(2B_R)\mathcal{H}^1(\widehat{J'}) \lesssim \sum_{i \in I_0 \setminus I'_0} \mu(Q_i) \lesssim \sum_{i \in J} \mu(10B_{Q_i} \cap R) \\ \lesssim_{\tau,A} \sum_{i \in J} \mu(Q_i) \lesssim_{\tau,A} \sum_{i \in J} \mu(Q_i \cap R_{\mathsf{Far}}) \lesssim_{\tau,A} \mu(R_{\mathsf{Far}}), \end{split}$$

because the cubes Q_i from the family J are disjoint. Since $\mu(R_{\mathsf{Far}}) \leq \alpha \mu(R)$ by Lemma 3.12, the lemma follows if $\alpha = \alpha(\tau, A)$ is chosen small enough.

To continue, we need the following result from [Tol5].

Lemma 3.41 (Lemma 3.4 in [Tol5]). Let μ_1 , μ_2 and μ_3 be finite measures. Then

$$\sum_{s\in\mathfrak{S}_3}\int \mathcal{C}_{\varepsilon}(\mu_{s_2})\overline{\mathcal{C}_{\varepsilon}(\mu_{s_3})}\,d\mu_{s_1} = c_{\varepsilon}^2(\mu_1,\mu_2,\mu_3) + \mathcal{R}, \qquad \mathcal{R}\leqslant C \sum_{s\in\mathfrak{S}_3}\int M_{\mathsf{R}}\mu_{s_2}M_{\mathsf{R}}\mu_{s_3}\,d\mu_{s_1},$$

where \mathfrak{S}_3 is the group of permutations of the three elements $\{1, 2, 3\}$, $\mathcal{C}_{\varepsilon}$ the truncated Cauchy integral, c_{ε}^2 the truncated curvature of measure (see (0.6) and below) and M_{R} the 1-dimensional radial maximal operator.

Lemma 3.42. For $E \subset \Gamma_{B_0}$, we have

$$c^2(\mathcal{H}^1_E, \mathcal{H}^1_{\Gamma_{B_0}}, \mathcal{H}^1_{\Gamma_{B_0}}) \lesssim \mathcal{H}^1(E)^{1/2} \operatorname{diam}(R)^{1/2}$$

Proof. By Lemma 3.41, we have

$$\begin{split} c^2(\mathcal{H}_E^1,\mathcal{H}_{\Gamma_{B_0}}^1,\mathcal{H}_{\Gamma_{B_0}}^1) \lesssim \limsup_{\varepsilon \to 0} \int_{\Gamma_{B_0}} |\mathcal{C}_{\varepsilon}(\mathcal{H}_E^1) \, \mathcal{C}_{\varepsilon}(\mathcal{H}_{\Gamma_{B_0}}^1)| \, d\mathcal{H}^1 \\ &+ \limsup_{\varepsilon \to 0} \int_E |\mathcal{C}_{\varepsilon}(\mathcal{H}_{\Gamma_{B_0}}^1)|^2 \, d\mathcal{H}^1 \\ &+ \int_{\Gamma_{B_0}} |M_{\mathsf{R}}(\mathcal{H}_E^1) \, M_{\mathsf{R}}(\mathcal{H}_{\Gamma_{B_0}}^1)|^2 \, d\mathcal{H}^1 \\ &+ \int_E |M_{\mathsf{R}}(\mathcal{H}_{\Gamma_{B_0}}^1)|^2 \, d\mathcal{H}^1 \\ &:= I_1 + I_2 + I_3 + I_4. \end{split}$$

Regarding I_1 , by the L^2 -boundedness of the Cauchy transform on Lipschitz graphs (with respect to $\mathcal{H}^1_{\Gamma_B}$) we have

$$I_{1} \leq \limsup_{\varepsilon \to 0} \| \mathcal{C}_{\varepsilon}(\mathcal{H}_{E}^{1}) \|_{L^{2}(\mathcal{H}_{\Gamma_{R}}^{1})} \| \mathcal{C}_{\varepsilon}(\mathcal{H}_{\Gamma_{B_{0}}}^{1}) \|_{L^{2}(\mathcal{H}_{\Gamma_{R}}^{1})}$$
$$\lesssim \mathcal{H}^{1}(E)^{1/2} \mathcal{H}^{1}(\Gamma_{B_{0}})^{1/2}$$
$$\lesssim \mathcal{H}^{1}(E)^{1/2} \operatorname{diam}(R)^{1/2}.$$

For I_2 we use the L^4 -boundedness of the Cauchy transform:

$$I_2 \leqslant \limsup_{\varepsilon \to 0} \mathcal{H}^1(E)^{1/2} \| \mathcal{C}_{\varepsilon}(\mathcal{H}^1_{\Gamma_{B_0}}) \|_{L^4(\mathcal{H}^1_{\Gamma_R})}^2 \lesssim \mathcal{H}^1(E)^{1/2} \operatorname{diam}(R)^{1/2}.$$

Using the fact that $M_{\mathsf{R}}(\mathcal{H}^1_{\Gamma_{B_0}}) \lesssim 1$, we derive

$$I_4 \leqslant \mathcal{H}^1(E) \lesssim \mathcal{H}^1(E)^{1/2} \operatorname{diam}(R)^{1/2},$$

and also

$$I_3 \lesssim \int_{\Gamma_{B_0}} |M_{\mathsf{R}}(\mathcal{H}_E^1)| \, d\mathcal{H}^1.$$

Since the operator $M_{\mathsf{R}}(\mathcal{H}_{\Gamma_R}^1)$ is bounded in $L^2(\mathcal{H}_{\Gamma_R}^1)$ (as it is comparable to the Hardy-Littlewood operator with respect to the measure $\mathcal{H}_{\Gamma_R}^1$), we deduce

$$I_3 \lesssim \|M_{\mathsf{R}}(\chi_E \mathcal{H}_{\Gamma_R}^1)\|_{L^2(\mathcal{H}_{\Gamma_R}^1)} \mathcal{H}^1(\Gamma_{B_0})^{1/2} \lesssim \mathcal{H}^1(E)^{1/2} \operatorname{diam}(R)^{1/2}.$$

So the lemma follows.

By Lemma 3.42 for $E = \hat{J'}$ and Lemma 3.40 we derive that

$$c^{2}(\mathcal{H}_{\widehat{J}'}^{1},\mathcal{H}_{\Gamma_{B_{0}}}^{1},\mathcal{H}_{\Gamma_{B_{0}}}^{1}) \lesssim \mathcal{H}^{1}(\widehat{J}')^{1/2} \operatorname{diam}(R)^{1/2} \lesssim \alpha^{1/4} \operatorname{diam}(R).$$

Therefore, recalling that $p_{\infty}(x, y, z) = \frac{1}{2}c(x, y, z)^2$ (see (0.15)),

$$p_{\infty}\left(\sigma\lfloor\widehat{J'},\sigma\lfloor\Gamma_{B_{0}},\sigma\lfloor\Gamma_{B_{0}}\right)\lesssim\alpha^{1/4}\Theta_{\mu}(2B_{R})^{3}\operatorname{diam}\left(R\right)\approx\alpha^{1/4}\Theta_{\mu}(2B_{R})^{2}\mu(R).$$

Furthermore, choosing $\alpha = \alpha(\varepsilon_0)$ small enough, we get from the latter estimate that

$$\sum_{i \in I_0 \setminus I'_0, j,k \in I_0} p_{\infty,\mathbf{F}} \left(\sigma \lfloor \widehat{J}_i, \sigma \lfloor \widehat{J}_j, \sigma \lfloor \widehat{J}_k \right) \lesssim \varepsilon_0^{1/40} \Theta_\mu (2B_R)^2 \,\mu(R), \tag{3.55}$$

and we are done with the case when at least one of the indices i, j or k is in $I_0 \setminus I'_0$.

Now let $(i, j, k) \in (I'_0)^3$. By definition, if $p_{\infty, \mathbf{F}}\left(\sigma \lfloor \widehat{J}_i, \sigma \lfloor \widehat{J}_j, \sigma \lfloor \widehat{J}_k\right) \neq 0$, then there exist $x \in \widehat{J}_i, y \in \widehat{J}_j$ and $z \in \widehat{J}_k$ satisfying (3.53). Then it follows easily that

$$dist \left(\widehat{J}_{i}, \widehat{J}_{j}\right) \geq \frac{1}{2} \varepsilon_{0}^{-1/20} \left(\ell(J_{i}) + \ell(J_{j})\right), dist \left(\widehat{J}_{i}, \widehat{J}_{k}\right) \geq \frac{1}{2} \varepsilon_{0}^{-1/20} \left(\ell(J_{i}) + \ell(J_{k})\right), dist \left(\widehat{J}_{j}, \widehat{J}_{k}\right) \geq \frac{1}{2} \varepsilon_{0}^{-1/20} \left(\ell(J_{j}) + \ell(J_{k})\right).$$

$$(3.56)$$

We denote by $J_{\mathbf{F}}$ the set of those indices $(i, j, k) \in (I'_0)^3$ such that the inequalities (3.56) hold, so that

$$p_{\infty,\mathbf{F}}(\sigma \lfloor \Gamma_{B_0} \setminus G_R) \leqslant \sum_{(i,j,k) \in J_{\mathbf{F}}} p_{\infty} \left(\sigma \lfloor \widehat{J}_i, \sigma \lfloor \widehat{J}_j, \sigma \lfloor \widehat{J}_k \right).$$

Consider $(i, j, k) \in J_{\mathbf{F}}$ and

$$x, x' \in \widehat{J}_i \cup Q_i, \qquad y, y' \in \widehat{J}_j \cup Q_j \qquad \text{and} \qquad z, z' \in \widehat{J}_k \cup Q_k.$$

Due to (3.56) and (3.51), taking into account that $\ell(J_h) \approx_{\tau,A} \operatorname{diam}(\widehat{J}_h) \approx_{\tau,A} \operatorname{diam}(Q_h)$ for each $h \in I$ by Lemma 3.21, the sets $\widehat{J}_i \cup Q_i$, $\widehat{J}_j \cup Q_j$ and $\widehat{J}_k \cup Q_k$ are far to each other in the sense that

dist
$$(\widehat{J}_i \cup Q_i, \widehat{J}_j \cup Q_j) \gtrsim \varepsilon_0^{-1/20} (\ell(J_i) + \ell(J_j)),$$

dist $(\widehat{J}_i \cup Q_i, \widehat{J}_k \cup Q_k) \gtrsim \varepsilon_0^{-1/20} (\ell(J_i) + \ell(J_k)),$
dist $(\widehat{J}_j \cup Q_j, \widehat{J}_k \cup Q_k) \gtrsim \varepsilon_0^{-1/20} (\ell(J_j) + \ell(J_k)),$
(3.57)

where ε_0 is chosen small enough. Furthermore, applying Lemma 3.39 three times gives

$$p_{\infty}(x, y, z) \leq 2 p_{\infty}(x', y', z') + c (T_x(y, z) + T_y(x, z) + T_z(x, y)),$$

where

$$T_{z_1}(z_2, z_3) := \frac{\ell_{z_1}^2}{|z_1 - z_2|^2 |z_1 - z_3|^2} \quad \text{for} \quad z_1, z_2, z_3 \in \mathbb{C}.$$

Then, integrating on $x \in \widehat{J}_i$, $y \in \widehat{J}_j$, and $z \in \widehat{J}_k$ with respect to σ , we get

$$p_{\infty}\left(\sigma\lfloor\widehat{J}_{i},\sigma\lfloor\widehat{J}_{j},\sigma\lfloor\widehat{J}_{k}\right) \leqslant 2 p_{\infty}(x',y',z') \sigma(\widehat{J}_{i}) \sigma(\widehat{J}_{j}) \sigma(\widehat{J}_{k}) + c \iiint_{\substack{x\in\widehat{J}_{i}\\y\in\widehat{J}_{j}\\z\in\widehat{J}_{k}}} [T_{x}(y,z) + T_{y}(x,z) + T_{z}(x,y)] \ d\sigma(x) \ d\sigma(y) \ d\sigma(z).$$

On the other hand, by analogous arguments, we have

$$p_{\infty}(x',y',z') \|\nu_{i}\| \|\nu_{j}\| \|\nu_{k}\| \leq 2 p_{\infty}(\nu_{i},\nu_{j},\nu_{k}) + c \iiint [T_{x}(y,z) + T_{y}(x,z) + T_{z}(x,y)] d\nu_{i}(x) d\nu_{j}(y) d\nu_{k}(z).$$

Thus, recalling that $\|\nu_h\| = \sigma(\widehat{J}_h)$ for any $h \in I'_0$, from the preceding inequalities we get

$$p_{\infty}\left(\sigma \lfloor \widehat{J}_{i}, \sigma \lfloor \widehat{J}_{j}, \sigma \lfloor \widehat{J}_{k}\right) \lesssim p_{\infty}(\nu_{i}, \nu_{j}, \nu_{k}) + \iiint [T_{x}(y, z) + T_{y}(x, z) + T_{z}(x, y)] d\nu_{i}(x) d\nu_{j}(y) d\nu_{k}(z) + \iiint \sum_{\substack{x \in \widehat{J}_{i} \\ y \in \widehat{J}_{j} \\ z \in \widehat{J}_{k}}} [T_{x}(y, z) + T_{y}(x, z) + T_{z}(x, y)] d\sigma(x) d\sigma(y) d\sigma(z).$$

$$(3.58)$$

Now recall that $A_h = \operatorname{spt} \nu_h \subset Q_h$ for any $h \in I'_0$. This and Lemma 3.24 imply that for each $x \in Q_i$ and $y \in Q_j$ there exist $\tilde{x} \in \Gamma_R$ and $\tilde{y} \in \Gamma_R$, correspondingly, such that $\operatorname{dist}(x, \tilde{x}) \leq_{\tau,A} \ell(J_i)$ and $\operatorname{dist}(y, \tilde{y}) \leq_{\tau,A} \ell(J_j)$. Due to this fact and (3.57), it holds that

$$\measuredangle(L_{x\tilde{y}}, L_{xy}) \lesssim \frac{|y - \tilde{y}|}{|x - y|} \lesssim_{\tau, A} \frac{\ell(J_j)}{\varepsilon_0^{-1/20} \left(\ell(J_i) + \ell(J_j)\right)} \lesssim_{\tau, A} \varepsilon_0^{1/20}$$

and

$$\measuredangle(L_{\tilde{x}y}, L_{xy}) \lesssim \frac{|x - \tilde{x}|}{|x - y|} \lesssim_{\tau, A} \frac{\ell(J_i)}{\varepsilon_0^{-1/20} \left(\ell(J_i) + \ell(J_j)\right)} \lesssim_{\tau, A} \varepsilon_0^{1/20}$$

So it follows that $\measuredangle(L_{\tilde{x}\tilde{y}}, L_{xy}) \lesssim_{\tau,A} \varepsilon_0^{1/20}$. By Lemma 3.23 and the definitions at the beginning of Subsection 3.12.1,

$$\measuredangle(L_{xy}, L_R) \leqslant C_F \theta_0$$
 and $\theta_V(L_R) \geqslant (1 + C_F) \theta_0$.

Consequently,

$$\theta_V(L_{\tilde{x}\tilde{y}}) \ge \theta_V(L_R) - \measuredangle(L_{xy}, L_R) - \measuredangle(L_{\tilde{x}\tilde{y}}, L_{xy}) \ge \frac{1}{2}\theta_0$$

if $\varepsilon_0 = \varepsilon_0(\theta_0, \tau, A)$ is chosen small enough. Now use Lemma 3.6 to conclude that

$$p_{\infty}(\nu_i, \nu_j, \nu_k) \lesssim_{\theta_0} p_0(\nu_i, \nu_j, \nu_k).$$

$$(3.59)$$

Moreover, from (3.57) and the fact that $\ell(J_h) \approx_{\tau,A} \operatorname{diam}(Q_h)$ for any h we conclude that

$$\begin{split} p_0(\nu_i,\nu_j,\nu_k) \lesssim &\int \sum_{\substack{Q \in \mathsf{Tree}(R) \setminus \mathsf{Stop}(R): \, x \in 2B_Q \\ }} p_0^{[\delta,Q]}(x,\nu_j,\nu_k) \, d\nu_i(x) \\ \lesssim &\sum_{\substack{Q \in \mathsf{Tree}(R) \setminus \mathsf{Stop}(R)}} p_0^{[\delta,Q]}(\nu_i \lfloor 2B_Q,\nu_j,\nu_k), \end{split}$$

where $\delta = \delta(\varepsilon_0, \tau, A)$ is chosen small enough. Furthermore, using that $\nu = g\mu$ and arguing as in the case of $p_{\infty, \mathbf{F}}(\sigma \lfloor G_R)$ we get

$$\begin{split} \sum_{(i,j,k)\in J_{\mathbf{F}}} p_0(\nu_i,\nu_j,\nu_k) \lesssim \sum_{\substack{Q\in\mathsf{Tree}(R)\backslash\mathsf{Stop}(R)}} p_0^{[\delta,Q]}(\nu\lfloor 2B_Q,\nu,\nu) \\ \lesssim_{\tau,A} \sum_{\substack{Q\in\mathsf{Tree}(R)\backslash\mathsf{Stop}(R)}} p_0^{[\delta,Q]}(\mu\lfloor 2B_Q,\mu\lfloor 2B_R,\mu\lfloor 2B_R) \\ \lesssim_{\tau,A} \alpha^2 \Theta_{\mu}(2B_R)^2 \mu(R) \\ \lesssim \varepsilon_0^{1/20} \Theta_{\mu}(2B_R)^2 \mu(R), \end{split}$$

where $\alpha = \alpha(\theta_0, \varepsilon_0, \tau, A)$ is chosen small enough. From this, (3.58) and (3.59) by summing on $(i, j, k) \in J_{\mathbf{F}}$ we deduce that

$$\sum_{\substack{(i,j,k)\in J_{\mathbf{F}}\\ (i,j,k)\in J_{\mathbf{F}}}} p_{\infty} \left(\sigma \lfloor \hat{J}_{i}, \sigma \lfloor \hat{J}_{j}, \sigma \lfloor \hat{J}_{k}\right) \\ \lesssim \varepsilon_{0}^{1/40} \Theta_{\mu}(2B_{R})^{2} \mu(R) \\ + \iiint_{\substack{|x-y| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}(\ell_{x}+\ell_{y})\\ |y-z| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}(\ell_{x}+\ell_{z})}} [T_{x}(y,z) + T_{y}(x,z) + T_{z}(x,y)] \, d\sigma(x) \, d\sigma(y) \, d\sigma(z) \\ |y-z| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}(\ell_{x}+\ell_{z})\\ + \iiint_{\substack{|x-y| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}(\ell_{x}+\ell_{y})\\ |x-z| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}(\ell_{x}+\ell_{z})}} [T_{x}(y,z) + T_{y}(x,z) + T_{z}(x,y)] \, d\nu(x) \, d\nu(y) \, d\nu(z), \end{cases}$$
(3.60)

where $\varepsilon_0 = \varepsilon_0(\theta_0)$ was chosen small enough. Recall the definition of ν in (3.52).

To estimate the first triple integral in the right side of (3.60), notice that

$$\begin{aligned}
\iint_{\substack{|x-y| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}(\ell_{x}+\ell_{y}) \\ |x-z| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}(\ell_{x}+\ell_{z})}} T_{x}(y,z) \, d\sigma(y) \, d\sigma(z) \\
\leqslant \left(\int_{\substack{|x-y| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}\ell_{x}}} \frac{\ell_{x}}{|x-y|^{2}} \, d\sigma(y) \right) \left(\int_{\substack{|x-z| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}\ell_{x}}} \frac{\ell_{x}}{|x-z|^{2}} \, d\sigma(z) \right) \quad (3.61) \\
= \left(\int_{\substack{|x-y| \ge \frac{1}{2}\varepsilon_{0}^{-1/20}\ell_{x}}} \frac{\ell_{x}}{|x-y|^{2}} \, d\sigma(y) \right)^{2} \lesssim \varepsilon_{0}^{1/10} \, \Theta_{\mu}(2B_{R})^{2},
\end{aligned}$$

where the last inequality follows from splitting the domain $\{y : |x - y| \ge \frac{1}{2}\varepsilon_0^{-1/20}\ell_x\}$ into annuli and the linear growth of σ with constant $\lesssim \Theta_{\mu}(2B_R)$ (see (0.12)). Analogous estimates hold permuting x, y, z, and also interchanging σ by ν (the implicit constant in the analogue of (3.61) for ν depends on τ and A then). Indeed, this is a consequence of the following result.

Lemma 3.43. It holds that

 $\nu(B(x,r)) \lesssim_{\tau,A} \Theta_{\mu}(2B_R) r$, where $r \ge \ell_x > 0$ and $x \in \operatorname{spt} \nu \subset R \setminus R_{\mathsf{Far}}$.

Proof. Recall that $\nu = g\mu$ with g bounded by a constant depending in τ and A, see (3.52). If r > diam R, then $\text{spt } \nu \subset B(x, r)$ and thus

$$\nu(B(x,r)) \lesssim_{\tau,A} \mu(2B_R) \approx_{\tau,A} \Theta_{\mu}(2B_R) \operatorname{diam}(R) \lesssim_{\tau,A} \Theta_{\mu}(2B_R) r.$$

Consequently, we may suppose below that $\ell_x \leq r \leq \operatorname{diam}(R)$.

First let $d(x) \leq C(\tau, A)\ell_x$, where $C(\tau, A) > 0$ will be chosen later. Then there should exist $P \in \mathsf{DbTree}(R)$ such that $B(x, r) \subset 2B_P$ and diam $(P) \approx_{\tau, A} r$ so that

$$\nu(B(x,r)) \lesssim_{\tau,A} \mu(B(x,r)) \lesssim_{\tau,A} \mu(2B_P) \approx_{\tau,A} \Theta_{\mu}(2B_R) \operatorname{diam}(P) \approx_{\tau,A} \Theta_{\mu}(2B_R)r.$$

Now let $d(x) \ge C(\tau, A)\ell_x > 0$. Set $y = (\Pi(x), F(\Pi(x))) \in \Gamma_R$. As shown in the proof of Lemma 3.24, $d(y) \le c(\tau, A)\ell_x$ with some $c(\tau, A) > 0$. Choose $Q' \in \mathsf{DbTree}(R)$ so that

$$\operatorname{dist}(y, Q') + \operatorname{diam}(Q') \leq 2d(y).$$

Taking into account that $x \in R \setminus R_{\mathsf{Far}}$, from Lemma 3.26 and the properties of Γ_R we deduce that dist $(x, y) \lesssim \sqrt[4]{\varepsilon_0} d(x) \leqslant \sqrt[8]{\varepsilon_0} d(x)$ if ε_0 is chosen small enough. Thus

$$d(x) \leqslant \operatorname{dist} (x, Q') + \operatorname{diam} (Q') \leqslant \operatorname{dist} (x, y) + 2d(y) \leqslant \sqrt[8]{\varepsilon_0} d(x) + 2c(\tau, A)\ell_x$$
$$\leqslant \sqrt[8]{\varepsilon_0} d(x) + \frac{2c(\tau, A)}{C(\tau, A)} d(x) \leqslant (\sqrt[8]{\varepsilon_0} + \frac{1}{2})d(x) < d(x),$$

if we choose $C(\tau, A) \ge 4c(\tau, A)$. Hence we get a contradiction if $d(x) \ge C(\tau, A)\ell_x > 0$. \Box

By plugging the estimates obtained into (3.60), choosing $\varepsilon_0 = \varepsilon_0(\tau, A)$ small enough and recalling (3.55) we get

$$p_{\infty,\mathbf{F}}(\sigma \lfloor \Gamma_{B_0} \setminus G_R) \lesssim \varepsilon_0^{1/40} \Theta_\mu (2B_R)^2 \mu(R).$$

Now it remains to estimate the last two terms of (3.54). The arguments are similar to the preceding ones.

3. Since $\sigma \lfloor G_R = \nu \lfloor G_R$, we have

$$p_{\infty,\mathbf{F}}(\sigma \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R) = p_{\infty,\mathbf{F}}(\nu \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R))$$

and

$$p_{\infty,\mathbf{F}}(\sigma \lfloor G_R, \sigma \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R) = p_{\infty,\mathbf{F}}(\nu \lfloor G_R, \nu \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R)$$

Concerning the term $p_{\infty,\mathbf{F}}(\sigma \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R)$, the main difference with respect to the estimates above for $p_{\infty,\mathbf{F}}(\sigma \lfloor \Gamma_{B_0} \setminus G_R)$ is that $T_x(y,z)$ equals zero in this case, and instead of integrating over $\sigma \lfloor \widehat{J}_i$ and ν_i and then summing on i, one integrates over $\sigma \lfloor G_R$. Then one obtains

$$\begin{split} p_{\infty,\mathbf{F}}(\sigma \lfloor G_{R}, \sigma \lfloor \Gamma_{B_{0}} \setminus G_{R}, \sigma \lfloor \Gamma_{B_{0}} \setminus G_{R}) \\ &\lesssim \varepsilon_{0}^{1/40} \Theta_{\mu}(2B_{R})^{2} \,\mu(R) \\ &+ \iiint_{|x-y| \geq \frac{1}{2}\varepsilon_{0}^{-1/20}\ell_{y}} [T_{y}(x,z) + T_{z}(x,y)] \, d\sigma(x) \, d\sigma(y) \, d\sigma(z) \\ &|x-z| \geq \frac{1}{2}\varepsilon_{0}^{-1/20}\ell_{z} \\ &|y-z| \geq \frac{1}{2}\varepsilon_{0}^{-1/20}\ell_{y} + \ell_{z}) \\ &+ \iiint_{|x-y| \geq \frac{1}{2}\varepsilon_{0}^{-1/20}\ell_{y}} [T_{y}(x,z) + T_{z}(x,y)] \, d\nu(x) \, d\nu(y) \, d\nu(z). \\ &|x-z| \geq \frac{1}{2}\varepsilon_{0}^{-1/20}\ell_{z} \\ &|y-z| \geq \frac{1}{2}\varepsilon_{0}^{-1/20}(\ell_{y} + \ell_{z}) \end{split}$$

The last two triple integrals are estimated as in (3.61), and then it follows that

$$p_{\infty,\mathbf{F}}(\sigma \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R) \lesssim \varepsilon_0^{1/40} \,\Theta_\mu(2B_R)^2 \,\mu(R).$$

4. Finally, the arguments for $p_{\infty,\mathbf{F}}(\sigma \lfloor G_R, \sigma \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R)$ are very similar. In this case, both terms $T_x(y,z)$ and $T_y(x,z)$ vanish, and analogously we also get

$$p_{\infty,\mathbf{F}}(\sigma \lfloor G_R, \sigma \lfloor G_R, \sigma \lfloor \Gamma_{B_0} \setminus G_R) \lesssim \varepsilon_0^{1/40} \Theta_\mu(2B_R)^2 \,\mu(R).$$

This finishes the proof of Lemma 3.36.

3.13 The packing condition for Top cubes and the end of the proof of Main Lemma

3.13.1 Properties of the trees

In order to prove the packing condition for Top cubes we will first extract some necessary results from Lemmas 3.9, 3.14, 3.29, 3.30 and 3.38. We suppose that all the parameters and thresholds from Section 3.7 are chosen properly. Recall also the definition (3.28) of G_R .

Lemma 3.44. Let μ be a finite measure with compact support such that

$$p_0(\mu) < \infty.$$

Considering the David-Mattila dyadic lattice \mathcal{D} associated with μ , let $R \in \mathcal{D}^{db}$. Then there exists a $C_F \theta_0$ -Lipschitz function $F : L_R \to L_R^{\perp}$, where $C_F > 0$ is independent of R, a family of pairwise disjoint cubes $\operatorname{Stop}(R) \subset \mathcal{D}(R)$ and a set $G_R \subset R$ such that

(a) G_R is contained in $\Gamma_R = F(L_R)$ and moreover $\mu \lfloor G_R$ is absolutely continuous with respect to $\Theta_\mu(2B_R)\mathcal{H}^1_{\Gamma_R}$;

(b) for any $Q \in \mathsf{Tree}(R)$,

$$\Theta_{\mu}(2B_Q) \lesssim A \Theta_{\mu}(2B_R);$$

(c) if $R \in \mathsf{T}_{VF}(\theta_0)$, then

if $R \notin \mathsf{T}_{VF}(\theta_0)$, then

$$\sum_{\substack{Q \in \operatorname{Stop}(R) \\ Q \notin \operatorname{HD}(R) \cup \operatorname{UB}(R) \cup \operatorname{US}(R)}} \mu(Q) \leqslant \sqrt{\tau} \, \mu(R) + \frac{1}{\alpha^2 \Theta_{\mu}(2B_R)^2} \sum_{\tilde{Q} \in \operatorname{Tree}(R)} p_0^{[\delta, \tilde{Q}]}(\mu \lfloor 2B_{\tilde{Q}}, \mu \lfloor 2B_R, \mu \lfloor 2B_R)).$$

3.13.2 New families of stopping cubes

According to Section 3.8 and Lemma 3.44, each $R \in \mathcal{D}^{db}$ generates several families of cubes fulfilling certain properties. In this subsection we will introduce some variants of these families. The idea is to have stopping cubes that are always different from R and are in \mathcal{D}^{db} , cf. Remark 3.5.

Recall that each cube in HD(R) is in \mathcal{D}^{db} and is clearly different from R due to the fact that $Q \in HD(R)$ satisfies $\Theta_{\mu}(2B_Q) > A \Theta_{\mu}(2B_R)$ with $A \gg 1$.

Now we turn our attention to the family $\mathsf{UB}(R)$. By Lemma 3.5, if $Q \in \mathsf{UB}(R)$, i.e. it is γ -unbalanced, there exists a family of pairwise disjoint cubes $\{P\}_{P \in I_Q} \subset \mathcal{D}^{db}(Q)$ such that diam $(P) \gtrsim \gamma \operatorname{diam}(Q)$ and $\Theta_{\mu}(2B_P) \gtrsim \gamma^{-1} \Theta_{\mu}(2B_Q)$ for each $P \in I_Q$, and

$$\sum_{P \in I_Q} \Theta_{\mu}(2B_P)^2 \,\mu(P) \gtrsim \gamma^{-2} \,\Theta_{\mu}(2B_Q)^2 \,\mu(Q). \tag{3.62}$$

Let I'_Q be a family of (not necessarily doubling) cubes contained in Q, with side length comparable to $a \operatorname{diam}(Q)$ with some a > 0, disjoint from the ones from I_Q , so that

$$Q = \bigcup_{P \in I_Q} P \cup \bigcup_{P \in I'_Q} P.$$

To continue, we introduce additional notation. Given a cube $Q \in \mathcal{D}$, we denote by $\mathcal{MD}(Q)$ the family of maximal cubes (with respect to inclusion) from $\mathcal{D}^{db}(Q)$. By Lemma 3.2, this family covers μ -almost all Q. Furthermore, using the definition just given, we denote by \widetilde{I}_Q the family $\bigcup_{P \in I'_Q} \mathcal{MD}(P)$. Moreover, we set

$$\widetilde{\mathsf{UB}}(R) = \bigcup_{Q \in \mathsf{UB}(R)} (I_Q \cup \widetilde{I}_Q).$$

One can deduce from (3.62) that $R \notin UB(R)$ for a and γ small enough.

Now consider $\mathsf{BS}(R)$. Each cube in this family is in \mathcal{D}^{db} by construction. Moreover, $R \notin \mathsf{BS}(R)$ due to the condition $\measuredangle(L_Q, L_R) > \theta(R) > 0$ for each $Q \in \mathsf{BS}(R)$.

To continue, we set

$$\mathsf{O}(R) = \mathsf{Stop}(R) \setminus (\mathsf{HD}(R) \cup \mathsf{UB}(R) \cup \mathsf{BS}(R)) = \mathsf{LD}(R) \cup \mathsf{BP}(R) \cup \mathsf{F}(R)$$

and

$$\widetilde{\mathsf{O}}(R) = \left\{ \bigcup_{Q \in \mathcal{D}} \mathcal{MD}(Q) : Q \text{ is a son of some cube from } \mathsf{O}(R) \right\}.$$

This guarantees that $R \notin \tilde{O}(R)$ as cubes in $\tilde{O}(R)$ are descendants of cubes in Tree(R).

Finally, let

$$\mathsf{Next}(R) = \mathsf{HD}(R) \cup \widetilde{\mathsf{UB}}(R) \cup \widetilde{\mathsf{O}}(R) \cup \mathsf{BS}(R).$$

By construction, all cubes in Next(R) are disjoint, doubling and different from R. Moreover,

$$R \setminus \bigcup_{Q \in \mathsf{Next}(R)} Q = R \setminus \bigcup_{Q \in \mathsf{Stop}(R)} Q.$$
(3.63)

Using the small boundaries property of the David-Mattila lattice and the definition (3.28), one can also show that

$$\mu\left(R \setminus \bigcup_{Q \in \mathsf{Stop}(R)} Q\right) = \mu(G_R). \tag{3.64}$$

For the record, notice also that, by construction, if $P \in Next(R)$, then

$$\Theta_{\mu}(2B_S) \lesssim_{\tau,A} \Theta_{\mu}(2B_R)$$
 for all $S \in \mathcal{D}$ such that $P \subset S \subset R$. (3.65)

3.13.3 The corona decomposition

Recall that we assumed that μ has compact support. Let

$$R_0 := \operatorname{spt} \mu.$$

Obviously we may suppose that $R_0 \in \mathcal{D}^{db}$. We will construct the family Top contained in R_0 inductively applying Lemma 3.44 so that $\mathsf{Top} = \bigcup_{k \ge 0} \mathsf{Top}_k$. Let

$$\mathsf{Top}_0 = \{R_0\}.$$

Assuming Top_k to be defined, we set

$$\mathsf{Top}_{k+1} = \bigcup_{R \in \mathsf{Top}_k} \mathsf{Next}(R).$$

Note that cubes in Next(R), with $R \in Top_k$, are pairwise disjoint.

3.13.4 The families of cubes ID_H , ID_U and ID

We distinguish two types of cubes $R \in \mathsf{Top}$. We write $R \in ID_H$ (increasing density because of high density cubes) if

$$\mu\bigg(\bigcup_{Q\in \mathsf{HD}(R)}Q\bigg) \geqslant \frac{1}{4}\,\mu(R)$$

Also, we write $R \in ID_U$ (increasing density because of unbalanced cubes) if

$$\mu\bigg(\bigcup_{Q\in\widetilde{\mathsf{UB}}(R)}Q\bigg)\geqslant \frac{1}{4}\,\mu(R).$$

Additionally, let

$$ID = ID_H \cup ID_U.$$

Lemma 3.45 (Lemma 5.4 and its proof in [AT]). If $R \in ID$, then

$$\begin{split} &\Theta_{\mu}(2B_R)^2\,\mu(R)\lesssim \frac{1}{A^2}\sum_{Q\in\mathsf{HD}(R)}\Theta_{\mu}(2B_Q)^2\,\mu(Q)\\ &\Theta_{\mu}(2B_R)^2\,\mu(R)\lesssim \frac{\gamma^2}{\tau^2}\sum_{Q\in\widetilde{\mathsf{UB}}(R)}\Theta_{\mu}(2B_Q)^2\,\mu(Q). \end{split}$$

Moreover, if A is such that $A^{-1} \leqslant \tau^2$ and $\gamma \leqslant \tau^3$, then

$$\Theta_{\mu}(2B_R)^2 \,\mu(R) \leqslant c\tau^4 \sum_{Q \in \operatorname{Next}(R)} \Theta_{\mu}(2B_Q)^2 \,\mu(Q),$$

where c > 0 is some absolute constant.

3.13.5The packing condition

Recall that we assume linear growth of μ , i.e.

$$\mu(B(x,r)) \leqslant C_* r \qquad \forall x \in \operatorname{spt} \mu, \quad r > 0,$$
(3.66)

for some constant $C_* > 0$ (see (0.12)). Using this assumption, we will prove the following.

Lemma 3.46. If the parameters and thresholds in Section 3.7 are chosen properly, then

$$\sum_{R \in \mathsf{Top}} \Theta_{\mu} (2B_R)^2 \,\mu(R) \leqslant \mathsf{c}_5 \, p_0(\mu) + c \, C_*^2 \,\mu(\mathbb{C}), \tag{3.67}$$

where $c_5 = c_5(\tau, A, \theta_0, \gamma, \varepsilon_0, \alpha, \delta) > 0$ and c > 0.

Proof. For a given $k \ge 0$, we set $\mathsf{Top}_0^k = \bigcup_{0 \le j \le k} \mathsf{Top}_j$ and $ID_0^k = ID \cap \mathsf{Top}_0^k$. To prove (3.67), first we deal with the cubes from the ID family. By Lemma 3.45,

$$\begin{split} \sum_{R \in ID_0^k} \Theta_\mu(2B_R)^2 \,\mu(R) &\leqslant c\tau^2 \sum_{R \in ID_0^k} \sum_{Q \in \mathsf{Next}(R)} \Theta_\mu(2B_Q)^2 \,\mu(Q) \\ &\leqslant c\tau^2 \sum_{R \in \mathsf{Top}_0^{k+1}} \Theta_\mu(2B_R)^2 \,\mu(R), \end{split}$$

because the cubes from Next(R) with $R \in \mathsf{Top}_0^k$ belong to Top_0^{k+1} . So we have

$$\begin{split} &\sum_{R \in \mathsf{Top}_0^k} \Theta_{\mu} (2B_R)^2 \,\mu(R) \\ &= \sum_{R \in \mathsf{Top}_0^k \setminus ID_0^k} \Theta_{\mu} (2B_R)^2 \,\mu(R) + \sum_{R \in ID_0^k} \Theta_{\mu} (2B_R)^2 \,\mu(R) \\ &\leqslant \sum_{R \in \mathsf{Top}_0^k \setminus ID_0^k} \Theta_{\mu} (2B_R)^2 \,\mu(R) + c\tau^2 \sum_{R \in \mathsf{Top}_0^k} \Theta_{\mu} (2B_R)^2 \,\mu(R) + c\tau^2 \sum_{R \in \mathsf{Top}_{k+1}} \Theta_{\mu} (2B_R)^2 \,\mu(R) \\ &\leqslant \sum_{R \in \mathsf{Top}_0^k \setminus ID_0^k} \Theta_{\mu} (2B_R)^2 \,\mu(R) + c\tau^2 \sum_{R \in \mathsf{Top}_0^k} \Theta_{\mu} (2B_R)^2 \,\mu(R) + c \,\tau^2 \, C_*^2 \mu(R_0), \end{split}$$

where we took into account that $\Theta_{\mu}(2B_R) \leq C_*$ for every $R \in \mathsf{Top}$ (and in particular for all $R \in \mathsf{Top}_{k+1}$). So, having τ small enough, we deduce that

$$\sum_{R \in \mathsf{Top}_0^k} \Theta_\mu (2B_R)^2 \,\mu(R) \leqslant 1.1 \sum_{R \in \mathsf{Top}_0^k \setminus ID_0^k} \Theta_\mu (2B_R)^2 \,\mu(R) + c\tau^2 \, C_*^2 \mu(R_0). \tag{3.68}$$

Let us estimate the first term in the right hand side of (3.68). First note that

$$\mu\left(R\setminus\bigcup_{Q\in\mathsf{HD}(R)\cup\widetilde{\mathsf{UB}}(R)}Q\right)\geqslant\frac{1}{2}\,\mu(R)\qquad\text{for any}R\in\mathsf{Top}_0^k\setminus ID_0^k.$$

Next, by applying the inequalities (c) in Lemma 3.44 and recalling (3.63) and (3.64) we get

$$\begin{split} \mu(R) &\leqslant 2\,\mu\bigg(R \setminus \bigcup_{Q \in \mathsf{Next}(R)} Q\bigg) + 2\,\mu\bigg(\bigcup_{Q \in \widetilde{\mathsf{O}}(R) \cup \mathsf{BS}(R)} Q\bigg) \\ &\leqslant 2\,\mu(G_R) + 2\mu\bigg(\bigcup_{Q \in \widetilde{\mathsf{O}}(R)} Q\bigg) + 2\sum_{\substack{Q \in \mathsf{BS}(R)\\ (\text{if } R \in \mathsf{T}_{VF}(\theta_0))}} \mu(Q) + 2\sum_{\substack{Q \in \mathsf{BS}(R)\\ (\text{if } R \notin \mathsf{T}_{VF}(\theta_0))}} \mu(Q) \\ &\leqslant 2\,\mu(G_R) + 2\sqrt{\tau}\,\mu(R) + 2\sum_{\substack{Q \in \mathsf{BS}(R)\\ (\text{if } R \notin \mathsf{T}_{VF}(\theta_0))}} \mu(Q) \\ &+ \frac{2\alpha^{-2}}{\Theta_{\mu}(2B_R)^2} \sum_{Q \in \mathsf{Tree}(R)} p_0^{[\delta,Q]}(\mu\lfloor 2B_Q, \mu\lfloor 2B_R, \mu\lfloor 2B_R). \end{split}$$

Suppose that τ is small enough to get

$$\begin{split} \mu(R) &\leqslant 2.1\,\mu(G_R) + 2.1\sum_{\substack{Q\in \mathsf{BS}(R)\\ (\text{if } R\notin \mathsf{T}_{VF}(\theta_0))}}\mu(Q) \\ &+ \frac{2.1\alpha^{-2}}{\Theta_{\mu}(2B_R)^2}\sum_{Q\in \mathsf{Tree}(R)}p_0^{[\delta,Q]}(\mu\lfloor 2B_Q,\mu\lfloor 2B_R,\mu\lfloor 2B_R). \end{split}$$

So we deduce from (3.68) that

$$\sum_{R \in \mathsf{Top}_{0}^{k}} \Theta_{\mu}(2B_{R})^{2} \mu(R)$$

$$\leq 3 \sum_{R \in \mathsf{Top}_{0}^{k} \setminus ID_{0}^{k}} \Theta_{\mu}(2B_{R})^{2} \mu(G_{R}) \qquad (3.69)$$

$$+ \frac{3}{\alpha^{2}} \sum_{R \in \mathsf{Top}_{0}^{k}} \sum_{Q \in \mathsf{Tree}(R)} p_{0}^{[\delta,Q]} (\mu \lfloor 2B_{Q}, \mu \lfloor 2B_{R}, \mu \lfloor 2B_{R})$$

$$+ 3 \sum_{R \in \mathsf{Top}_{0}^{k} \setminus (ID_{0}^{k} \cup \mathsf{T}_{VF}(\theta_{0}))} \Theta_{\mu}(2B_{R})^{2} \sum_{Q \in \mathsf{BS}(R)} \mu(Q)$$

$$+ c\tau^{2} C_{*}^{2} \mu(R_{0}).$$

In order to deal with the first sum on the right hand side we take into account that $\Theta_{\mu}(2B_R) \leq C_*$ for all $R \in \text{Top}$ by (3.66) and that the sets G_R with $R \in \text{Top}$ are pairwise

disjoint. Then we get

$$\sum_{R \in \mathsf{Top}_0^k \setminus ID_0^k} \Theta_\mu(2B_R)^2 \,\mu(G_R) \leqslant c \, C_*^2 \,\mu(R_0).$$

On the other hand, the double sum in (3.69) does not exceed

$$2\sum_{Q\in\mathcal{D}}p_0^{[\delta,Q]}(\mu\lfloor 2B_Q,\mu\lfloor 2B_R,\mu\lfloor 2B_R)\leqslant c(\delta)\,p_0(\mu),$$

by the finite superposition of the domains of integration. Recall that $\delta = \delta(\gamma, \varepsilon_0)$. So we obtain

$$\sum_{R \in \mathsf{Top}_0^k} \Theta_{\mu}(2B_R)^2 \,\mu(R) \leqslant c \, C_*^2 \,\mu(R_0) + c(\tau, A, \gamma, \varepsilon_0, \alpha) \, p_0(\mu)$$

$$+ c \sum_{R \in \mathsf{Top}_0^k \setminus (ID_0^k \cup \mathsf{T}_{VF}(\theta_0))} \Theta_{\mu}(2B_R)^2 \sum_{Q \in \mathsf{BS}(R)} \mu(Q).$$
(3.70)

The third term in (3.70) without the constant may be written as the sum

$$\sum_{R\in \operatorname{Top}_0^k \backslash (ID_0^k \cup \operatorname{T}_{VF}(\theta_0))} \Theta_\mu(2B_R)^2(S_1(R) + S_2(R)),$$

where

$$S_1(R) = \sum_{Q \in \mathsf{BS}(R) \cap \mathsf{T}_{VF}(\theta_0) \setminus ID_0^{k+1}} \mu(Q) \text{ and } S_2(R) = \sum_{Q \in \mathsf{BS}(R) \cap \mathsf{T}_{VF}(\theta_0) \cap ID_0^{k+1}} \mu(Q).$$

Note that we have the intersection with $\mathsf{T}_{VF}(\theta_0)$ in these sums. This is so because for any $Q \in \mathsf{BS}(R)$, where $R \in \mathsf{Top} \setminus \mathsf{T}_{VF}(\theta_0)$, it holds that

$$\theta_V(L_Q) \ge \measuredangle(L_Q, L_R) - \theta_V(L_R) \ge 2(1 + C_F)\theta_0 - (1 + C_F)\theta_0 = (1 + C_F)\theta_0,$$

and thus $Q \in \mathsf{T}_{VF}(\theta_0)$.

Let us estimate $S_1(R)$. Since $Q \in \mathsf{T}_{VF}(\theta_0) \setminus ID_0^{k+1}$, we deduce from (c) in Lemma 3.44 that

$$\begin{split} \mu(Q) &\leqslant 2\,\mu\Big(Q \setminus \bigcup_{P \in \mathsf{Next}(Q)} P\Big) + 2\,\mu\bigg(\bigcup_{P \in \tilde{\mathsf{O}}(Q) \cup \mathsf{BS}(Q)} P\bigg) \\ &\leqslant 2\,\mu(G_Q) + 2\mu\bigg(\bigcup_{Q \in \tilde{\mathsf{O}}(R)} Q\bigg) + 2\sum_{\substack{P \in \mathsf{BS}(Q)\\(\text{if } Q \in \mathsf{T}_{VF}(\theta_0))}} \mu(P) \\ &\leqslant 2\,\mu(G_Q) + 2\sqrt{\tau}\,\mu(Q) + \frac{2\alpha^{-2}}{\Theta_{\mu}(2B_Q)^2} \sum_{P \in \mathsf{Tree}(Q)} p_0^{[\delta,P]}(\mu\lfloor 2B_P, \mu\lfloor 2B_Q, \mu\lfloor 2B_Q). \end{split}$$

If τ is small enough, then

$$\mu(Q) \leqslant 2.1\,\mu(G_Q) + \frac{2.1\alpha^{-2}}{\Theta_{\mu}(2B_Q)^2} \sum_{P \in \mathsf{Tree}(Q)} p_0^{[\delta,P]}(\mu \lfloor 2B_P, \mu \lfloor 2B_Q, \mu \lfloor 2B_Q).$$

Recall that $\mathsf{BS}(R) \cap \mathsf{T}_{VF}(\theta_0) \setminus ID_0^{k+1} \subset \mathsf{Next}(R)$. So we deduce that

$$S_1(R) \leqslant 2.1 \sum_{Q \in \mathsf{Next}(R)} \left(\mu(G_R) + \frac{\alpha^{-2}}{\Theta_\mu(2B_R)^2} \sum_{P \in \mathsf{Tree}(Q)} p_0^{[\delta,P]}(\mu \lfloor 2B_P, \mu \lfloor 2B_Q, \mu \lfloor 2B_Q) \right).$$

Consequently, using that $\Theta_{\mu}(2B_R) \lesssim C_*$, we obtain

$$\begin{split} &\sum_{R\in \mathsf{Top}_0^k \setminus (ID_0^k \cup \mathsf{T}_{VF}(\theta_0))} \Theta_\mu(2B_R)^2 S_1(R) \\ &\leqslant c \, C_*^2 \sum_{R\in \mathsf{Top}_0^k} \sum_{Q\in \mathsf{Next}(R)} \mu(G_Q) \\ &\quad + \frac{c}{\alpha^2} \sum_{R\in \mathsf{Top}_0^k \setminus (ID_0^k \cup \mathsf{T}_{VF}(\theta_0))} \sum_{Q\in \mathsf{Next}(R)} \sum_{P\in \mathsf{Tree}(Q)} p_0^{[\delta,P]}(\mu\lfloor 2B_P, \mu\lfloor 2B_Q, \mu\lfloor 2B_Q)) \\ &\leqslant c \, C_*^2 \sum_{R\in \mathsf{Top}_0^{k+1}} \mu(G_R) + \frac{c}{\alpha^2} \sum_{R\in \mathsf{Top}_0^{k+1}} \sum_{P\in \mathsf{Tree}(R)} p_0^{[\delta,P]}(\mu\lfloor 2B_P, \mu\lfloor 2B_R, \mu\lfloor 2B_R)). \end{split}$$

Take into account that the sets G_R with $R \in \mathsf{Top}$ are disjoint and that the last (double) sum is controlled by $c(\delta) p_0(\mu)$ by the finite superposition of the domains of integration. So we have

$$\sum_{R\in\operatorname{Top}_0^k\backslash (ID_0^k\cup\operatorname{T}_{VF}(\theta_0))}\Theta_\mu(2B_R)^2S_1(R)\leqslant c\,C_*^2\mu(R_0)+c(\tau,A,\delta,\alpha)\,p_0(\mu).$$

Now we estimate $S_2(R)$. Since $\mathsf{BS}(R) \cap \mathsf{LD}(R) = \emptyset$, for each $Q \in \mathsf{BS}(R)$ we have $\Theta_{\mu}(2B_Q) \ge \tau \Theta_{\mu}(2B_R)$ and thus

$$S_2(R) \leqslant \frac{1}{\tau^2 \Theta_\mu (2B_R)^2} \sum_{Q \in \mathsf{BS}(R) \cap \mathsf{T}_{VF}(\theta_0) \cap ID_0^{k+1}} \Theta_\mu (2B_Q)^2 \mu(Q).$$

Since $Q \in ID_0^{k+1}$, by Lemma 3.45,

$$S_{2}(R) \leqslant \frac{1}{\tau^{2}\Theta_{\mu}(2B_{R})^{2}} \sum_{Q \in \mathsf{BS}(R)\cap\mathsf{T}_{VF}(\theta_{0})\cap ID_{0}^{k+1}} c\tau^{4} \sum_{P \in \mathsf{Next}(Q)} \Theta_{\mu}(2B_{P})^{2}\mu(P)$$
$$\leqslant \frac{c\tau^{2}}{\Theta_{\mu}(2B_{R})^{2}} \sum_{Q \in \mathsf{BS}(R)\cap\mathsf{T}_{VF}(\theta_{0})\cap ID_{0}^{k+1}} \sum_{P \in \mathsf{Next}(Q)} \Theta_{\mu}(2B_{P})^{2}\mu(P).$$

Consequently, taking into account that $\mathsf{BS}(R) \cap \mathsf{T}_{VF}(\theta_0) \cap ID_0^{k+1} \subset \mathsf{Next}(R)$ and $\mathsf{Top}_0^k \setminus (ID_0^k \cup \mathsf{T}_{VF}(\theta_0)) \subset \mathsf{Top}_0^k$, we obtain

$$\begin{split} \sum_{R\in\operatorname{Top}_0^k\backslash (ID_0^k\cup\operatorname{T}_{VF}(\theta_0))} \Theta_\mu(2B_R)^2 S_2(R) &\leqslant c\tau^2 \sum_{R\in\operatorname{Top}_0^k} \sum_{Q\in\operatorname{Next}(R)} \sum_{P\in\operatorname{Next}(Q)} \Theta_\mu(2B_P)^2 \mu(P) \\ &\leqslant c\tau^2 \sum_{R\in\operatorname{Top}_0^{k+2}} \Theta_\mu(2B_R)^2 \mu(R) \\ &\leqslant c\tau^2 \sum_{R\in\operatorname{Top}_0^k} \Theta_\mu(2B_R)^2 \mu(R) + c\tau^2 C_*^2 \mu(R_0). \end{split}$$

Coming back to (3.70), we deduce that

$$\begin{split} \sum_{R\in \operatorname{Top}_0^k} &\Theta_\mu(2B_R)^2 \,\mu(R) \\ &\leqslant c \, C_*^2 \,\mu(R_0) + c(\tau, A, \delta, \alpha) \, p_0(\mu) + c\tau^2 \, \sum_{R\in \operatorname{Top}_0^k} \Theta_\mu(2B_R)^2 \mu(R) \end{split}$$

Choosing τ small enough and recalling the information in Section 3.7 yield

$$\sum_{R \in \mathsf{Top}_0^k} \Theta_{\mu} (2B_R)^2 \, \mu(R) \leqslant \mathsf{c}_5 p_0(\mu) + c \, C_*^2 \, \mu(R_0),$$

where c_5 actually depends on all the parameters and thresholds mentioned in Section 3.7. Letting $k \to \infty$ finishes the proof of Lemma 3.46.

3.13.6 The end of the proof of Main Lemma

We first prove an additional property. For $Q, \tilde{Q} \in \mathcal{D}$ with $Q \subset \tilde{Q}$, define

$$\delta_{\mu}(Q,\tilde{Q}) = \int_{2B_{\tilde{Q}} \backslash 2B_Q} \frac{1}{|y - z_Q|} d\mu(y),$$

where z_Q is the center of B_Q , see Lemma 3.1. Then the following statement holds.

Lemma 3.47. For all $Q \in \text{Next}(R)$ there exists a cube $\tilde{Q} \in \text{DbTree}(R)$ such that $\delta_{\mu}(Q, \tilde{Q}) \lesssim_{\tau,A} \Theta_{\mu}(2B_R)$ and $2B_{\tilde{Q}} \cap \Gamma_R \neq \emptyset$.

Proof. Take $Q' \supset Q$ such that $Q' \in \mathsf{Stop}(R)$. By Lemma 3.11, there exists $\tilde{Q} \in \mathsf{DbTree}(R)$ such that $Q' \subset \tilde{Q}$ and $r(Q') \approx_{\tau,A} r(\tilde{Q})$. Moreover, one can easily deduce from Lemma 3.26 that $2B_{\tilde{Q}} \cap \Gamma_R \neq \emptyset$ if ε_0 is small enough (since $\tilde{Q} \in \mathsf{DbTree}(R)$, there is $x \in \tilde{Q} \setminus R_{\mathsf{Far}}$).

Furthermore, split

$$\delta_{\mu}(Q,\tilde{Q}) = \int_{2B_{\tilde{Q}} \backslash 2B_{Q'}} \frac{1}{|y-z_Q|} d\mu(y) + \int_{2B_{Q'} \backslash 2B_Q} \frac{1}{|y-z_Q|} d\mu(y).$$

In the first integral we have $|y - z_Q| \gtrsim r(Q') \approx_{\tau,A} r(\tilde{Q})$ as $y \notin 2B_{Q'}$ and therefore

$$\int_{2B_{\tilde{Q}} \setminus 2B_{Q'}} \frac{1}{|y - z_Q|} d\mu(y) \lesssim_{\tau,A} \Theta_{\mu}(2B_{\tilde{Q}}) \lesssim_{\tau,A} \Theta_{\mu}(2B_R),$$

where we also used the right hand side inequality in (3.15) in Lemma 3.9. To estimate the second integral we take into account that by construction there are no doubling cubes strictly between Q and Q'. This together with Lemma 3.4 and properties of Q' and \tilde{Q} imply by standard estimates (in particular, splitting the domain of integration into annuli with respect to the intermediate cubes between Q and Q') that

$$\int_{2B_{Q'}\setminus 2B_Q} \frac{1}{|y-z_Q|} d\mu(y) \lesssim \Theta_{\mu}(100B(Q')) \lesssim_{\tau,A} \Theta_{\mu}(2B_{\tilde{Q}}) \lesssim_{\tau,A} \Theta_{\mu}(2B_R).$$

Thus we are done.

Lemma 3.44, Lemma 3.47 and the property (3.65) allow us to use arguments as in [Tol6, Lemma 17.6] in order to show that if $\mu(B(x,r)) \leq C_*r$ for all $x \in \mathbb{C}$, then

$$c^2(\mu) \lesssim \sum_{R \in \mathsf{Top}} \Theta_\mu(2B_R)^2 \, \mu(R)$$

for our family Top. By combining this estimate and the identity (0.15) with Lemma 3.46 for fixed suitable parameters from Section 3.7, we obtain

$$p_{\infty}(\mu) \lesssim p_0(\mu) + C_*^2 \mu(\mathbb{C})$$

as wished.

3.14 The case of curvature

Here we come back to the notion of curvature $c^2(\mu)$. Recall that $p_{\infty}(\mu) = \frac{1}{4}c^2(\mu)$ as has been already mentioned.

It is easy to see that one can exchange p_0 for c^2 in the stopping conditions. Then we can prove the following analogue of Lemma 3.46 by the arguments used above.

Lemma 3.48. If the parameters and thresholds in Section 3.7 are chosen properly, then

$$\sum_{R\in\mathsf{Top}}\Theta_{\mu}(2B_R)^2\,\mu(R)\leqslant\mathsf{c}_6\,c^2(\mu)+c\,C^2_*\,\mu(\mathbb{C}),\tag{3.71}$$

where $c_6 = c_6(\tau, A, \theta_0, \gamma, \varepsilon_0, \alpha, \delta) > 0$ and c > 0.

A more direct way to prove it is to use Lemma 3.46 and the inequality (2.5).

Now recall the following theorem from [AT]:

If μ is a finite compactly supported measure such that $\mu(B(x,r)) \leq r$ for all $x \in \mathbb{C}$ and r > 0, then

$$c^{2}(\mu) + \mu(\mathbb{C}) \approx \iint_{0}^{\infty} \beta_{\mu,2}(x,r)^{2} \Theta_{\mu}(x,r) \frac{dr}{r} d\mu(x) + \mu(\mathbb{C}), \qquad (3.72)$$

where the implicit constants are absolute.

Note that the part \leq of (3.72) was proved in [AT] by means of the David-Mattila lattice and a corona type construction similar to the one we considered in this chapter. However, the part \geq was proved in [AT] by the corona decomposition of [Tol3] that involved the usual dyadic lattice $\mathcal{D}(\mathbb{C})$, instead of the David-Mattila lattice \mathcal{D} .

Using Lemma 3.48 we can also prove the part \gtrsim of (3.72) using only the David-Mattila lattice and an associated corona type construction and thus unify the approach with the proof of the part \leq in [AT]. As predicted in [Tol6, Section 19], this indeed simplifies some of the technical difficulties arising from the lack of a well adapted dyadic lattice to the measure μ in [Tol3].

Clearly, we need to show that

$$\iint_0^\infty \beta_{\mu,2}(x,r)^2 \,\Theta_\mu(x,r) \,\frac{dr}{r} \,d\mu(x) \lesssim c^2(\mu) + \mu(\mathbb{C})$$

or, equivalently, in a discrete form that

$$\sum_{Q \in \mathcal{D}} \beta_{\mu,2} (2B_Q)^2 \,\Theta_\mu(2B_Q) \,\mu(Q) \lesssim c^2(\mu) + \mu(\mathbb{C}). \tag{3.73}$$

By the packing condition (3.71) for $C_* = 1$, to prove (3.73) it suffices to show that for every $R \in \text{Top}$ the following estimate holds true:

$$S(R) = \sum_{Q \in \widetilde{\operatorname{Tree}}(R)} \beta_{\mu,2} (2B_Q)^2 \,\Theta_{\mu}(2B_Q) \,\mu(Q) \lesssim \Theta_{\mu}(2B_R)^2 \mu(R),$$

where $\widetilde{\mathsf{Tree}}(R)$ contains cubes in R not strictly contained in $\widetilde{\mathsf{Stop}}(R)$. By $\mathsf{St}(R)$ we denote cubes in $\mathsf{Stop}(R)$ not strictly contained in $\widetilde{\mathsf{Stop}}(R)$. Obviously, $\beta_{\mu,2}(2B_Q)^2 \leq 4\Theta_{\mu}(2B_Q)$ for any $Q \in \widetilde{\mathsf{Tree}}(R)$ and therefore

$$S(R) \leqslant \sum_{Q \in \mathsf{Tree}(R) \setminus \mathsf{Stop}(R)} \beta_{\mu,2} (2B_Q)^2 \, \Theta_\mu(2B_Q) \, \mu(Q) + \sum_{Q \in \mathsf{St}(R)} \, \Theta_\mu(2B_Q)^2 \, \mu(Q).$$

By Lemma 3.4, the density of all intermediate cubes between Stop(R) and Stop(R), i.e. cubes in St(R), is controlled by the density of cubes from Stop(R) so it can be shown that

$$\sum_{P \in \mathsf{St}(R)} \, \Theta_{\mu}(2B_P)^2 \, \mu(P) \lesssim \sum_{Q \in \mathsf{Stop}(R)} \, \Theta_{\mu}(2B_Q)^2 \, \mu(Q).$$

Moreover,

$$\sum_{Q \in \mathsf{Stop}(Q)} \Theta_{\mu}(2B_Q)^2 \, \mu(Q) \lesssim_A \Theta_{\mu}(2B_R)^2 \sum_{Q \in \mathsf{Stop}(R)} \mu(Q) \lesssim_A \Theta_{\mu}(2B_R)^2 \mu(R),$$

as cubes in $\mathsf{Stop}(R)$ are disjoint subsets of R.

What is more, arguments similar to those in Lemmas 3.7 and 3.12 imply that

$$\begin{split} \sum_{\substack{Q \in \mathsf{Tree}(R) \backslash \mathsf{Stop}(R)}} & \beta_{\mu,2} (2B_Q)^2 \, \Theta_{\mu}(2B_Q) \, \mu(Q) \\ & \lesssim_{\gamma} \Theta_{\mu} (2B_R)^2 \sum_{\substack{Q \in \mathsf{Tree}(R) \backslash \mathsf{Stop}(R)}} \frac{c_{[\delta,Q]}^2 (\mu \lfloor 2B_Q)}{\Theta_{\mu}(2B_R)^2} \\ & \lesssim_{\alpha,\gamma} \Theta_{\mu} (2B_R)^2 \mu(R). \end{split}$$

Thus, $S_R \leq_{\gamma,\alpha,A} \Theta_{\mu}(2B_R)^2 \mu(R)$, where γ , α and A depend on other parameters and thresholds and are suitably chosen and fixed at the end.

3.15 Further generalisations

Now consider the kernels defined in (0.24), i.e.

$$K_t(z) = \kappa_N(z) + t \cdot \kappa_n(z), \quad t \in \mathbb{R}, \qquad K_\infty(z) = \kappa_n(z), \quad n \leq N, \quad n, N \in \mathbb{N},$$

where, according to (0.16),

$$\kappa_m(z) = \frac{(\operatorname{Re} z)^{2m-1}}{|z|^{2m}}, \qquad m \in \mathbb{N}.$$

Let us show that results similar to Theorems 3.1 and 3.2 also hold for them.

First of all, we recall that

$$p_{\kappa_n}(z_1, z_2, z_3) \leqslant C(n) \cdot c(z_1, z_2, z_3)^2$$
 for all $(z_1, z_2, z_3) \in \mathbb{C}^2$ (3.74)

with some absolute constant C(n) > 0 as proved in [CMPT2, Lemma 7] (see also Remark 2.3 in Chapter 2). The inequality (3.74) and identity (0.15) readily imply that

$$p_{\kappa_n}(\mu) \leqslant C(n) \cdot p_{k_\infty}(\mu) \tag{3.75}$$

for any measure μ . Recall that $p_{k_{\infty}} \equiv p_{\kappa_1}$ by definition.

Furthermore, Lemma 3.6 that we used in the proof (see Section 3.6) may be exchanged for the following more general result.

Lemma 3.49 (Proposition 3.3 in [CP]). If $(z_1, z_2, z_3) \in V_{\mathsf{Far}}(\theta)$, then

$$p_{\kappa_N}(z_1, z_2, z_3) \ge C(N, \theta) \cdot p_{k_\infty}(z_1, z_2, z_3) \quad \text{for some } C(N, \theta) > 0.$$
 (3.76)

Using this lemma, one can easily replace the permutations $p_{\kappa_0} \equiv p_{\kappa_2}$ in the stopping conditions in Sections 3.8.1 with the permutations p_{κ_N} . Then the same arguments lead to the following analogue of Main Lemma for the kernels κ_N .

Lemma 3.50. There exist absolute constants C(N) > 0 and c > 0 such that for any finite measure μ with C_* -linear growth it holds that

$$p_{k_{\infty}}(\mu) \leqslant C(N) \cdot p_{\kappa_N}(\mu) + cC_*^2 \mu(\mathbb{C}).$$
(3.77)

Under the assumptions of Lemma 3.50, we infer from (3.75) and (3.77) that

$$p_{\kappa_n}(\mu) \leqslant C(n,N) \cdot p_{\kappa_N}(\mu) + cC_*^2\mu(\mathbb{C}), \qquad n \leqslant N,$$

with some C(n, N) > 0. This and the perturbation method yield the following result.

Theorem 3.3. There exist constants $\tilde{t}_0 = \tilde{t}_0(n, N) > 0$ and c > 0 such that for any finite measure μ with C_* -linear growth it holds that

$$\sup_{\varepsilon>0} \|T_{\kappa_n,\varepsilon}1\|_{L^2(\mu)} \leqslant \tilde{t}_0^{-1} \sup_{\varepsilon>0} \|T_{\kappa_N,\varepsilon}1\|_{L^2(\mu)} + cC_*\sqrt{\mu(\mathbb{C})}.$$
(3.78)

As a corollary, we get the following theorems with the same \tilde{t}_0 as above.

Theorem 3.4. Let K_t be a kernel of the form (0.24), where $t \in (-\tilde{t}_0, \tilde{t}_0)$, and E a 1-set. If the SIO T_{K_t} is $L^2(\mathcal{H}^1|E)$ -bounded, then E is rectifiable.

Theorem 3.5. Let μ be an AD-regular measure and K_t a kernel of the form (0.24), where $t \in (-\tilde{t}_0, \tilde{t}_0)$. The measure μ is uniformly rectifiable if and only if the SIO T_{K_t} is $L^2(\mu)$ -bounded.

Note that $t_0 \in (0, 1)$ as follows from Remark 1.2, see also Figure 1.1.

Theorems 3.4 and 3.5 complement Theorems 1.2, 2.2 and 2.3 dealing with the kernels K_t . It is important that the pointwise permutations p_{K_t} corresponding to $t \in (-\tilde{t}_0, \tilde{t}_0)$ change sign for n and N from Remark 1.1 so that the direct curvature method fails.

Bibliography

[AT]	J. Azzam and X. Tolsa, Characterization of <i>n</i> -rectifiability in terms of Jones'
	square function: Part II. Geom. Funct. Anal. 25 (2015), no. 5, 1371–1412.
[Cal]	A. P. Calderón, Cauchy integrals on Lipschitz curves and related operators, Proc. Nat. Acad. Sci. U.S.A. 74 (1977) 1324–1327.
[Cho]	V. Chousionis, Singular integrals on Sierpinski gaskets, Publ. Mat. 53 (2009), no. 1, 245–256.
[CL]	V. Chousionis and S. Li, Nonnegative kernels and 1-rectifiability in the Heisenberg group. Anal. PDE 10 (2017), no. 6, 1407–1428.
[CMPT1]	V. Chousionis, J. Mateu, L. Prat and X. Tolsa, Calderón-Zygmund kernels and rectifiability in the plane, Adv. Math. 231 (1) (2012) 535–568.
[CMPT2]	V. Chousionis, J. Mateu, L. Prat and X. Tolsa, Capacities associated with Calderón-Zygmund kernels, Potential Anal. 38 (3) (2013) 913–949.
[CP]	V. Chousionis and L. Prat, Some Calderón-Zygmund kernels and their relation to rectifiability and Wolff capacities, Math. Z. 231 (2016), no. 1-2, 435–460.
[Chr]	M. Christ, Lectures on Singular Integral Operators, Regional Conference Series in Mathematics 77, Amer. Math. Soc., 1990.
[Chu]	P. Chunaev, A new family of singular integral operators whose L^2 -boundedness implies rectifiability. J. Geom. Anal. 27 (2017), no. 4, 2725–2757; see also arXiv:1601.07319 (2016).
[CMT1]	P. Chunaev, J. Mateu and X. Tolsa, Singular integrals unsuitable for the curva- ture method whose L^2 -boundedness still implies rectifiability, arXiv:1607.07663 (2016) (accepted for publication in Journal d'Analyse Mathematique).
[CMT2]	P. Chunaev, J. Mateu and X. Tolsa, A family of singular integral operators which control the Cauchy transform, arXiv:1803.02854 (2018).
[CMM]	R. Coifman, A. McIntosh, Y. Meyer, L'intégrale de Cauchy définit un opérateur borné sur L^2 pour les courbes lipschitziennes, Ann. of Math. (2) 116 (2) (1982) 361–387.
[Dav1]	G. David, Opérateurs intégraux singuliers sur certaines courbes du plan complexe, Ann. Sci. École Norm. Sup. (4) 17 (1) (1984) 157–189.
[Dav2]	G. David, Unrectifiable 1-sets have vanishing analytic capacity, Rev. Mat. Iberoam. 14 (2) (1998) 369–479.

- [Dav3] G. David, Des intégrales singulières bornées sur un ensemble de Cantor, C. R. Acad. Sci. Paris Sér. I Math. 332 (5) (2001) 391–396.
- [DM] G. David and P. Mattila. Removable sets for Lipschitz harmonic functions in the plane. Rev. Mat. Iberoamericana 16(1) (2000), 137–215.
- [DS1] G. David and S. Semmes, Singular Integrals and Rectifiable Sets in \mathbb{R}^n : Beyond Lipschitz graphs, Astérisque No. 193 (1991).
- [DS2] G. David and S. Semmes, Analysis of and on Uniformly Rectifiable Sets, Mathematical Surveys and Monographs, 38. American Mathematical Society, Providence, RI, (1993).
- [DT] G. David and T. Toro, Reifenberg Parameterizations for Sets with Holes, Mem. Amer. Math. Soc. 215 (2012), no. 1012.
- [Far] H. M. Farag, The Riesz kernels do not give rise to higher-dimensional analogues of the Menger-Melnikov curvature. Publ. Mat., 43(1):251–260, 1999.
- [Gir] D. Girela-Sarrión, Geometric conditions for the L^2 -boundedness of singular integral operators with odd kernels with respect to measures with polynomial growth in \mathbb{R}^d , arXiv:1505.07264v2 (2015).
- [Huo] P. Huovinen, A nicely behaved singular integral on a purely unrectifiable set, Proc. Amer. Math. Soc. 129 (11) (2001) 3345–3351.
- [JN] B. Jaye and F. Nazarov, Three revolutions in the kernel are worse than one, International Mathematics Research Notices (2017), to appear; see also arXiv:1307.3678 (2013).
- [Jon1] P.W. Jones, Square functions, Cauchy integrals, analytic capacity, and harmonic measure, in: Harmonic Analysis and Partial Differential Equations, El Escorial, 1987, in: Lecture Notes in Math., vol. 1384, Springer, Berlin, 1989, pp. 24–68.
- [Jon2] P. W. Jones, Rectifiable sets and the traveling salesman problem, Invent. Math. 102 (1990), no. 1, 1–15.
- [Leg] J. C. Léger, Menger curvature and rectifiability, Ann. of Math. 149 (1999), 831–869.
- [Mat1] P. Mattila, Geometry of Sets and Measures in Euclidean Spaces, Cambridge University Press, 1995.
- [Mat2] P. Mattila, Singular integrals, analytic capacity and rectifiability, in: Proceedings of the conference dedicated to Professor Miguel de Guzmán (El Escorial, 1996), J. Fourier Anal. Appl. 3 (1997), Special Issue, 797–812.
- [Mat3] P. Mattila, Singular integrals and rectifiability, in: Proceedings of the 6th International Conference on Harmonic Analysis and Partial Differential Equations (El Escorial, 2000), Publ. Mat. Vol. Extra (2002), 199–208.
- [MMV] P. Mattila, M. S. Melnikov and J. Verdera, The Cauchy integral, analytic capacity, and uniform rectifiability, Ann. of Math. (2) 144 (1996), 127–136.
| [Mel] | M. S. Melnikov, Analytic capacity: discrete approach and curvature of a measure, Sbornik: Mathematics 186(6) (1995), 827–846. |
|-------|---|
| [MV] | M. S. Melnikov and J. Verdera, A geometric proof of the L^2 boundedness of the Cauchy integral on Lipschitz graphs. Internat. Math. Res. Notices (1995), 325–331. |

- [NTV] F. Nazarov, X. Tolsa and A. Volberg, On the uniform rectifiability of ADregular measures with bounded Riesz transform operator: the case of codimension 1, Acta Math. 213 (2014), no. 2, 237–321.
- [PBM] A. P. Prudnikov, Yu. A. Brychkov and O. I. Marichev, Integrals and Series. Vol. 1. Elementary functions, Gordon and Breach Science Publishers, New York, 1986.
- [Tol1] X. Tolsa, Painlevé's problem and the semiadditivity of analytic capacity, Acta Math. 190 (1) (2003) 105–149.
- [Tol2] X. Tolsa, L^2 -boundedness of the Cauchy transform implies L^2 -boundedness of all Calderón-Zygmund operators associated to odd kernels, Publ. Mat., 48(2) (2004) 445–479.
- [Tol3] X. Tolsa, Bilipschitz maps, analytic capacity, and the Cauchy integral, Ann. of Math. 162:3 (2005), 1241–1302.
- [Tol4] X. Tolsa, Uniform rectifiability, Calderón–Zygmund operators with odd kernel, and quasiorthogonality, Proc. Lond. Math. Soc., 98(2) (2009) 393–426.
- [Tol5] X. Tolsa, Analytic Capacity, the Cauchy Transform, and Non-Homogeneous Calderón-Zygmund Theory, Progress in Mathematics 307, Birkhäuser/Springer, Cham, 2014.
- [Tol6] X. Tolsa, Rectifiable Measures, Square Functions Involving Densities, and the Cauchy Transform, Mem. Amer. Math. Soc. 245 (2017), no. 1158, v+130 pp.